

#### WASHINGTON

# Everett City Council Preliminary Agenda 6:30 p.m., Wednesday, May 15, 2024 City Council Chambers

Roll Call

Pledge Of Allegiance

Land Acknowledgment

Approval Of Minutes: May 8, 2024

Mayor's Comments

**Public Comment** 

Council Comments/Liaison Reports

Administration Update

City Attorney

**CONSENT ITEMS:** 

(1) Adopt Resolution Authorizing Claims Against The City Of Everett In The Amount Of \$3,083,393.13 For The Period Ending April 27, 2024 Through May 3, 2024.

Documents:

#### RES CLAIMS PAYABLE APR 27, 2024 - MAY 3, 2024.PDF

(2) Award Invitation For Bid #2024-018 Val-Matic Butterfly Valves And Rotork Actuators To Ferguson Enterprises For \$618,465.46, Which Includes Washington State Sales Tax.

Documents:

#### VALMATIC BUTTERFLY VALVES AWARD.PDF

(3) Award Invitation For Bid #2024-024 Biosolids Loading, Hauling & Land Application To Elysian Fields For A Not To Exceed The Amount Of \$402,773.24, Which Is Not Subject To Washington State Sales Tax.

Documents:

#### **BIOSOLIDS AWARD.PDF**

(4) Authorize Mayor To Sign Over-Pressurization Settlement Agreement With Hos Bros Construction Inc.

Documents:

#### OVER-PRESSURIZATION SETTLEMENT AGREEMENT.PDF

(5) Authorize The Mayor To Sign The Snohomish County Grant Agreement Funding Supporting Structures For The Pallet Shelters Program.

Documents:

#### PALLET SHELTER EXPANSION AGREEMENT.PDF

#### PROPOSED ACTION ITEMS:

(6) CB 2404-05 – 2nd Reading - Adopt An Ordinance Closing The Special Construction Fund Entitled "Fire Station No.2 And Fire Station No.6 Additions And Alterations", Fund 342, Program 029, As Amended By Ordinance No. 3867-22. (3rd & Final Reading 5/22/24)

Documents:

CB 2404-05.PDF

(7) CB 2404-06 – 2nd Reading - Adopt An Ordinance Relating To Uniform Requirements For Food Service Establishments (FSE) And Non-FSE FOG Dischargers Of The Publicly Owned Treatment Works, Repealing And Replacing Ordinance 3071-08 (Chapter 14.42 EMC). (3rd & Final Reading 5/22/24)

Documents:

CB 2404-06.PDF

(8) CB 2404-07 – 2nd Reading - Adopt An Ordinance Relating To Uniform Requirements For Users Of The Publicly Owned Treatment Works, Repealing And Replacing Ordinance 3070-08 (Chapter 14.40 EMC). (3rd & Final Reading 5/22/24)

Documents:

CB 2404-07.PDF

#### **ACTION ITEMS:**

(9) CB 2404-01 – 3rd & Final Reading - Adopt An Ordinance Relating To Amendments To City's Procurement Policy, Amending Ordinance 3781-20.

Documents:

CB 2404-01.PDF

(10) CB 2404-02 – 2nd, 3rd & Final Readings - Adopt An Ordinance Relating To Ambulance And Emergency First Aid Services, Repealing, Amending, And Recodifying Certain Sections Of Chapter 5.72 Of The Everett Municipal Code.

Documents:

CB 2404-02.PDF

(11) CB 2404-03 – 2nd, 3rd & Final Readings - Adopt An Ordinance Relating To Special Business Licenses And Regulations And Repealing Chapters 5.20, 5.92, 5.96 And 5.102 Of The Everett Municipal Code.

Documents:

CB 2404-03.PDF

(12) CB 2404-04 – 2nd, 3rd & Final Readings - Adopt An Ordinance Relating To Admissions Tax, Repealing Chapter 3.20 Of The Everett Municipal Code And Adopting A New Chapter 3.20 Of The Everett Municipal Code.

Documents:

CB 2404-04.PDF

**Executive Session** 

Adjourn

#### PARTICIPATION IN REMOTE COUNCIL MEETINGS

- o Call in to listen to the Council meetings: 425.616.3920, conference ID: 724 887 726#
- Participate remotely via Zoom by registering to speak at <u>everettwa.gov/speakerform</u>. You
  must register no later than 30 minutes prior to the meeting. Or contact Angela Ely at
  425.257.8703 or <u>aely@everettwa.gov</u> and identify the topic you wish to address.
- Provide written public comments by email to Council@everettwa.gov or mail to 2930
   Wetmore Avenue, Suite 9A, Everett, WA 98201. Emailing comments 24 hours prior to the meeting will ensure your comment is distributed to councilmembers and appropriate staff.
- Persons seeking to comment on non-agenda items may be asked to submit the comments in writing if the comment does not address an issue of broad public interest.

#### AGENDAS, BROADCAST AND RECORDINGS

- The Council agendas and meeting recordings can be found, in their entirety, at everettwa.gov/citycouncil.
- The Council meetings are broadcast on government-access cable Comcast Channel 21 and Frontier Channel 29. They are rebroadcast on Monday and Tuesday at noon;
   Thursday at 2 p.m. and 7 p.m.; Friday and Sunday at 7 p.m.; Saturday at 10 a.m.
- o Watch live meetings and recordings at YouTube.com/EverettCity.

#### **CONTACT THE COUNCIL**

If you do not wish to participate in the meeting, we provide these other methods of contacting your elected officials: Email the Council at <a href="Council@everettwa.gov">Council@everettwa.gov</a>.

- o Call the Council offices at 425.257.8703
- o You may call in just to listen to the meeting: 425.616.3920, conference ID 724 887 726#

The City of Everett does not discriminate on the basis of disability in the admission or access to, or treatment in, its programs or activities. Requests for assistance or accommodations can be arranged by contacting the Everett City Council Office at 425.257.8703. For additional information, please visit

our website at https://www.everettwa.gov/3129/American-Disabilities-Act-ADA-and-Title-.



<b>RESOLUTION NO.</b>	
-----------------------	--

## Be it Resolved by the City Council of the City of Everett:

Whereas the claims payable by check against the City of Everett for the period April 27, 2024 through May 3, 2024, having been audited and approved by the proper officers, have been paid and the disbursements made by the same, against the proper funds in payment thereof, as follows:

<u>Fund</u>	<u>Department</u>	<u>Amount</u>	<u>Fund</u>	<u>Department</u>	Amount
002	General Government	10,794.44	101	Parks & Recreation	23,412.85
003	Legal	1,854.21	110	Library	10,012.42
009	Misc Financial Funds	491,912.70	112	Community Theater	5,400.00
010	Finance	12.81	120	Public Works-Streets	62,253.87
015	Information Technology	1,549.28	130	Develop & Const Permit Fee	1,404.19
021	Planning & Community Dev	34,936.50	138	Hotel/Motel Tax	39,565.00
026	Animal Shelter	1,295.52	146	Property Management	37,674.06
031	Police	2,523.98	152	Cum Reserve Library	34.93
032	Fire	320.01	153	<b>Emergency Medical Services</b>	26,003.20
			155	Capital Reserve Fund	79,194.46
	TOTAL GENERAL FUND	\$ 545,199.45	156	Criminal Justice	91.68
			162	Capital Projects Reserve	20,064.69
			197	CHIP Loan Program	47,945.27
			198	Community Dev Block Gran	31,168.18
			303	PW Improvement Projects	237,750.91
			336	Water & Sewer Sys Improv	206,018.27
			354	Parks Capital Construction	35,206.31
			401	Public Works-Utilities	1,338,215.55
			402	Solid Waste Utility	326.00
			425	Public Works-Transit	37,461.07
			430	Everpark Garage	18,274.30
			440	Golf	27,780.40
			501	MVD-Transportation Service	146,254.78
			503	Self-Insurance	45,957.29
			505	Computer Reserve	2,213.05
			637	Police Pension	5,551.83
			638	Fire Pension	3,689.70
			665	Other Special Agency Funds	49,269.42
				TOTAL CLAIMS	\$ 3,083,393.13

Councilperson introducing Res	solution	
Passed and approved this	day of	, 2024
Council President		



**Project title:** 

Award Invitation for Bid #2024-018 Val-Matic Butterfly Valves and Rotork Actuators to Ferguson Enterprises

Council Bill # interoffice use	
	Part
Agenda dates requested:	
May 15, 2024  Briefing Proposed action Consent X Action Ordinance Public hearing Yes X No  Budget amendment:	Fiscal s Funds a Washin
Yes X No	
PowerPoint presentation: Yes X No  Attachments: Bid Tabulation  Department(s) involved: Procurement & Public Works  Contact person: Theresa Bauccio-Teschlog  Phone number: (425) 257-8901  Email: tbauccio@everettwa.gov	Bids we securely life spar water a Rotork a The Inviadvertis Enterpr Invitation
Initialed by:  Sh  Department head  Administration  Council President	

Project:	Val-Matic Butterfly Valves and Rotork Actuators
Partner/Supplier:	Ferguson Enterprises
Location:	City of Everett Water Filtration Plant (WFP)
Preceding action:	02/28/2024 Authorize Release of Invitation for Bid #2024-018 Val-Matic Butterfly Valves and Rotork Actuators.
Fund:	401/Public Works

#### summary statement:

re budgeted in Public Works fund 401. The Award cost is \$618,465.46, which includes gton state sales tax.

#### summary statement:

re requested from companies to provide Val-Matic Butterfly Valves with Rotork Actuators y attached to the valve body. The current valves have reached or exceeded their useful n. The valves are required to control the direct filtration of raw water into clean potable t the plant. The WFP standardized to the combination of Va-Matic butterfly valves and actuators to match the previously upgraded system.

itation for Bid was released on March 6, 2024, to the City of Everett website and sed in the Everett Herald. On April 9, 2024, one bid was received from Ferguson ises. Staff evaluated the bid response and found that it meets the specifications in the on for Bid. Staff recommends an award to Ferguson Enterprises.

#### mendation (exact action requested of Council):

nvitation for Bid #2024-018 Val-Matic Butterfly Valves and Rotork Actuators to Ferguson ises for \$618,465.46, which includes Washington state sales tax.



City of Everett Procurement 2930 Wetmore Ave, Suite 9E Everett WA 98201

(425)-257-8840 phone (425)-257-8864 fax

# REQUEST FOR QUOTATION PRELIMINARY RESULTS

**Quotation #2024-018** 

DATE 4/30/2024

Name: Val-Matic® Butterfly Valves & Rotork Actuators

Contact: Bert Cueva, Senior Procurement Specialist

SUPPLIER	TOTAL
Ferguson Enterprises	\$618,465.46

The process to determine all supplier's responsiveness and responsibility is underway.

Final award information will be posted at: <a href="https://www.everettwa.gov/2713/Bid-opportunities">https://www.everettwa.gov/2713/Bid-opportunities</a>

Please direct any questions to: bids@everettwa.gov.

Thank you for your interest and participation.



# **EVERETT** City Council Agenda Item Cover Sheet

Award Invitation for Bid #2022-024 Biosolids Loading, Hauling & Land Application to Elysian **Project title:** 

Fields LLC

Agenda dates i	reque	ested:
May 15, 2024		
Briefing		
Proposed actio	n	
Consent	Χ	
Action		
Ordinance		
Public hearing		
Yes	X	No
Budget amend	ment	t:
Yes	X	No
PowerPoint pr	esen	tation:
Yes	X	No
Attachments:		
Bid Tabulation		

#### Contact person:

Theresa Bauccio-Teschlog

**Procurement & Public Works** 

#### Phone number:

(425) 257-8901

#### **Email:**

tbauccio@everettwa.gov

Initialed by:			
sh			
Department head	 	 	 •••
Administration	 	 	 

**Council President** 

Consideration:	Award Invitation for Bid #2024-024
Project:	2024 Biosolids Haul and Land Application
Partner/Supplier:	Elysian Fields LLC
	Everett Water Pollution Control Facility
	2/28/2024 Authorizing Release of Invitation for Bid for the Biosolids Haul and Land Application Project.
Fund:	401 Utilities

#### **Fiscal summary statement:**

The biosolids project will be paid from Fund 401-5-200-112-643-480. The job cost number for this project is PW3835-2.

Staff estimates that between 6,000 and 8,000 wet tons of biosolids will be loaded from the Water Pollution Control Facility (WPCF), hauled, and applied to approved Washington State Department of Ecology land application sites. Elysian Fields LLC can accept 8,000 wet tons from the WPCF. The estimated cost for 6,000 wet tons is \$291,075.06 to \$402,773.24 for 8,000 wet tons.

#### **Project summary statement:**

The City's Water Pollution Control Facility removes approximately 2,000 dry tons, which equates to approximately 7,150 wet tons of biosolids annually. The solicitation stated that the estimated amount was between 6,000 to 8,000 wet tons. Bid amounts are based on 8,000 wet tons. However, companies were also asked for the maximum number of wet tons they could accept. Elysian Fields can accept up to 8,000 wet tons.

The biosolids are used for agricultural applications. The award of IFB 2024-024 will allow Elysian Fields LLC. to apply Department of Ecology-regulated biosolids at facilities that have been permitted to receive biosolids as a soil amendment.

Invitation for Bid #2024-024 was released on March 6, 2024, and three bids were received on April 23, 2024. The solicitation was sent to five suppliers, posted on the City website, and advertised in the Everett Herald. Responses were received, as shown in the attached bid tabulation.

#### Recommendation (exact action requested of Council):

Award Invitation for Bid #2024-024 Biosolids Loading, Hauling & Land Application to Elysian Fields for a not to exceed the amount of \$402,773.24, which is not subject to Washington state sales tax.

#### City of Everett Preliminary Bid Tabulation

Contract Title:
Bid Number:
Submittal Deadline:
Project Manager:
Procurement Professional:
Addenda Issued:
Apparent Low Bidder:
Awarded to:

Biosolids Loading, Hauling, and Land Application

2024-024

4/23/2024

Joseph Ferguson

Bert Cueva

None

Elysian Fields LLC

		Sno-Valley Farms Inc Snohomish, WA			Boulder Park Mansfield, WA			Elysian Fields LLC Snohomish, WA		
Item	Rate per wet tons to load and haul Class B Biosolids	Unit Price	Quantity	Extended	Unit Price	Quantity	Extended	Unit Price	Quantity	Extended
1	Site A	\$52.75	5110.00	\$269,552.50	\$84.66	8000.00	\$677,280.00	\$48.50	5749.80	\$278,865.30
1	Site B	\$51.75	3570.00	\$184,747.50				\$48.80	1575.00	\$76,860.00
1	Site C							\$69.68	675.20	\$47,047.94
			Total	\$454,300.00		Total	\$677,280.00		Total	\$402,773.24



**Project title:** Over-Pressurization Settlement Agreement with Hos Bros Construction Inc.

Council Bill #	interoffice use	Project: Authorize Mayor to Sign Over-Pressurization Settlement Agreement				
		Other Party: Hos Bros Construction Inc.				
Agenda dates	s requested:	Location: Boulevard Bluffs Neighborhood				
		Preceding action: None				
Briefing Proposed acti		Fund: Tort Fund				
Proposed acti						
Consent Action	5/15/24	Fiscal summary statement:				
Ordinance Public hearing	σ	Settlement Agreement will require Hos Brothers to pay 50% of adjusted damage claims related to an over-pressurization event on February 3, 2023.				
Yes	X No					
Pudget amon	dmont	Project summary statement:				
Budget amen Yes	X No	•				
************		Hos Brothers is a contractor on the Soundview Project, which is a private development				
PowerPoint p		project in Everett. (Hos Brothers is not a City contractor.) Hos Brothers installed piping, fittings, and related water facilities routing municipal water in or around the Soundview				
Yes	X No	Technology Center.				
Attachments		resimilarity defice.				
Settlement Agreement		On February 3, 2023, the City received notice that some Everett residents in the				
Department(	s) involved:	Boulevard Bluffs Neighborhood were experiencing higher than normal water pressure,				
Public Works,	/Legal	which can cause damage to water heaters and similar items.				
Contact perso	ons:	Upon investigation, it was found that the overpressure occurred after valves were				
Ryan Sass, Da		operated at the Soundview Project before the set point had been calibrated on the water-supply line's pressure reducing valve. The City and Hos Brothers dispute who is at				
Phone number	er:	fault.				
X8942 (Sass);	X8624 (Hall)	The proposed settlement agreement provides that the City and Hes Brothers will call				
Email:		The proposed settlement agreement provides that the City and Hos Brothers will split 50/50 the cost of settling adjusted damage claims regarding the over-pressurization				
rsass@everet		event.				
dhall@everet	twa.gov					
		Recommendation (exact action requested of Council):				
Initialed by: $\mathcal{RS}$		Authorize Mayor to Sign Over-Pressurization Settlement Agreement with Hos Bros Construction Inc.				
Department he	ead					
Administration						
Council Preside	 Pnt					

## CITY OF EVERETT / HOS BROTHERS CONSTRUCTION OVER-PRESSURIZATION EVENT – FEBRUARY 3, 2023 SETTLEMENT AGREEMENT

#### WHEREAS:

- A. Hos Bros Construction Inc. ("Hos Brothers") entered into a subcontract with a third party with respect to the Soundview Project, which is a private development in Everett. Under this subcontract, Hos Brothers installed certain piping, fittings, and related water facilities routing municipal water in or around the Soundview Technology Center.
- **B.** On February 3, 2023, the City of Everett ("City") received notice that some Everett residents were experiencing higher than normal water pressure and resulting damage to pipes and water heaters associated with work on the Soundview Project (the "Over-Pressurization Event").
- C. Upon investigation, it was found that the Over-Pressurization Event occurred after valves were operated at the Soundview Project before the set point had been calibrated on the water-supply line's pressure reducing valve ("PRV"). The City and Hos Brothers dispute whether either of them is at fault for the Over-Pressurization Event.
- **D.** In connection with the Over-Pressurization Event, some Everett residents, or their insurers, have already asserted (or may in the future assert) claims for property damage arising from the Over-Pressurization Event. In this Settlement Agreement, all such past, present, and future Over-Pressurization Event claims by residents, or their insurers, are referred to as "Residential Claims," and these residents are referred to as "Residential Claimants."
- **E.** The City and Hos Brothers dispute liability for the Over-Pressurization Event, but have determined to resolve the Residential Claims efficiently and expeditiously without incurring unnecessary litigation costs, expenses, and attorney fees.
- **F.** It is further recognized that other third parties may have liability or fault, which has not yet been determined or adjudicated.

#### NOW, THEREFORE, IT IS AGREED BY THE UNDERSIGNING PARTIES:

- 1. The Parties. The Parties to this Settlement Agreement are the City of Everett and Hos Bros Construction, Inc., which are referred to as either a "Party" singularly or the "Parties" collectively.
- **2. Effective Date.** The Effective Date of this Settlement Agreement is the date of last signature by either Party to this Settlement Agreement.
- 3. Residential Claims. This Settlement Agreement contains the Parties' settlement procedures for all Residential Claims, including those Residential Claims that

may have already been reported or asserted against the City and/or Hos Brothers and all Residential Claims that may be asserted against the City and/or Hos Brothers in the future.

- 4. Claims Adjustment Procedure. The Parties recognize that each of them, through their respective "Designee," should have an opportunity to review the Residential Claims in accordance with the following Claims Adjustment Procedure:
  - a. Claim Filings. For each Residential Claim filed with the City in accordance with RCW 4.96.020 on the required Tort Claim Form, the City or the City Designee shall forward that documentation to the Hos Brothers Designee identified in Section 9 below.
  - **b.** Claim Documentation. For each Residential Claim filed, the City or the City Designee will gather the claim documentation that it determines is sufficient to proceed with adjustment of the Residential Claim.
    - c. Claim Analysis.
    - (i) Adjustment Standards. In making the adjustments described herein, the Parties agree to use recognized standards for the adjustment of real and personal property liability claims, and consideration will be given to the age and condition of personal property to arrive at an actual cash value for those items where depreciation may be applied (the "Adjustment Standards").
    - City Initial Recommendation. After documentation has been (ii) gathered and reviewed, the City or the City Designee shall prepare a draft report of its findings and propose a recommended settlement amount in accordance with the Adjustment Standards, and shall forward the report and recommendation to the Hos Brothers Designee for review and consideration. Hos Brothers or the Hos Brothers Designee shall within 14 days inform the City's Designee if Hos Brothers or the Hos Brothers' Designee agrees with the City's recommendation or wishes to meet to discuss the Residential Claim in accordance with the Adjustment Standards. If Hos Brothers agrees with the City's recommended settlement amount, that amount shall be the agreed "Adjusted Settlement Amount" for resolution of the Residential Claim. If Hos Brothers does not agree with the City's recommended settlement amount, the Parties shall confer in good faith regarding an Adjusted Settlement Amount for that Residential Claim.
    - (iii) Residential Claimant Reimbursement. After determination of an agreed Adjusted Settlement Amount for a Residential Claim, the City shall offer the Adjusted Settlement Amount to the Residential Claimant, on the condition that the Residential Claimant executes a "Residential Claimant Release" in substantially the form attached

hereto as **Exhibit A** or such other form as the Parties may mutually agree in writing from time to time. If the Residential Claimant in question accepts the offered Adjusted Settlement Amount and returns to the City an executed Residential Claimant Release, then the City shall within two weeks issue to such Residential Claimant a "Settlement Payment" in the amount of the Adjusted Settlement Amount. If the Residential Claimant in question declines the offered Adjusted Settlement Amount or refuses to return to the City an executed Residential Claimant Release, then the City and Hos Brothers shall confer in good faith regarding revision of the Adjusted Settlement Amount or how otherwise to proceed in accordance with the Adjustment Standards with respect to such Residential Claimant.

- Resolution of Disputes Regarding Adjustment of a Residential (iv) Claim. If the Parties for any reason do not agree on an Adjusted Settlement Amount or revision thereof, and this disagreement remains unresolved for more than 30 days, then either Party may refer the matter to a mutually agreed arbitrator, who shall summarily on review of submitted documents (which shall include each Party's proposed Adjusted Settlement Amount or revision thereof) determine in accordance with the Adjustment Standards which of the two Parties' proposed Adjusted Settlement Amounts or revision thereof is to be offered to the Residential Claimant. This determination shall be final and binding on the Parties unless both Parties agree in writing to reject the determination. The arbitrator's cost shall be borne by the Party whose proposed Adjusted Settlement Amount or revision thereof was not selected by the arbitrator.
- Payment to a Residential Claimant, the City shall notify Hos Brothers' Designee of such payment and provide Hos Brothers' Designee with a copy of the associated executed Residential Claimant Release and documentation showing costs incurred by the City for adjustment services for that Residential Claim provided by any third-party adjuster. Within 15 days after Hos Brothers' Designee receives notice of the City's issuance of a Settlement Payment and a copy of the associated executed Residential Claim Release, Hos Brothers shall issue a "Reimbursement Payment" to the City in the amount of the sum of (A) fifty percent (50%) of the City's Settlement Payment to the Residential Claimant and (B) fifty percent (50%) of the City's incurred reasonable costs for third-party adjustment services on that Residential Claim.
- (vi) The Parties agree that the adjustment and settlement processes set forth in this Section 4 are subject to the joint defense or common

interest protection or privilege. The Parties shall, subject to applicable law, keep confidential all documents and records created by either Party (or by a Party's agents or contractors) in connection with (A) the adjustment of Residential Claims, (B) the determination of settlement amounts for Residential Claims, or (C) the arbitration process set forth in subsection 4(iv).

5. Settlement Consideration. The Parties acknowledge and agree that the City's and Hos Brothers' respective payments made pursuant to Section 4 above are good and valuable consideration for this Settlement Agreement, given the risks and costs associated with potential litigation, which is intended to be avoided by this Settlement Agreement. It is agreed that, upon the City's receipt of the Reimbursement Payment associated with a Residential Claim, such Residential Claim is released under the terms set forth in Section 6 herein, irrespective of whether any other Residential Claim has not been resolved.

#### 6. Releases.

- **a. Persons/Entities Released.** Unless specifically excepted in this Settlement Agreement, the following persons and entities are released in accordance with subsection (b) below: the City, Hos Brothers Construction Inc. ("Hos Bros"), the Hos Brothers Designee(s), American Alternative Insurance Corp. ("AAIC"), Gallagher Bassett ("GB"), excess insurers, reinsurers, and each of their respective past and present owners, elected officials, officers, managers, employees, agents, attorneys, representatives, and insurers, and further including their present and future parent and subsidiary companies, affiliates, related entities, predecessors, successors, assigns, spouses, directors, servants, heirs, executors, administrators, and beneficiaries (collectively the "Party Releasees").
- Claims Released. The Party Releasees as to each other are released b. and discharged fully and completely from and against each and every Residential Claim for which (i) the City has issued to the Residential Claimant a Settlement Payment, (ii) the City has obtained an executed Residential Claimant Release, and (iii) the City has received a full Reimbursement Payment from Hos Brothers (the "Released Residential Claims"). Such release encompasses without limitation any and all liability, rights, claims, demands, actions, judgments, and causes of action, in law or in equity, under statutory or common law, contractual or extra-contractual, known or unknown, past, present or future, claimed or unclaimed, direct or indirect, fixed or contingent, liquidated or uncertain, latent or patent, of any kind whatsoever, arising out of or in any way relating to the Released Residential Claims and the Claims Adjustment Procedure with respect to such Released Residential Claims, regardless of whether any such Released Residential Claim is for or comprises property damages, personal injury, torts, statutory claims, environmental claims or damages, attorney fees, costs, hedonic, punitive, or administrative damages or any other type of damages related to or arising from the Released Residential Claims.

- c. Claims Not Released. The Parties understand and agree that third parties to this Settlement Agreement may have liability for the Residential Claims and other matters that may be associated with the Over-Pressurization Event or the Soundview Project. This Settlement Agreement does not release, waive, or affect any claims or rights associated with the Over-Pressurization Event or the Soundview Project other than as between the settling Parties as set forth herein. The Parties reserve all rights to assert claims to enforce the terms of this Settlement Agreement.
- 7. No Admission of Fault or Liability. This Settlement Agreement shall not be used by either Party as evidence of fault or liability as against the counter-Party except to enforce the terms of this Settlement Agreement. The Parties stipulate and agree that the settlement allocation herein shall not be interpreted or construed or used as evidence of the counter-Party's share or putative share of fault, other than to enforce the terms of this Settlement Agreement.
- **8.** Other Claims and Lawsuits. The Parties covenant and agree that neither Party has or will file a lawsuit against the other associated with or related to any Residential Claim except to enforce the terms of this Settlement Agreement.
- 9. Party Designees. The Parties each respectively appoint the following Designees for the purposes of receiving the notices and communications set forth in this Settlement Agreement:
  - a. The City of Everett Designee(s)

**Name: Christine Muth-Schulz** 

Address: 2930 Wetmore Ave., 10-C

Everett, WA 98201

Email Address: cschulz@everettwa.gov

**Telephone Number: 425-257-8702** 

b. Hos Bros Construction Inc. Designee(s)

Name: Donna M. Chamberlin

Address: Lewis Brisbois Bisgaard & Smith, LLP

1111 3<sup>rd</sup> Avenue, Suite 2700

Seattle, Washington 98101

Email Address: Donna.Chamberlin@lewisbrisbois.com

**Telephone Number: (206) 508-1930** 

- 10. Waiver. The waiver or indulgence of any breach or condition precedent of this Settlement Agreement shall not be deemed or construed as a wavier of any other breach or condition precedent; nor shall any breach or waiver of a condition precedent affect the enforceability of the remainder of this Settlement Agreement.
- 11. Conferences Between the Parties. If and when the Parties are required by this Settlement Agreement to confer regarding a matter, such conference shall be by telephone, videoconference, or in person, unless the parties agree otherwise.
- 12. Survival. This Settlement Agreement shall survive the cessation of each Party's obligations hereunder.
- 13. Contract Integration. The Parties understand and agree that this Settlement Agreement represents a fully integrated contract which supplants and fully replaces any previous communication or writing between the Parties concerning the settlement terms.
- **14. Amendments.** No amendments, modifications, addenda, or revisions shall be made to this Settlement Agreement unless mutually agreed to by the parties in writing.
- 15. Use of Titles. The Parties understand and agree that the titles in this Settlement Agreement are for the purpose of organization only and are not to be used to interpret the Settlement Agreement.
- 16. Severability. Should any part of this Settlement Agreement be deemed unenforceable or substituted, the remaining parts of this Settlement Agreement shall remain valid and enforceable by substituting terms as appropriate for the set-aside or substituted part of the Settlement Agreement.
- 17. Attorney Fees and Costs. Each Party to this Settlement Agreement agrees to carry the burden of its own attorney fees and costs, including the cost of any litigation associated with the enforcement of the terms of this Settlement Agreement.
- 18. Governing Law and Venue. This Settlement Agreement shall be construed, interpreted, applied, and governed under the laws of the State of Washington, and exclusive jurisdiction and venue for any action related to this Settlement Agreement shall lie in the state or federal courts located in or with jurisdiction over Snohomish County, Washington.
- 19. **Dispute Resolution.** The Parties agree to work in good faith to resolve any dispute that pertains to, arises from, or is associated with the Over-Pressurization Event, this Settlement Agreement, and/or the Claims Adjustment Procedure. Should either Party have a dispute, such Party shall contact the counter-Party's Designee in writing (including by email), providing sufficient facts and notice to enable the Parties to confer on the issues. The Parties shall confer in good faith to resolve the dispute at least 30 days in advance of the filing any lawsuit concerning the dispute. Regardless of the foregoing, disputes concerning claims adjustment will be resolved in accordance with Section 4(c)(iv).

- **20. Records Cooperation.** Hos Brothers acknowledges and agrees that the City is subject to the Washington Public Records Act. This Settlement Agreement is not confidential. Hos Brothers will cooperate with the City so that the City may comply with its obligations under the Washington Public Records Act.
- 21. Binding on Others. The terms of this Settlement Agreement are binding upon the heirs, successors, affiliated entities, executors and personal representatives of any Party to this Agreement.
- **22. Reliance on Own Counsel.** Each Party represents, acknowledges, and agrees that it has entered into this Settlement Agreement based upon consultation with and the advice of its own attorney, and that it has read and understands this Settlement Agreement, or has been given the opportunity to have its attorney explain the meaning and effect of this Settlement Agreement, and fully accepts the terms of this Settlement Agreement.
- **23. Form of Execution.** This Settlement Agreement is signed with AdobeSign, which is fully binding.
- 24. **Signatory Representations.** The undersigned declare and represent that they are competent and duly and actually authorized to enter into this Settlement Agreement for the Party herein for which they are signing.

# IT IS SO AGREED:

CITY OF EVERETT WASHINGTON	HOS BROS CONSTRUCTION INC.
Cassie Franklin, Mayor	Signature:  Name of Signer: Title of Signer:
Date ATTEST	
Office of the City Clerk	

## EXHIBIT A FORM RESIDENTIAL RELEASE

# PROPERTY DAMAGE RELEASE

Property Address:		Everett, Washington.
The undersigned, consideration of Amount"), for the undersigned at marital community, agents, succe having a claim or right in the pro releases, acquits and forever disc their past, present and future agents insurers (including without limits heirs and successors of and from rights, damages, costs, loss of ser	dollars (\$ dollars (\$ and for the undersigned's excessors and assigns, and on be perty located at the address charges the City of Everett, Ints, servants, employees, adation American Alternative I any and all claims, actions, rvice, expenses and compensions.	(the "Settlement ecutors, heirs, administrators, behalf of any other person shown above, hereby Hos Brothers, and each of dministrators, representatives, Insurance Corp. ("AAIC")), causes of action, demands, asation of any nature
whatsoever that the undersigned a tort, contract, statutory or other to an occurrence on or about Feb including without limitation all c the City of Everett. The undersig above is a general and global rele	theory of recovery, arising bruary 3, 2023 at or near the claims asserted in Tort Claim and acknowledges and agree	from or in any way related address shown above, n No filed with
The undersigned expressly waive exist as of the effective date of the suspect to exist, and which, if known to enter into this settlement.	nis release, but that the unde	ersigned does not know of or
The undersigned directs that the follows: \$ to		tlement Amount as
The undersigned is solely respon materialmen fees, and lien claims Tort Claim No		
It is understood and agreed that the are not to be construed as an admitted that said releasees expressly denoted avoid litigation.	nission of liability by any p	arties hereby released, and

[signature on following page]

The undersigned further declares and represents that no promise, inducement or
agreement not herein expressed has been made to the undersigned, that this Release
contains the entire agreement between the parties hereto, and that the terms of this
Release are contractual and not a mere recital.

THE UNDERSIGNED HAS READ	THE FOREGOING RELEASE AND
FULLY UNDERSTANDS IT.	

Signed this	day of	
at		[City], Washington.



**Project title: Everett Pallet Shelter Expansion** 

Council Bill # N/A				
Agenda dates requested:				
Briefing Proposed action Consent 05/15/24 Action Ordinance Public hearing Yes x No				
Budget amendment:  x Yes No				
PowerPoint presentation:  Yes x No  Attachments:  Everett Pallet Shelter  Expansion Grant  Agreement_SD				
Department(s) involved: Community Development				
Contact person: Kembra Landry				
<b>Phone number:</b> 425-257-7155				
Email: Klandry@everettwa.gov				
Initialed by:  JW  Department head				
Administration				
Council President				

Project:	County ARPA Funding to Expand Everett's Pallet Shelters
<u>.</u>	Snohomish County; U.S. Department of the Treasury
Location:	
Preceding action:	N/A
Fund:	TBD

#### **Fiscal summary statement:**

In 2022, Snohomish County awarded the City of Everett \$250,000 with American Rescue Plan Act (ARPA) funding to expand the City's Pallet Shelter program. Funds were intended to secure city purchased assets to support the provision of facilities on site for expansion of shelter program needs.

#### **Project summary statement:**

Following the City of Everett's launch of the Pallet Community in partnership with Everett Gospel Mission, we recognize the need to further establish additional emergency shelter program as well as expand the site at EGM. Snohomish County has identified the City's efforts to establish housing security for unhoused persons as an appropriate use of ARPA Funding and has awarded the City with additional funds to further pallet shelter efforts.

Funds will be used for supporting structures of pallet shelter programs, such as restroom facilities or sheltered gathering spaces. Supporting structures will be purchased and owned by the City. Matching funds will be met by the City through the provision of pallet shelter sleeping structures for residents.

#### Recommendation (exact action requested of Council):

Authorize the Mayor to sign the Snohomish County grant agreement funding supporting structures for the Pallet Shelters program.

Snohomish County Human Services 3000 Rockefeller Avenue, M/S 305 | Everett, WA 98201 (425) 388-7200



SS	Contract Number:		HCS-23-AR-210	08-198	Maxim	num Contract Amount	\$250,000
TRA	Title of Project / Service: Everett Pallet Shelter Expansion						
CONTRACT SPECIFICS	Start Date: 07/01	/2023	End	Date: (	03/31/2025	Status Determina	tion: Subrecipient
DN ON	Agency Name:	City of	Everett				
CTIN ATIC	Address:	2930 Wetmore Ave., Suite 8-B					
CONTRACTING ORGANIZATION	City, State & Zip:	Everett, WA 98201			IRS Tax No. / EIN:	91-6001248	
CON ORG	Contact Person:	Kembra	a Landry			Unique Entity Identif	fier: C1HRGLALTFU9
	Telephone:	425-25	7-7155		Email Addre	ess: KLandry@evere	ettwa.gov
Funding Authority: U.S. Department of the Treasury  ALN* No. & Title: 21.027 Coronavirus State and Local Fiscal Recovery Funds  Funding Specifics: American Rescue Plan Act of 2021, PL-117-2, sec. 9901  Federal Agency: U.S. Treasury Federal Award ID No: 21.027 Federal Award Date: 5/11/21							
	reactal Agency.						
COUNTY	Program Division		Co	ontact P	erson	Contact Emai	I Contact Phone
00	Housing and Comm	unity Sei	rvices	Robin H	ood	robin.hood@snoco	o.org 425-388-7266
			_	_		g, which are incorporate	ed herein by reference:
	rms and Conditions HS s Associate Agreement I					n Services Department. nan Services Department	t:
	Terms and Conditions		Attached as Exh			Assurances Certificate	Attached as Exhibit E
Stateme	nt of Work/Project Desc	ription	Attached as Exh	ibit B	Invoice with	h ARPA Certification	Attached as Exhibit F
	Approved Contract Budget Attached as Exhibit C  Certification Re Lobbying Attached as Exhibit D						
In the event of any inconsistency in this contract, the inconsistency shall be resolved by giving precedence in the following order: (a) appropriate provisions of state and federal law, (b) Specific Terms and Conditions, (c) Basic Terms and Conditions, (d) Business Associate Agreement, (e) other attachments incorporated by reference, and (f) other documents incorporated by reference.							
THE CONTRACTING ORGANIZATION IDENTIFIED ABOVE (HEREINAFTER REFERRED TO AS AGENCY), AND SNOHOMISH COUNTY (HEREINAFTER REFERRED TO AS COUNTY), HEREBY ACKNOWLEDGE AND AGREE TO THE TERMS OF THIS CONTRACT. SIGNATURES FOR BOTH PARTIES ARE REQUIRED BELOW. BY SIGNING, THE AGENCY IS CERTIFYING THAT IT IS NOT DEBARRED, SUSPENDED, OR OTHERWISE EXCLUDED FROM PARTICIPATING IN FEDERALLY FUNDED PROGRAMS.  *ALN is the Assistance Listing Number formerly knows as CFDA number for federal funding.							
FOR TH	IE CONTRACTING C	RGANIZ	ZATION:		FOR SN	IOHOMISH COUNTY:	
(Signatur	re)			(Date)		e Brell Vujovic, Director ent of Human Services	(Date)

#### **EXHIBIT A**

#### SPECIFIC TERMS AND CONDITIONS

#### **EVERETT PALLET SHELTER PROJECT**

#### I. INTRODUCTION

Snohomish County has appropriated funds from its portion of Coronavirus Local Fiscal Recovery Funds pursuant to the American Rescue Plan Act (ARPA), PL 117-2, section 9901, codified at 42 U.S.C. Section 802 *et seq*. to be utilized in the following activities in response to the Coronavirus (COVID-19) pandemic:

- A. To respond to the public health emergency or its negative economic impacts, including assistance to households, small businesses, and nonprofits, or aid to impacted industries such as tourism, travel, and hospitality; and
- B. For the provision of government services to the extent of the reduction in revenue due to the COVID-19 public health emergency relative to revenues collected in the most recent full fiscal year prior to the emergency.

#### II. TERMS AND CONDITIONS

This American Rescue Plan Act (APRA), PL 117-2, Section 9901, Coronavirus State and Local Fiscal Recovery Fund Agreement (hereinafter the "Contract") is made by and between the County and Agency, a corporation incorporated under the laws of the State of Washington. The Project is allowable under the following statutory eligible use: to respond to the COVID-19 public health emergency or its negative economic impacts, including assistance to households, small businesses, and nonprofits, or to aid impacted industries such as tourism, travel, and hospitality. As such, the Project is subject to a number of conditions as outlined in the statute, the Interim Final Rule promulgated on May 13, 2021, 31 CFR Part 35, Coronavirus State and Local Fiscal Recovery Funds Frequently Asked Questions promulgated by the U.S. Department of the Treasury as amended, and Compliance and Reporting Guidance – State and Local Fiscal Recovery Funds as promulgated by the U.S. Department of the Treasury as amended. Additional information is available on the U.S. Department of the Treasury website at: https://home.treasury.gov/policy-issues/coronavirus/assistance-for-state-localand-tribal-governments/state-and-local-fiscal-recovery-funds. The Agency is also subject to compliance with regulatory requirements under the Uniform Guidance at 2 CFR Part 200. In compliance with the terms of the Contract, the Agency shall perform the tasks and services and carry out the Project as described in Statement of Work, Exhibit B.

A. Compliance with Specific Laws, Regulations, and Agreements.

The Agency also agrees to comply with all other applicable federal statutes, regulations, and executive orders, and the Agency shall provide for such compliance by other parties in any agreements it enters into with other parties relating to this award. Federal regulations applicable to the funding provided in this Contract include, without limitation, the following:

- 1. 2 CFR Part 200, including any future amendments to 2 CFR Part 200, and any successor or replacement Office of Management and Budget (OMB) Circular or regulation, including, Subpart A (Acronyms and Definitions), Subpart B (General Provisions), Subpart C (Pre-Federal Award Requirements and Contents of Federal Awards) [excluding 204 (Notices of Funding Opportunities), 205 (Federal awarding agency review of merit of proposal), 210 (Pre-Award Costs), 213 (Reporting a determination of a nonfederal entity is not qualified for a federal award)], Subpart D (Post Federal Award Requirements) [excluding 305(b)(8) and (9) regarding Federal Payment, 308 (Revision of budget or program plan), 309 (modification to period of performance)], Subpart E (Cost Principles), and F (Audit Requirements).
- 2. Universal Identifier and System for Award Management (SAM), 2 C.F.R. Part 25 and pursuant to which the award term set forth in Appendix A to 2 C.F.R. Part 25 is hereby incorporated by reference.
- 3. Reporting Subaward and Executive Compensation Information, 2 C.F.R., Part 170, pursuant to which the award term set forth in Appendix A to 2 C.F.R. Part 170 is hereby incorporated by reference.
- 4. Recipient Integrity and Performance Matters, pursuant to which the award term set forth in 2 C.F.R. Part 200, Appendix XII to Part 200 is hereby incorporated by reference.
- 5. Governmentwide Requirement for Drug-Free Workplace, 31 CFR Part 20.
- 6. New Restrictions on Lobbying, 31 CFR Part 21.
- 7. Uniform Relocation Assistance and Real Property Acquisitions Act of 1970 (42 USC §§ 4601-4655) and implementing regulations.
- 8. Generally applicable federal environmental laws and regulations.
- Agency shall comply with the Section 603(c) of the Social Security Act, Treasury's implementing regulations implementing that section, and guidance issued by Treasury regarding the foregoing.

- a. The Agency agrees to undertake the Contract funded project ("Project") more fully described in the Statement of Work (hereinafter Exhibit B). Contract funding is to address necessary expenditures to respond to the public health emergency or its negative economic impacts due to the (COVID-19) pandemic.
- b. Such expenditures must be incurred during the period identified on the Contract Face Page.
- 10. Agency shall comply with the U.S. Department of the Treasury Coronavirus State and Local Fiscal Recovery Funds Guidance and Frequently Asked Questions as amended. Additional information is available on the U.S Department of the Treasury website: <a href="Coronavirus State and Local Fiscal Recovery Funds">Coronavirus State and Local Fiscal Recovery Funds</a> | U.S. Department of the Treasury.
- 11. The Agency shall ensure compliance with all relevant sections of the Revised Code of Washington (RCW) and Washington Administrative Code (WAC) and that RCW and WAC requirements will be followed and adjusted as the RCW and WAC requirements are amended, revised, eliminated, or added.
- 12. The Agency shall ensure that staff and volunteers who have access to children or vulnerable adults are required to have a background check in conformance with Section XXI. Background Checks contained in the Basic Terms and Conditions referenced on the Contract Face Page. A background check is required at the time of employment or commencement of duties under this Contract. The Agency shall ensure that all persons convicted of crimes preventing contact with vulnerable populations are prohibited from having access to those populations.
- 13. The Agency shall comply with all terms and conditions of the Basic Terms and Conditions between the County and the Agency referenced on the Contract Face Page.
- 14. The Agency shall comply with all applicable terms and conditions of the Business Associate Agreement between the County and the Agency referenced on the Contract Face Page.
- B. Protections for Whistleblowers.
  - 1. In accordance with 41 U.S.C. § 4712, Agency may not discharge, demote, or otherwise discriminate against an employee in reprisal for disclosing to any of the list of persons or entities provided below, information that the employee reasonably believes is evidence of gross mismanagement of a federal contract or grant, a gross waste of federal funds, an abuse of authority relating to a federal contract or grant, a substantial and specific

danger to public health or safety, or a violation of law, rule, or regulation related to a federal contract (including the competition for or negotiation of a contract) or grant.

- 2. The list of persons and entities referenced in the paragraph above includes the following:
  - a. A member of Congress or a representative of a committee of Congress;
  - b. An Inspector General;
  - c. The Government Accountability Office;
  - d. A Treasury employee responsible for contract or grant oversight or management;
  - e. An authorized official of the Department of Justice or other law enforcement agency;
  - f. A court or grand jury; or
  - g. A management official or other employee of Agency, contractor, or subcontractor who has the responsibility to investigate, discover, or address misconduct.
- 3. Agency shall inform its employees in writing of the rights and remedies provided under this section, in the predominant native language of the workforce.
- C. Increasing Seat Belt Use in the United States.

Pursuant to Executive Order 13043, 62 FR 19217 (Apr. 18, 1997), Agency should encourage its contractors to adopt and enforce on-the-job seat belt policies and programs for their employees when operating company-owned, rented or personally owned vehicles.

D. Reducing Text Messaging While Driving.

Pursuant to Executive Order 13513, 74 FR 51225 (Oct. 6, 2009), Agency should encourage its employees, subrecipients, and contractors to adopt and enforce policies that ban text messaging while driving, and Agency should establish workplace safety policies to decrease accidents caused by distracted drivers.

#### E. Financial Award

The Agency is hereby awarded, in the form of a subaward, the total sum indicated in the Approved Contract Budget (Exhibit C) to provide the funding for the full undertaking and performance of the Project.

All funds shall be available only in strict accordance with the provisions of this Contract, the Basic Terms and Conditions referenced on the Contract Face Page, and the following:

- 1. Single Audit Act (31 USE §§7501-7507);
- 2. U.S. Office of Management and Budget Uniform Guidance 2 CFR Part 200:
  - a. 2 CFR § 200.303 regarding internal controls;
  - b. 2 CFR §§ 200.330 through 200.332 regarding subrecipient monitoring and management; and
  - c. Subpart F regarding audit requirements;
- 3. Snohomish County Local Standards ("Local Standards"); and
- 4. Other applicable federal, State, and local laws, regulations, and policies governing the funds provided in this Contract.

#### F. Nondiscrimination and Affirmative Action

In addition to the Nondiscrimination and Affirmative Action Provisions (Section XVII of the Basic Terms and Conditions, the Agency shall comply with the following statutes and regulations prohibiting discrimination:

- Title VI of the Civil Rights Act of 1964 (42 U.S.C. 2000d et seq.) and Treasury's implementing regulations at 31 C.F.R. Part 22, which prohibits discrimination on the grounds of race, color, or national origin under program or activities receiving federal financial assistance;
- 2. The Fair Housing Act, Title VII-IX of the Civil Rights Act of 1968 (42 U.S.C. 3601 et seq.), which prohibits discrimination in housing on the basis of race, color, national origin, sex, familial status, or disability;
- 3. Section 504 of the Rehabilitation Act of 1973, as amended (29 U.S.C. 794), which prohibits discrimination on the basis of handicap under any program or activity receiving or benefitting from federal assistance;

- 4. The Age Discrimination Act of 1975, as amended (42 U.S.C. 6101 et seq.) and Treasury's implementing regulations at 31 C.F.R. Part 23, which prohibit discrimination on the basis or age in programs or activities receiving federal financial assistance:
- 5. The American with Disabilities Act of 1990, as amended (42 U.S.C. 12101 et seq.), which prohibits discrimination on the basis of disability under programs, activities and services provided or made available by state and local governments or instrumentalities or agencies thereto; and
- 6. Hatch Act. Agency agrees to comply, as applicable, with requirements of the Hatch Act (5 U.S.C. 1501-1508 and 7324-7328, which limits certain political activities of federal employees as well as certain other employees who work with federally funded programs.

#### G. Conflicts

In addition to complying with the Conflicts of Interest and Kickbacks (Section VIII) provisions of the Basic Terms and Conditions, Agency shall disclose to the County any potential conflict of interest affecting the awarded funds in accordance with 2 C.F.R. § 200.112.

#### H. Public Records

In addition to complying with the Public Records (Section XIII) provisions of the Basic Terms and Conditions, Agency acknowledge that by accepting funds under this Contract, it may be considered the functional equivalent of a public agency under the Public Records Act, chapter 42.56 RCW.

## I. Compliance with Funding Source Requirements

In addition to complying with the Compliance with Funding Source Requirements (Section XI) of the Basic Terms and Conditions, Agency agrees to the following:

- 1. The Agency understands and agrees that the funds disbursed under this Contract may only be used in compliance with Section 603(c) of the Social Security Act, Treasury's regulations implementing that section, and guidance issued by Treasury regarding the foregoing.
- 2. The Agency, by signing this Contract, acknowledges that it has the institutional, managerial, and financial capability to ensure proper planning, management, and completion of the Project set forth in Exhibit B.

#### J. Definitions

For purposes of this Contract, the following terms shall have the following meanings in addition to definitions incorporated by reference:

- 1. Cultural Competence: A set of congruent behaviors, attitudes and policies that come together in a system or agency and enable that system or agency to work effectively in cross-cultural situations. A culturally competent system of care acknowledges and incorporates at all levels the importance of language and culture, assessment of cross-cultural relations, knowledge and acceptance of dynamics of cultural differences, expansion of cultural knowledge and adaptation of services to meet culturally unique needs.
- 2. Housing Services: The services or activities designed to assist individuals or families in locating, obtaining or retaining suitable housing. Component services or activities may include tenant counseling, helping individuals and families to identify and correct substandard housing conditions on behalf of individuals and families who are unable to protect their own interests and assisting individuals and families to understand leases, secure utilities and make moving arrangements.
- 3. Individual: A service recipient who may also be known as client, consumer, patient, or participant depending on the type of service provided.
- 4. Individual Treatment: Planned therapeutic or counseling activity provided to a sole eligible individual by one (1) or more counselors.
- 5. MCO: Acronym for Managed Care Organization.
- 6. Mental Health Professional (MHP): Personnel employed by the Agency who meet the criteria as defined in WAC 246-341-0200.
- 7. Outpatient Counseling: The provision of substance abuse treatment, mental health treatment and other support services according to a prescribed plan in a non-residential setting.
- 8. Quality Assurance: A focus on compliance to minimum requirements (e.g. rules, regulations, and contract terms) as well as reasonably expected levels of performance, quality, and practice.
- 9. Referral: A process of directing an individual to available specialty care or services.

#### III. FISCAL MANAGEMENT

Every subcontract approved by the County and entered into by the Agency under this Contract shall be in writing and shall incorporate all of the clauses in this Section, with word changes where appropriate to properly identify the parties to the subcontract.

## A. Accounting Standards

The Agency agrees to comply with OMB Uniform Guidance and 2 CFR part 200 and to adhere to the accounting principles and procedures required therein, to use adequate internal controls, and to maintain necessary source documentation for all costs incurred.

#### B. Audit and Recovery

All disbursements of funds to the Agency under this Contract shall be subject to audit and recovery of disallowed costs from the Agency. In the event of Agency's noncompliance with Section 603 of the Social Security Act, other applicable laws, Treasury's implementing regulations, guidance, or any reporting or other program requirements, the County may impose additional conditions or take other available remedies as set forth in 2 C.F.R. § 200.339. In the case of a violation of Section 603(c) of the Social Security Act regarding the use of funds, funds shall be subject to recoupment.

#### C. Accounting for Funds

In order to ensure and to provide documentation that the funds are used only as provided in this Contract, the Agency shall account for all funds under this Contract in a separate account or fund.

#### D. Repayment of Funds to County

The Agency shall return funds disbursed to it by the County under this Contract for return by the County to the U.S. Department of the Treasury, upon the occurrence of any of the following events:

- If Agency has any unspent funds on hand as of the earlier of the Contract end date referenced on the Face Page of this Contract or the termination of this Contract under Section XLIII of the Basic Terms and Conditions, Agency shall return all unspent funds to the County within ten (10) calendar days;
- 2. If overpayments are made; or

3. If an audit of the Project by the U.S. Department of the Treasury, the State, or the County determines that the funds have been expended for purposes not permitted by the Section 603 of the Social Security Act, other applicable laws, Treasury's implementing regulations, guidance, or any reporting or other program requirements, the U.S. Department of the Treasury, the County, or this Contract.

In any case, the County shall make a written demand upon the Agency for repayment, such demand to be made in compliance with Section XXXV of the Basic Terms and Conditions, the Agency shall be obligated to repay to the County the funds demanded within sixty (60) calendar days of the demand. No exercise of the County of the right to demand repayment of funds by the Agency shall foreclose the County from making an additional demand for repayment if a return of additional funds is required by the U.S. Department of the Treasury; the County's right to demand repayment from the Agency may be exercised as often as necessary to recoup from the Agency all funds required to be returned by the County to the U.S. Department of the Treasury.

The Agency is solely responsible for seeking repayment from any subcontractor in conformance with its debt collection policy.

#### E. Debts Owed the Federal Government

- 1. Any funds paid to Agency in excess of the amount to which Agency is finally determined to be authorized to retain under the terms of this Contract, that are determined by the Treasury Office of Inspector General to have been misused or that are determined by Treasury to be subject to a repayment obligation pursuant to section 603(e) of the Act and have not been repaid by Agency shall constitute a debt to the federal government.
- 2. Any debts determined to be owed the federal government must be paid promptly by Agency. A debt is delinquent if it has not been paid by the date specified in Treasury's initial written demand for payment, unless other satisfactory arrangements have been made or if the Agency knowingly or improperly retains funds that are a debt as defined in paragraph 1, Treasury will take any actions available to it to collect such a debt.
- 3. Any debts determined to be owed to the County must be promptly paid by Agency. A debt is delinquent if it has not been paid by the date specified in County's initial written demand for payment, unless other satisfactory arrangements have been made or if the Agency knowingly or improperly retains funds that are a debt. The County will take any actions available to it to collect such a debt.

#### F. Cost Principles

The Agency shall administer its Project in conformance with OMB Uniform Guidance and 2 CFR part 200. These principles shall be applied for all costs incurred whether charged on a direct or indirect basis. Pre-award costs, as defined in 2 C.F.R. § 200.458, may not be paid with funding under this Contract. The Agency is not required to provide cost sharing or matching funds under this Contract.

#### G. Indirect Costs

If indirect costs are charged, the Agency will develop an indirect cost allocation plan for determining the Agency's appropriate share of such costs and shall submit such plan to the County for approval in a form specified by the County.

#### H. Federal and State Prevailing Wage Requirements

Use of federal, state, or local funds to reimburse costs associated with labor performed for any type of maintenance, repair, rehabilitation, construction, etc. may trigger State Prevailing Wage requirements per RCW Chapter 39.12. Projects that include construction costs will require performance and payment bonds from the prime contractor.

#### I. Cost Reimbursement

Reimbursement for services delivered under this Contract shall be on a costreimbursement basis. Reimbursement shall be provided for services provided pursuant to the Statement of Work (Exhibit B). The Agency shall submit, in a format prescribed by the County and set forth in Exhibit F to this Contract, an invoice and certification detailing, on a monthly basis, all costs associated with the program based on the Approved Contract Budget (Exhibit C). Use of funds available under this Contract will be reviewed monthly. The Agency certifies that the work to be performed under this Contract does not duplicate any work to be charged against any other contract, subcontract, or source.

No payments in advance or in anticipation of services or supplies to be provided under this Contract shall be made by the County.

#### J. Program Income

The receipt and expenditure of program income, as defined in 24 CFR § 570.500(a), shall be recorded as part of the financial transactions under this Contract in compliance with 24 CFR § 570.504. The Agency shall report monthly to the County on the Program Income Report, all program income generated by activities carried out with funds made available under this Contract.

The Agency must use such income during the Contract period for activities permitted under this Contract and shall reduce requests for additional funds by the amount of any such program income balance on hand. All unexpended program income shall be returned to the County at the end of the Contract period in accordance with 24 CFR § 570.503(b)(3) and 24 CFR § 570.504(c).

#### IV. PROJECT REQUIREMENTS

#### A. Service and Assistance Animals

Persons with disabilities may request a reasonable accommodation for any assistance animal, including an emotional support animal, under both Title VI of the Civil Rights Act of 1964, as amended (P.L. 88-352), and Section 504 of the Rehabilitation Act of 1973 (29 U.S.C. § 794), as amended.

#### B. Procurement

Supplemental to the requirements of the Basic Terms and Conditions, unless specified otherwise in this Contract, the Agency shall procure all materials, property, supplies, or services in accordance with the requirements of 2 CFR § 200.318; Section 6002 of the Solid Waste Disposal Act, as amended by the Resource Conservation and Recovery Act; 24 CFR § 135; and 24 CFR § 576.404.

#### C. Faith-Based Activities

The Agency shall comply with the provisions of 24 CFR § 576.406(b), which, in part, prohibit the Agency from engaging in inherently religious activities, such as worship, religious instruction, or proselytization, as a part of the programs or services funded by this Contract. If the Agency conducts such activities, the activities must be offered separately, in time or location, from the programs or services funded under this Contract, and participation must be voluntary for the beneficiaries under this Contract.

#### D. Political Activities

The Agency agrees that no funds provided, nor personnel employed, under this Contract shall be in any way or to any extent be applied to, or engaged in, the conduct of political activities in violation of 24 CFR § 570.207(a)(3).

#### E. Public Information

1. The Agency shall ensure recognition of the role of the County in providing services through this Contract. All activities and items utilized pursuant to this Contract shall be prominently labeled as to funding source.

2. Any publication produced with funds from this award must display the following language: "This project [is being] [was] supported, in whole or in part, by U.S. Department of the Treasury Coronavirus State and Local Fiscal Recovery Funds awarded to Snohomish County on May 11, 2021.

#### V. PROJECT MANAGEMENT

Recognizing that the body of knowledge regarding COVID-19 is rapidly evolving, the Agency shall take all reasonable measures, based on the existing state-of-the-art knowledge related to the transmission of COVID-19 as made available by the U.S. Centers for Disease Control and Prevention, the Washington State Department of Health, and the Snohomish Health District, to prevent the spread of COVID-19.

#### A. Location and Hours of Service

Services provided under this Contract shall be available in Snohomish County for Snohomish County residents as specified in Exhibit B.

#### B. Individual Grievances

The Agency shall have policies and procedures in place for individual grievances in the case of denial or termination of services or failure to act upon a request for services with reasonable promptness.

#### C. Toll-Free Telephone Service

The Agency shall maintain a telephone system that provides for toll-free calls from all residents of Snohomish County using the services described herein.

#### D. Change in Personnel

The Agency shall notify the County within ten (10) days of change in personnel which may affect the faithful execution of this Contract.

#### E. Subcontracting

The Agency is prohibited from subcontracting any funding and/or services contained within the Contract unless otherwise negotiated with the County. All subcontracting arrangements require prior written approval from the County.

#### F. Emergency Procedures

The Agency shall have a plan for serving individuals during periods when normal services may be disrupted. Disruptions to normal services may include,

but are not limited to, earthquakes, floods, snowstorms, other natural disasters, and pandemics. Particular attention should be made for those individuals who are most at risk, including those individuals who meet the criteria established for being at higher risk for severe illness resulting from COVID-19 as identified by the U.S. Centers for Disease Control and Prevention, the Washington State Department of Health, and the Snohomish Health District. When services are delivered at the Agency's workplace, the plan shall include: contact information for high-risk individuals, a list of emergency services, and stores of emergency provisions including Personal Protective Equipment (PPE).

## G. Continuing Education

The Agency shall ensure their staff are effectively trained to implement the services specified in Exhibit B. The Agency is encouraged to inquire about the availability of additional training funds and opportunities to support their continuing education efforts.

#### VI. RECORDS

- A. The Agency shall maintain records and financial documents sufficient to evidence compliance with Section 603(c) of the Social Security Act, Treasury's implementing regulations implementing that section, and guidance issued by Treasury regarding the foregoing.
- B. The Treasury Office of the Inspector General and the Government Accountability Office, or their authorized representatives, and the County shall have the right of access to records (electronic and otherwise) of Agency in order to conduct audits or other investigations.
- C. Records shall be maintained for a period of six (6) years after all funds have been expended or returned to the County or Treasury, or as otherwise required by law.

#### D. Records to Be Maintained

The Agency shall maintain all records required by Section XXIV. Maintenance of Records in the Basic Terms and Conditions referenced on the Contract Face Page pertaining to the activities funded under this Contract and as further described in Exhibit B. The Agency shall furnish such records to the County or other authorized officials, as requested. The Agency shall maintain records including, but not limited to:

- 1. Records providing a full description of each activity undertaken;
- 2. Records used for data collection for reports as required;

- 3. Records of compliance with conflict of interest requirements;
- 4. Records of compliance with the faith-based activities requirements;
- 5. Records of compliance with the nondiscrimination requirements including:
  - a. Data indicating the race, ethnicity, disability status, sex, and family characteristics of persons and households who have applied for, participated in, or benefited from the Project; and
  - b. Data which records its affirmative action in equal opportunity employment including, but not limited to, employment, upgrading, demotions, transfers, recruitment or recruitment advertising, layoffs, or terminations, pay or other compensation, and selection for training.
- 6. Financial Records, including supporting documentation for all Project costs;
- 7. Any other reporting obligations established by the U.S. Department of the Treasury as they relate to this award.
- B. Individual Information and Confidentiality

The Agency understands that individual information collected under this Contract is private and the use or disclosure of such information, when not directly connected with the administration of the County's or Agency's responsibilities with respect to services provided under this Contract, may be prohibited by federal, State, and local laws regarding privacy and obligations of confidentiality, unless written consent is obtained from such person receiving service and, in the case of a minor, that of a responsible parent or guardian.

In particular, pursuant to 24 CFR § 576.500(x), the address or location of any housing of a participant will not be made public, except as provided under a preexisting privacy policy of the Agency and as consistent with federal, State, and local laws regarding privacy and obligations of confidentiality. The address or location of any domestic violence, sexual assault, or stalking shelter project assisted under this Contract will not be made public, except with written authorization of the person responsible for the operation of such project.

#### VII. PERFORMANCE EVALUATION AND MONITORING

The Agency agrees to participate with the County in any monitoring (on-site and/or desk) or evaluation of the Project conducted by the County to determine Contract compliance and to make available all information in its possession relevant to such evaluation.

The County will monitor the performance of the Agency against the goals and performance standards set forth in this Contract. Remedies for substandard

performance that is not corrected to the County's satisfaction may include Contract suspension or termination following the procedures described in the Basic Terms and Conditions.

When making future funding decisions, the County will consider timeliness and compliance with Contract and reporting requirements from past contracts.

#### VIII. AFTER-THE-AGREEMENT/CLOSE-OUT REQUIREMENTS

The Agency's obligation to the County shall not end until all close-out requirements are completed. Activities during this close-out period shall include, but are not limited to: making final payments, disposing of program assets (including the return of all unused materials, equipment, unspent cash advances, program income balances, and accounts receivable to the County), and determining the custodianship of records. Notwithstanding the foregoing, the terms of this Contract shall remain in effect during any period that the Agency has control over American Rescue Plan Section 9901 Coronavirus State and Local Fiscal Recovery Funds dollars. The County will close-out the award when it determines, in its sole discretion, that all applicable administrative actions and all required work of the grant have been completed.

#### IX. FALSE STATEMENTS

Agency understands that making false statements or claims with this award is a violation of federal law and may result in criminal, civil, or administrative sanctions, including fines, imprisonment, civil damages and penalties, debarment from participating in federal awards or contract, and/or any other remedy available by law.

#### X. DISCLAIMER

The United States has expressly disclaimed any and all responsibility or liability to the County or third persons for the actions of the County or third persons resulting in death, bodily injury, property damages, or any other losses resulting in any way from the performance of this award or any other losses resulting in any way from the performance of the award of Federal funds to the County under section 603(c) of the Act, or any contract or subcontract under such award.

The County expressly disclaims any and all responsibility or liability to the Agency or third persons for the actions of the Contractor or third persons resulting in death, bodily injury, property damages, or any other losses resulting in any way from the performance of this Contract or any other losses resulting in any way from the performance of the Contract, or any subcontract thereto.

The Contract does not in any way establish an agency relationship between or among the United States, the County, and Agency.

#### XI. ENTIRE AGREEMENT; MODIFICATION

- A. This Contract constitutes the entire agreement between the parties as to the subject matter hereof and supersedes all prior discussion and understandings between them.
- B. In addition to the Unilateral Amendment provision contained in Section XLII. B. of the Basic Terms and Conditions, Agency agrees that in consideration of the fluid nature of the COVID-19 pandemic response, unilateral amendment may be utilized by the County in the event any terms or conditions related to the funds provided under this Contract are updated or if new requirements are issued by the federal government.

#### XII. TIME OF THE ESSENCE

Time is of the essence in the performance of each party's obligations under this Contract. Each party will carry out its obligations under this Contract diligently and in good faith.

#### **EXHIBIT B**

#### STATEMENT OF WORK

#### **EVERETT PALLET SHELTER PROJECT**

#### I. PROJECT SUMMARY

Project Name: Everett Pallet Shelter

Identification Number: HCS-23-AR-2108-198

Project Expenditure Category (EC): 2.16 Long-term Housing Security: Services

for Unhoused People

#### **Project Demographics:**

#### For individual/household beneficiaries:

The intended beneficiaries will be from one or more of the following categories (check as many as apply):

X	Impacted	Х	Disproportionately Impacted
Х	Low- or moderate- income households <sup>1</sup>	Х	Low-income households and populations <sup>3</sup>
	Households that experienced unemployment		Households and populations residing in Qualified Census Tract(s) Number(s) Specify:
X	Households that experienced food or housing insecurity		Households that qualify for certain federal programs <sup>4</sup> Specify:
	Households that qualify for certain federal programs <sup>2</sup> Specify:		Households receiving services provided by Tribal governments
	For services to address lost instructional time in K-12 schools: any students that lost access to inperson instruction for a significant period of time		Households residing in U.S. territories or receiving services from these governments Specify:
	Other households or populations that experienced a negative economic impact other than those listed above Specify:		For services to address education disparities, Title I eligible schools
			Other households or populations that experienced a disproportionate negative impact other than those listed above Specify:

Low- or moderate-income households and Qualified Census Tracts are those with (i) income at or below 300 percent of the Federal Poverty Guidelines for the size of the household or (ii) income at or below 65 percent of the Area Median Income for the county and the size of household.

- These programs are: Children's Health Insurance Program; Childcare Subsidies through the Child Care and Development Fund Program; Medicaid; National Housing Trust Fund for affordable housing programs only; and Home Investment Partnerships Program for affordable housing programs only.
- Low-income households and Qualified Census Tracts are those with (i) income at or below 185 percent of the Federal Poverty Guidelines for the size of the household or (ii) income at or below 40 percent of the Area Median Income for the county and the size of household.
- These program are: Temporary Assistance for Needy Families; Supplemental Nutrition Assistance Program; Free- and Reduced-Price Lunch Program, School Breakfast Program; Medicare Part D Low-Income Subsidies; Supplemental Security Income; Head Start and Early Head Start; Special Supplemental Nutrition Program for Women, Infants, and Children; Section 8 and PHA Project Based Vouchers; Low-Income Home Energy Assistance Program; and Pell Grants.

#### **Project Overview** [50-250 words]

The project funds will expand the City of Everett's Pallet Shelter program. Funds would support the City in procurement of supporting structures (hygiene/restroom units, gathering space, etc.) for the operations and sheltering of Women and Children. The Women and Children Pallet shelter provides temporary housing and support services and will be situated on City-owned land. Matching City funds will support the purchase of the 20 residential pallet structures for the site, as well as labor for the connection of the supporting structures to utilities (water, electricity).

#### **Evidence Base/Evaluation:**

Calhoun, K.H., Wilson, J.H., Chassman, S. *et al.* Promoting Safety and Connection During COVID-19: Tiny Homes as an Innovative Response to Homelessness in the USA. *J. Hum. Rights Soc. Work* **7**, 236–245 (2022). <a href="https://doi.org/10.1007/s41134-022-00217-0">https://doi.org/10.1007/s41134-022-00217-0</a>

Wong A, Chen J, Dicipulo R, Weiss D, Sleet DA, Francescutti LH. Combatting Homelessness in Canada: Applying Lessons Learned from Six Tiny Villages to the Edmonton Bridge Healing Program. *International Journal of Environmental Research and Public Health*. 2020; 17(17):6279. https://doi.org/10.3390/ijerph17176279

**Data Elements to Be Collected**: Include all data elements required by the Final Rule and the Compliance and Reporting Guidance

Data elements to be collected include:

- 1. Unduplicated number of individuals being assisted.
- 2. Gender
- 3. Age
- 4. Veteran Status
- 5. Race (American Indian or Alaska Native, Asian, Black or African American, Native Hawaiian or Other Pacific Islander, White, Other, Mixed Race)
- 6. Ethnicity (Hispanic or Latinx, Not Hispanic or Latinx)
- 7. Monthly income

Additional Key Performance Indicators to be included in reports include the number of successful exits to longer-term housing and connection to services such as detox, mental health and substance use disorder treatment, ID assistance, employment assistance and medical support.

Data elements are to be disaggregated by race, ethnicity, gender, income, and other relevant factors to the extent possible.

#### II. STATUTORY ELIGIBLE USE

The Project complies with the following Statutory Eligible Use: to respond to the COVID-19 negative economic impacts, including assistance to households.

#### III. REPORTING

The Agency shall submit to the County such reports as the County requests pursuant to the requirements of federal, state, and local law, regulations, and guidance as applicable. At a minimum, the Agency shall submit, in a format prescribed by the County, the following reports:

Report Title	Description	Due Date
Quarterly Performance Report	Report on the above data elements and key performance indicators for each quarter as well as cumulatively.	July 10, 2024 Oct 10, 2024
Annual Performance Progress Report	Narrative that highlights work-to- date, best practices, lessons learned and how lessons learned are being integrated into program implementation.	July 10, 2024 and at closeout

#### IV. PROJECT DESCRIPTION

Mι	ust ind	clude at a minimum: (format to department standard)
	S	ervice area(s)
	$\boxtimes$	Snohomish County
>	S	ervice location(s):
		Other (Specify): City of Everett

- Target population: Individuals and families experiencing homelessness.
- For individual and/or household beneficiaries: The project shall:
  - o Increase non-congregate shelter beds for people experiencing homelessness consistent with COVID-19 pandemic recommendations.
  - Provide services to those who have been disproportionally impacted by the pandemic as determined by income status and decrease in human services and homeless program and services during the pandemic.
  - Provide shelter to individuals and families who have had negative economic impacts since the pandemic began.
- ➤ The target population will be made aware of the services through the City of Everett's procedures referring individuals to the Volunteers of America's shelter program.
- Key Performance Indicators (KPIs):
  - Number of successful exits to longer-term housing;
  - Length of stay for individuals exited to longer-term housing; and
  - Number of individuals connected to services such as detox, mental health and substance use disorder treatment, ID assistance, employment assistance and medical support.

Exhibit B, Attachment 1

**Project Expenditure Category (EC)** [Check the primary category - Green indicates a new category, Yellow indicates changed number for an existing category with former category in () - \* indicates that the subrecipient must identify the amount of the total funds that are allocated to evidence-based interventions - # indicates that the subrecipient must report on whether the project is primarily serving a disproportionately impacted community]:

	Public Health
	1.12 (1.10) Mental Health Services*#
	1.13 (1.11) Substance Use Services*#
	Negative Economic Impacts
	2.1 Household Assistance: Food Programs*#
	2.2 Household Assistance: Rent, Mortgage, and Utility Aid*#
	2.4 Household Assistance: Internet Access*#
	2.10 (2.7) Assistance to Unemployed or Underemployed Workers including job training, subsidized employment, employment supports, and/or incentives*#
	2.11 (3.6) Healthy Childhood Development: Child Care*#
	2.12 (3.7) Healthy Childhood Development: Home Visiting*#
	2.13 (3.8) Healthy Childhood Environments: Services to Foster Youth or Families
	Involved in Child Welfare System*#
	2.14 (3.1) Healthy Childhood Environments: Early Learning*#
	2.15 (3.10) Long-term Housing Security: Affordable Housing*#
$\boxtimes$	2.16 (3.11) Long-term Housing Security: Services to Unhoused Persons
	2.17 Housing Support: Housing Vouchers and Relocation Assistance for
	Disproportionately Impacted Communities*#
	2.18 Housing Support: Other Housing Assistance*#
	2.19 (3.14) Social Determinants of Health: Community Health Workers or Benefits
_	Navigators*#
	2.22 Strong Healthy Communities: Neighborhood Features that Promote Health and Safety*
	2.24 (3.2) Addressing Educational Disparities: Aid to High-Poverty Districts#
	2.25 (3.3) Addressing Educational Disparities: Academic, Social, and Emotional
	Services*#
	2.26 (3.4) Addressing Educational Disparities: Mental Health Services*#
	2.27 Addressing Impacts of Lost Instructional Time
	2.34 (2.10) Aid to Non-Profit Organizations (Impacted or Disproportionately Impacted#
	2.37 (3.9) Economic Impact Assistance: Other*#
	Administrative
	7.1 Administrative Expenses
	7.2 (7.3) Transfer to Other Units of Government

# EXHIBIT C CONTRACT BUDGET - COST REIMBURSEMENT EVERETT PALLET SHELTER PROJECT

AGENCY NAME:	City of Everett					
CONTRACT PERIOD:	7/1/2023	to _	3/31/2025			
FUNDS AWARDED UNDER CONTR	RACT:					
REVENUE SOURCE	FUNDING PERIOD		AMOUNT	AMENDMENT	TOTAL	AMOUNT
American Rescue Plan Act Coronavirus State and Local Fiscal Recovery Funds	7/1/023-3/31/2025		\$ 250,000		\$	250,000
		$\dashv$				-
TO	 DTAL FUNDS AWARDE	—⊥L ED:	\$ 250,000	\$ -	\$	250,000
MATCH REQUIREMENTS FO	PR CONTRACT:	то	TAL MATCHING	RESOURCES:	\$	-
					-	
OTHER PROGRAM RESOURCES (	Identify):					
SOURCE City COVID Relief Funds (Municipal		_	FUNDING	PERIOD	AM	OUNT
Improvements	Dollars) - Capital	_	7/1/2023	to 3/31/24	\$	250,000
		$\dashv$				
		-	TOTAL OTHER	RESOURCES:	\$	250,000

#### **EXPENDITURES**

CATEGORY	sc	FUND DURCE PA CLFR	т	OTAL	MATCHING RESOURCES	_	THER DURCES
Salaries/Wages				-			
Benefits				-			
Supplies/Minor Equip.				-			
Prof. Services				-			
Postage				-			
Telephone				-			
Mileage/Fares				1			
Meals				-			
Lodging				-			
Advertising				-			
Leases/Rentals				-			
Insurance				-			
Utilities				-			
Repairs/Maint.				-			
Client Flex Funds				-			
Client Rent				-			
Printing				-			
Dues/Subscrip.				-			
Regis./Tuition				-			
Machinery/Equip.	\$	250,000	\$	250,000			
Administration				-			
Indirect				-			
Occupancy				-			
Miscellaneous				-			
Misc. Construction				-		\$	250,000
Acquisition				-			
Relocation				-			
				-			
TOTAL	\$	250,000	\$	250,000	\$ -	\$	250,000

#### **EXPENDITURE NARRATIVE**

A	MOUNT	CATEGORY	NARRATIVE (provide justification describing each category supported with funds awarded under this contract)
\$	250,000	Machinery/Equip.	Funds will be used in the purchase of supporting Pallet structures for Everett Pallet Project (Hygiene/Restroom units, meeting space, etc.)
		Construction - Other Resources	Funds will be used to support the installation (utility connection) of assets acquired and the acquisition of residential units.
\$	250,000	TOTAL	

#### **DETAIL SALARIES / WAGES**

POSITION	FUND SOURCE	% OF TIME TO FUND SOURCE	TOTAL MONTHLY	MONTHLY CHARGE TO FUND SOURCE	# OF MONTHS	TOTAL CHARGE TO FUND SOURCE
N/A						

TOTAL: \$0

NOTE: Above figures may reflect rounding

#### **EXHIBIT D**

#### CERTIFICATION REGARDING LOBBYING

#### AMERICAN RESCUE PLAN ACT OF 2021, SECTION 9901

The undersigned certifies, to the best of his or her knowledge and belief, that:

- (1) No Federal appropriated funds have been paid or will be paid, by or on behalf of the undersigned, to any person for influencing or attempting to influence an officer or employee of any agency, a Member of Congress, an officer or employee of Congress, or an employee of a Member of Congress in connection with the awarding of any Federal contract, the making of any Federal grant, the making of any Federal loan, the entering into of any cooperative agreement, and the extension, continuation, renewal, amendment, or modification of any Federal contract, grant, loan, or cooperative agreement.
- (2) If any funds other than Federal appropriated funds have been paid or will be paid to any person for influencing or attempting to influence an officer or employee of any agency, a Member of Congress, an officer or employee of Congress, or an employee of a Member of Congress in connection with this Federal contract, grant, loan, or cooperative agreement, the undersigned shall complete and submit Standard Form-LLL, "Disclosure Form to Report Lobbying", in accordance with its instructions.
- (3) The undersigned shall require that the language of this certification be included in the award documents for all sub-awards at all tiers (including subcontracts, sub-grants, and contracts under grants, loans, and cooperative agreements) and that all subrecipients shall certify and disclose accordingly.

This certification is a material representation of fact upon which reliance was placed when this transaction was made or entered into. Submission of this certification is a prerequisite for making or entering into this transaction imposed by section 1352, title 31, U.S. Code. Any person who fails to file the required certification shall be subject to a civil penalty of not less than \$10,000 and not more than \$100,000 for each such failure.

I hereby certify that I have read and understood the obligations described above, that the Contractor is in compliance with the above-described nondiscrimination requirements, and by my signature on this document, acknowledge my understanding that any intentional or negligent misrepresentation or falsification of any information submitted in conjunction with this document could subject me to punishment under federal, civil liability and/or in criminal penalties, including but not limited to fine or imprisonment or both under Title 18, United States Code, Sec. 1001, et seq. and punishment under federal law.

AGEN	CY NAME:	City of Everett		
Ву:				
Title:				
Date:				
-				

#### **DISCLOSURE OF LOBBYING ACTIVITIES**

Approved by OMB 1352 0348-0046
Complete this form to disclose lobbying activities pursuant to 31 U.S.C. (See reverse for public burden disclosure.)

1. Type of Federal Action:  a. contract  b. grant c. cooperative agreement d. loan e. loan guarantee f. loan insurance  4. Name and Address of Reporting	2. Status of Federal Action: a. bid/offer/application b. initial award c. post-award		year date of last	ling al change rial Chang	ge Only: quarter Subawardee:	
Prime Subawardee:			r Name and	•		
Subawardee	:					
Tier		Cong	ressional D	<b>istrict</b> . if	known:	
Congressional District, if known:	4c					
6. Federal Department/Agency:			7. Federal Program Name/Description:			
		CFDA Number, if applicable:				
8. Federal Action Number, if known:			9. Award Amount, if known: \$			
10a. Name and Address of Lobbying Entity (if individual, last name, first name, MI ):			<b>ividuals Pe</b> t from No. 10a	_	<b>Services</b> name, first r	6 (including address if name, MI ):
(attach Continuation Sheet(s) SF-LLLA, if r	necessary)					
11. Information requested through this form is authorized by the 31 U.S.C. section 1352. This disclosure of lobbying activities is material representation of fact upon which reliance was placed the tier above when this transaction was made or entered into the tier above when this transaction was made or entered into the disclosure is required pursuant to 31 U.S.C. 1352. This information will be available for public inspection. Any person of fails to file the required disclosure shall be subject to a civil per of not less than \$10,000 and not more than \$100,000 for each such failure.			Signature: Print Name Title: Telephone Date:			
Continuation Sheet(s) SF-LLLA	attached:		Yes		No	
Federal Use Only:						

#### INSTRUCTIONS FOR COMPLETION OF SF-LLL, DISCLOSURE OF LOBBYING ACTIVITIES

This disclosure form shall be completed by the reporting entity, whether sub-awardee or prime Federal recipient, at the initiation or receipt of a covered Federal action, or a material change to a previous filing, pursuant to title 31 U.S.C. section 1352. The filing of a form is required for each payment or agreement to make payment to any lobbying entity for influencing or attempting to influence an officer or employee of any agency, a Member of Congress, an officer or employee of Congress, or an employee of a Member of Congress in connection with a covered Federal action. Complete all items that apply for both the initial filing and material change report. Refer to the implementing guidance published by the Office of Management and Budget for additional information.

- 1. Identify the type of covered Federal action for which lobbying activity is and/or has been secured to influence the outcome of a covered Federal action.
- 2. Identify the status of the covered Federal action.
- 3. Identify the appropriate classification of this report. If this is a follow-up report caused by a material change to the information previously reported, enter the year and quarter in which the change occurred. Enter the date of the last previously submitted report by this reporting entity for this covered Federal action.
- 4. Enter the full name, address, city, State and zip code of the reporting entity. Include Congressional District, if known. Check the appropriate classification of the reporting entity that designates if it is, or expects to be, a prime or sub-award recipient. Identify the tier of the sub-awardee, e.g., the first sub-awardee of the prime is the 1st tier. Sub-awards include but are not limited to subcontracts, sub-grants and contract awards under grants.
- 5. If the organization filing the report in item 4 checks "Sub-awardee," then enter the full name, address, city, State and zip code of the prime Federal recipient. Include Congressional District, if known.
- 6. Enter the name of the Federal agency making the award or loan commitment. Include at least one organizational level below agency name, if known. For example, Department of Transportation, United States Coast Guard.
- 7. Enter the Federal program name or description for the covered Federal action (item 1). If known, enter the full Catalog of Federal Domestic Assistance (CFDA) number for grants, cooperative agreements, loans, and loan commitments.
- 8. Enter the most appropriate Federal identifying number available for the Federal action identified in item 1 (e.g., Request for Proposal (RFP) number; Invitation for Bid (IFB) number; grant announcement number; the contract, grant, or loan award number; the application/proposal control number assigned by the Federal agency). Include prefixes, e.g., "RFP-DE-90-001."
- 9. For a covered Federal action where there has been an award or loan commitment by the Federal agency, enter the Federal amount of the award/loan commitment for the prime entity identified in item 4 or 5.
- 10. (a) Enter the full name, address, city, State and zip code of the lobbying registrant under the Lobbying Disclosure Act of 1995 engaged by the reporting entity identified in item 4 to influence the covered Federal action.(b) Enter the full names of the individual(s) performing services, and include full address if different from 10 (a). Enter Last Name, First Name, and Middle Initial (MI).
- 11. The certifying official shall sign and date the form, print his/her name, title, and telephone number.

According to the Paperwork Reduction Act, as amended, no persons are required to respond to a collection of information unless it displays a valid OMB Control Number. The valid OMB control number for this information collection is OMB No. 0348-0046. Public reporting burden for this collection of information is estimated to average 30 minutes per response, including time for reviewing instructions, searching existing data sources, gathering and maintaining the data needed, and completing and reviewing the collection of information. Send comments regarding the burden estimate or any other aspect of this collection of information, including suggestions for reducing this burden, to the Office of Management and Budget, Paperwork Reduction Project (0348-0046), Washington, DC 20503.

#### **DISCLOSURE OF LOBBYING ACTIVITIES**

Approved by OMB
CONTINUATION SHEET

0348-0046

Describes Father	Davis	
Reporting Entity:	Page	or

Authorized for Local Reproduction

#### **EXHIBIT E**

#### **CIVIL RIGHTS ASSURANCES CERTIFICATION**

#### **AMERICAN RESCUE PLAN ACT OF 2021, SECTION 9901**

- A. The funds provided to Agency are available under section 603 of the Social Security Act, as added by section 9901 of the American Rescue Plan Act.
- B. The Agency understands and acknowledges that:

As a condition of receipt of federal financial assistance from the Department of the Treasury, with monies distributed through Snohomish County, the Agency named below (hereinafter referred to as the "Agency") provides the assurances stated herein. The federal financial assistance may include federal grants, loans and contracts to provide assistance to the Agency's beneficiaries, the use or rent of Federal land or property at below market value, Federal training, a loan of Federal personnel, subsidies, and other arrangements with the intention of providing assistance. Federal financial assistance does not encompass contracts of guarantee or insurance, regulated programs, licenses, procurement contracts by the Federal government at market value, or programs that provide direct benefits.

- C. The assurance apply to all federal financial assistance from or fund made available through the Department of Treasury.
- D. The Civil Rights Restoration Act of 1987 provides that the provisions of the assurances apply to all of the operations of the Agency's program(s) and activity(ies), so long as any portion of the Agency's program(s) or activity(ies) is federally assisted in the manner prescribed above.
- E. The Agency certifies the following:
  - 1. Agency ensures its current and future compliance with Title VI of the Civil Rights Act of 1964, as amended, which prohibits exclusion from participation, denial of the benefits of, or subjection to discrimination under programs and activities receiving federal financial assistance, of any person in the United States on the ground of race, color, or national origin (42 U.S.C. § 2000d et seq.), as implemented by the Department of the Treasury Title VI regulations at 31 CFR Part 22 and other pertinent executive orders such as Executive Order 13166, directives, circulars, policies, memoranda, and/or guidance documents.
  - 2. Agency acknowledges that Executive Order 13166, "Improving Access to Services for Persons with Limited English Proficiency," seeks to improve access to federally assisted programs and activities for individuals who,

because of national origin, have Limited English proficiency (LEP). Agency understands that denying a person access to its programs, services, and activities because of LEP is a form of national origin discrimination prohibited under Title VI of the Civil Rights Act of 1964 and the Department of the Treasury's implementing regulations. Accordingly, Agency shall initiate reasonable steps, or comply with the Department of the Treasury's directives, to ensure that LEP persons have meaningful access to its programs, services, and activities. Agency understands and agrees that meaningful access may entail providing language assistance services, including oral interpretation and written translation where necessary, to ensure effective communication in the Recipient's programs, services, and activities.

- 3. Agency agrees to consider the need for language services for LEP persons when Agency develops applicable budgets and conducts programs, services, and activities. As a resource, the Department of the Treasury has published its LEP guidance at 70 FR 6067. For more information on taking reasonable steps to provide meaningful access for LEP persons, please visit http://www.lep.gov.
- 4. Agency acknowledges and agrees that compliance with the assurances constitutes a condition of continued receipt of federal financial assistance and is binding upon Agency and Agency's successors, transferees, and assignees for the period in which such assistance is provided.
- 5. Agency acknowledges and agrees that it must require any sub-grantees, contractors, subcontractors, successors, transferees, and assignees to comply with assurances 1-4 above, and agrees to incorporate the following language in every contract or agreement subject to Title VI and its regulations between Agency and Agency's subgrantees, contractors, subcontractor, successor, transferees, and assignees:

The sub-grantee, contractor, subcontractor, successor, transferee, and assignee shall comply with Title VI of the Civil Rights Act of 1964, which prohibits recipients of federal financial assistance from excluding from a program or activity, denying benefits of, or otherwise discriminating against a person on the basis of race, color, or national origin (42 U.S.C. § 2000d et seq.), as implemented by the Department of the Treasury's Title VI regulations, 31 CFR Part 22, which are herein incorporated by reference and made a part of this contract (or agreement). Title VI also includes protection to persons with "Limited English Proficiency" in any program or activity receiving federal financial assistance, 42 U.S.C. § 2000d et seq., as implemented by the Department of the Treasury's Title VI regulations, 31 CFR Part 22, and herein incorporated by reference and made a part of this contract or agreement.

6. Agency understands and agrees that if any real property or structure is provided or improved with the aid of federal financial assistance by the

Department of the Treasury, this assurance obligates the Recipient, or in the case of a subsequent transfer, the transferee, for the period during which the real property or structure is used for a purpose for which the federal financial assistance is extended or for another purpose involving the provision of similar services or benefits. If any personal property is provided, this assurance obligates the Agency for the period during which it retains ownership or possession of the property.

- 7. Agency shall cooperate in any enforcement or compliance review activities by the Department of the Treasury of the aforementioned obligations. Enforcement may include investigation, arbitration, mediation, litigation, and monitoring of any settlement agreements that may result from these actions. The Agency shall comply with information requests, on-site compliance reviews and reporting requirements.
- 8. Agency shall maintain a complaint log and inform the Department of the Treasury of any complaints of discrimination on the grounds of race, color, or national origin, and limited English proficiency covered by Title VI of the Civil Rights Act of 1964 and implementing regulations and provide, upon request, a list of all such reviews or proceedings based on the complaint, pending or completed, including outcome. Agency also must inform the Department of the Treasury if Agency has received no complaints under Title VI.
- 9. Agency must provide documentation of an administrative agency's or court's findings of non-compliance of Title VI and efforts to address the non-compliance, including any voluntary compliance or other agreements between the Agency and the administrative agency that made the finding. If the Agency settles a case or matter alleging such discrimination, the Agency must provide documentation of the settlement. If Agency has not been the subject of any court or administrative agency finding of discrimination, please so state.
- 10. If the Agency makes sub-awards to other agencies or other entities, the Agency is responsible for ensuring that sub-recipients also comply with Title VI and other applicable authorities covered in this document State agencies that make sub-awards must have in place standard grant assurances and review procedures to demonstrate that that they are effectively monitoring the civil rights compliance of sub-recipients.
- 11. The United States of America has the right to seek judicial enforcement of the terms of this assurances document and nothing in this document alters or limits the federal enforcement measures that the United States may take in order to address violations of this document or applicable federal law.

I hereby certify that I have read and understood the obligations described above, that the Agency is in compliance with the above-described nondiscrimination requirements, and by my signature on this document, acknowledge my understanding that any intentional or negligent misrepresentation or falsification of any information submitted in conjunction with this document could subject me to punishment under federal, civil liability and/or in criminal penalties, including but not limited to fine or imprisonment or both under Title 18, United States Code, Sec. 1001, et seq. and punishment under federal law.

AGENCY NAME:	City of Everett
Ву:	
Title:	
Date:	



#### **INVOICE - Cost Reimbursement Contracts**

Snohomish County Human Services Department-3000 Rockefeller, M/S 305, Everett, WA 98201

Estimated: Actual: Amount of Payment: \$

Agency Name and Address: Contract #: HCS-23-AR-2108-198

Agency Name and Address:

City of Everett
2930 Wetmore Ave., Suite 8-B
Everett, WA 98201

Contract #:
Project Title:

Contract #:
Project Title:

Contract Manager:
Reporting Period:

HCS-23-AR-2108-198
Everett Pallet Shelter Project

Reporting Period:

To:

AUTHORIZING SIGNATURE: \_\_\_\_\_ DATE: \_\_\_\_\_

SUB	Account Title	Current	Contract To Date	Contract	Budget
OBJ		Expenditures	Expenditures	Budget	Balance
10	Salaries/Wages				
20	Personal Benefits				
30	Supplies				
40	Prof. Services				
42	Postage				
42	Telephone				
43	Mileage				
43	Internet				
43	IT Allocation				
44	Computer Software				
45	Registration/Tuition				
46	Administration				
47	Utilities				
48	Repair/Maintenance				
49	Printing/Copying				
49	Dues/Subscriptions				
49					
64	Machinery/Equipment				
	TOTALS				

CONTRACTING AGENCY MATCHING FUNDS:		REVIEWED FOR PAYMENT:
CURRENT PERIOD:	\$	AUTHORIZED FUND:
CONTRACT TO DATE:	\$	

ATTACH: AGENCY CERTIFICATION FORM



#### SNOHOMISH COUNTY HUMAN SERVICES DEPARTMENT 3000 ROCKEFELLER AVENUE, M/S 305 EVERETT, WA 98201

#### **AGENCY CERTIFICATION FORM**

- 1. **Agency Certification:** I have the authority and approval from the governing body to request reimbursement from Snohomish County from the County's allocation of the CLFR as created in Section 9901 of the American Rescue Plan Act of 2021 ("ARPA") for eligible expenditures included on the corresponding invoice for the reporting period referenced on the Contract Face Page.
- 2. I understand Snohomish County will rely on this certification as a material representation in processing this reimbursement.
- 3. I certify the use of funds submitted for reimbursement from the CLFR under this Agreement were used only to cover those costs in accordance Section 9901 of the American Rescue Plan Act of 2021, the Interim Final Rule at 31 CFR Part 35, and Department of Treasury FAQs and guidance.
- 4. I understand the use of funds pursuant to this certification must adhere to official federal guidance issued. I have reviewed the Section 9901 of the American Rescue Plan Act of 2021, the Interim Final Rule at 31 CFR Part 35 and Department of Treasury (Treasury) FAQs and guidance and certify costs meet the parameters set forth therein. Any funds expended by Agency or its subcontractor(s) in any manner that does not adhere to the Section 9901 of the American Rescue Plan Act of 2021, the Interim Final Rule at 31 CFR Part 35 and Treasury FAQs and guidance shall be returned to the County for return to the Treasury.
- 5. I understand the Agency receiving funds pursuant to this certification shall retain documentation of all uses of the funds, including but not limited to invoices and/or sales receipts in a manner consistent with §200.333 Retention requirements for records of 2 CFR Part 200 Uniform Administrative Requirements, Cost Principles, and Audit Requirements for Federal Awards (Uniform Guidance), Section 200.333- Retention requirements for records. Such documentation shall be produced for the County upon request and may be subject to audit by state and/or federal representatives.
- 6. I understand any funds provided pursuant to this certification cannot be used as a revenue replacement for lower than expected tax or other revenue collections.
- 7. I understand funds received pursuant to this certification cannot be used for expenditures for which the Agency has received any other funding (whether state, federal or private in nature) for the same expense.

I hereby certify that I have read the above certification, and that the information and my statements provided herein by me are true and correct to the best of my knowledge, and by my signature on this document, acknowledge my understanding that any intentional or negligent misrepresentation or falsification of any of the information in this document could subject me to punishment under federal and/or civil liability and/or in criminal penalties, including but not limited to fine or imprisonment or both under Title 18, United States Code, Sec. 1001, et seq. and punishment under federal law.

City of Everett

### **EVERETT** City Council Agenda Item Cover Sheet

**Project title:** 

Council Bill #

**Council President** 

An Ordinance Closing the Special Construction Fund Entitled "Fire Station No.2 and Fire Station No.6 Additions and Alterations", Fund 342, Program 029, as Amended by Ordinance No. 3867-22.

<b>Project:</b> Fire Station No.2 and Fire Station No.6 Additions and Alterations		
Partner/Supplier: Forma Construction and Burton Construction respectively		
<b>Location:</b> 16 <sup>th</sup> Street and Evergreen Way respectively		
Preceding action: Ordinance No. 3867-22		
Fund: Fund 342, Program 029 (CIP 1)		
Fiscal summary statement:		
The source of funds for Fire Station No. 2 and Fire Station No. 6 Additions and Alterations was Capital Improvement Program 1 (CIP-1) in the amount \$1,025,000. The project was completed at a total cost of \$999,276. All expenses have been paid. The remaining balance of \$25,724 will be transferred back to CIP 1.		
The proposed Ordinance will close the special construction fund for the Fire Station No.2 and Fire Station No.6 Additions and Alterations. Ordinance No. 3709-19 provided an original funding amount of \$850,000 for construction costs of the project. Ordinance No. 3804-21 amended the		
original amount for an additional \$80,000 above the amount in the existing Ordinance in order to complete Fire Station No.2's phase 2 construction for the amended total amount of \$930,000.  Ordinance No. 3867-22 amended the funding amount for an additional \$95,000 to capture costs		
due to supply chain issues which required construction materials to be acquired and warehoused minimizing the disruption to Fire Station No.2 construction phase for the amended total amount of \$1,025,000.		
Project summary statement:		
The alterations at Fire Station No.6 accomplished the goal of creating gender neutral restrooms and building renovations and improvements Phase 1 and 2 to Fire Station No.2, correcting deficiencies identified in the Facilities Condition Assessment. Construction was completed in		
early 2023.		
All work has been completed to the satisfaction of the Parks and Facilities Department.		
Recommendation (exact action requested of Council):		
Adopt an Ordinance closing the Special Construction Fund entitled "Fire Station No.2 and Fire		
Station No.6 Additions and Alterations", Fund 342, Program 029, as amended by Ordinance No. 3867-22.		
J00/-22.		



)RF	DINANC	F NO.	
ノハトレ		LINO.	

An ORDINANCE closing the special construction fund entitled "Fire Station No.2 and Fire Station No.6 Additions and Alterations", Fund 342, Program 029, as amended by Ordinance No. 3867-22.

#### WHEREAS,

- **A.** The "Fire Station No. 2 and Fire Station No. 6 Additions and Alterations Project", Fund 342, Program 029 was established to provide design and construction cost for the building renovations and improvements.
- **B.** The City Council has recognized the need to remodel Fire Station No. 2 and Fire Station No. 6. located on 16th Street and Evergreen Way respectively.
- **C.** Ordinance No. 3709-19 was established as Fund 342, Program 029 entitled "Fire Station No. 2 and Fire Station No. 6 Additions and Alterations" to accumulate the construction costs for the project in the amount of \$850,000.
- **D.** Ordinance No. 3804-21 was established amending Ordinance 3709-19 entitled "Fire Station No.2 and Fire Station No.6 Additions and Alterations" Fund 342, Program 029, to accumulate the construction costs for the project in the amount of \$930,000.
- **E.** Ordinance No. 3867-22 was established amending Ordinance 3804-21 entitled "Fire Station No.2 and Fire Station No.6 Additions and Alterations" Fund 342, Program 029, for an increase amount of \$95,000 to accumulate the construction costs for the project.
- **F.** The purpose of the special improvement project has been accomplished.
- **G.** There are neither outstanding obligations of the funds to be paid nor uncollected revenues to be received.

#### NOW, THEREFORE, THE CITY OF EVERETT DOES ORDAIN:

**Section 1.** That the fund entitled "Fire Station No. 2 and Fire Station No. 6 Additions and Alterations", Fund 342, Program 029, as established by Ordinance No. 3867-22, be closed.

<u>Section 2.</u> That the final revenues and expenses for the "Fire Station No. 2 and Fire Station No. 6 Additions and Alterations", Fund 342, Program 029, are as follows:

**REVENUES** 

Fund 342 CIP 1 \$1,025,000 **Total** \$1,025,000

**EXPENSES** 

 Construction
 \$999,276

 Transfer Out
 \$ 25,724

 Total
 \$1,025,000

**Section 3.** That the remaining balance of \$25,724 be transferred to CIP 1.

<u>Section 4.</u> The City Clerk and the codifiers of this Ordinance are authorized to make necessary corrections to this Ordinance including, but not limited to, the correction of scrivener's/clerical errors, references, ordinance numbering, section/subsection numbers, and any internal references.

<u>Section 5.</u> The City Council hereby declares that should any section, paragraph, sentence, clause or phrase of this ordinance be declared invalid for any reason, it is the intent of the City Council that it would have passed all portions of this ordinance independent of the elimination of any such portion as may be declared invalid.

<u>Section 6</u>. The enactment of this Ordinance shall not affect any case, proceeding, appeal or other matter currently pending in any court or in any way modify any right or liability, civil or criminal, which may be in existence on the effective date of this Ordinance.

**Section 7**. It is expressly the purpose of this Ordinance to provide for and promote the health, safety and welfare of the general public and not to create or otherwise establish or designate any particular class or group of persons who will or should be especially protected or benefited by the terms of this Ordinance. It is the specific intent of this Ordinance that no provision or any term used in this Ordinance is intended to impose any duty whatsoever upon the City or any of its officers or employees. Nothing contained in this Ordinance is intended nor shall be construed to create or form the basis of any liability on the part of the City, or its officers, employees or agents, for any injury or damage resulting from any action or inaction on the part of the City related in any manner to the enforcement of this Ordinance by its officers, employees or agents.

Cassie Franklin, Mayor		
ATTEST:		
City Clerk		
PASSED:		
VALID:		
PUBLISHED:		
FFFCTIVE DATE:		

### **EVERETT** City Council Agenda Item Cover Sheet

**Project title:** ORDINANCE Repealing and Replacing Chapter 14.42 of the Everett Municipal Code

Council Bill #	Project: 2024 Adoption of the new Chapter 14.42 EMC related to grease control		
CB 2404-06	Partner/Supplier: n/a		
Agenda dates requested:	Location: n/a		
5/08/24, 5/15/24, 5/22/24	Preceding action: 2008 Adoption of Chapter 14.42 EMC related to grease control under Ord. 3071-08		
Briefing Proposed action 5/8 & 5/15/24 Consent	Fund: n/a		
Action 5/22/24			
Ordinance X	Fiscal summary statement:		
Public hearing Yes x No	This action item has no fiscal impact associated at this time.		
Budget amendment:			
Yes x No	Project summary statement:		
PowerPoint presentation:  Yes x No  Attachments: Ordinance			
<b>Department(s) involved:</b> Public Works	The Grease Control Ordinance was first adopted in 2008. The new Chapter 14.42 EMC addresses the combined sewer overflow requirements contained in Section 8 (S8.B.3) of the City's National Pollutant Discharge Elimination System (NPDES) permit. Additionally,		
Contact person: Jeff Marrs	the updated reflects current practices, regulations, city policies and coding.		
Phone number: (425) 257-8967	The proposed Ordinance was available for a 30-day public comment period ending April 18, 2024.		
Email:			
jmarrs@everettwa.gov	Recommendation (exact action requested of Council):		
	Adopt An Ordinance relating to Uniform Requirements for Food Service Establishments (FSE) and non-FSE FOG dischargers of the Publicly Owned Treatment Works, repealing and replacing Ordinance 3071-08 (Chapter 14.42 EMC).		
Initialed by: RLS			
Department head			
Administration			
Council President			



	ANCE NO.	
UKDIN	AINCE INU.	

An ORDINANCE relating to Grease Control Regulations, repealing Everett Municipal Code Chapter 14.42 and adopting a new Everett Municipal Code Chapter 14.42.

#### WHEREAS,

- A. The City Council deems it necessary to set forth uniform requirements for certain users of the City's publicly owned treatment works in order to comply with all applicable State and Federal laws and regulations.
- B. The City's grease control regulations, codified at Chapter EMC 14.42, were last amended in 2008, and the City recognizes the need to update these regulations.
- C. Washington State's Department of Ecology and the United States Environmental Protection Agency have reviewed and approved the requirements for users that discharge fats, oils and grease to the City's publicly owned treatment works contained in this Ordinance.

#### NOW, THEREFORE, THE CITY OF EVERETT DOES ORDAIN:

**Section 1.** Chapter 14.42 EMC is hereby repealed in its entirety. For purposes of reference, chapter 14.42 EMC was adopted under Ord. 3071-08, 2008.

<u>Section 2.</u> A new chapter 14.42 EMC is hereby added to the Everett Municipal Code as set forth in <u>Exhibit A</u>. Permits issued under chapter 14.42 EMC adopted under Ord. 3071-08 shall hereafter be subject to the provisions of the new chapter 14.42 EMC attached as <u>Exhibit A</u> hereto.

<u>Section 3.</u> The City Clerk and the codifiers of this Ordinance are authorized to make necessary corrections to this Ordinance including, but not limited to, the correction of scrivener's/clerical errors, references, ordinance numbering, section/subsection numbers, and any internal references.

<u>Section 4</u>. The City Council hereby declares that should any section, paragraph, sentence, clause or phrase of this ordinance be declared invalid for any reason, it is the intent of the City Council that it would have passed all portions of this ordinance independent of the elimination of any such portion as may be declared invalid.

<u>Section 5</u>. The enactment of this Ordinance shall not affect any case, proceeding, appeal or other matter currently pending in any court or in any way modify any right or liability, civil or criminal, which may be in existence on the effective date of this Ordinance.

<u>Section 6</u>. It is expressly the purpose of this Ordinance to provide for and promote the health, safety and welfare of the general public and not to create or otherwise establish or designate any particular class or group of persons who will or should be especially protected or benefited by the terms of this Ordinance. It is the specific intent of this Ordinance that no provision or any term used in this Ordinance is intended to impose any duty whatsoever upon the City or any of its officers or employees. Nothing contained in this Ordinance is intended nor shall be construed to create or form the basis of any liability on the part of the City, or its officers, employees or agents, for any injury or damage resulting from any action or inaction on the part of the City related in any manner to the enforcement of this Ordinance by its officers, employees or agents.

Cassie Franklin, Mayor	
ATTEST:	
City Clerk	
PASSED:	
VALID:	
PUBLISHED:	
EFFECTIVE DATE:	

## EXHIBIT A TO GREASE CONTROL REGULATIONS ORDINANCE

#### **CHAPTER 14.42 EMC**

Reference Table for Codification			
Section Number in Attached Text	Code Section Number for Codification		
1.1	14.42.010		
1.2	14.42.020		
1.3	14.42.030		
1.4	14.42.040		
2.1	14.42.050		
2.2	14.42.060		
2.3	14.42.070		
2.4	14.42.080		
2.5	14.42.090		
2.6	14.42.100		
2.7	14.42.110		
2.8	14.42.120		
3	14.42.130		
3.1	14.42.140		
3.2	14.42.150		
3.3	14.42.160		
3.4	14.42.170		
3.5	14.42.180		
3.6	14.42.190		
3.7	14.42.200		
4.1	14.42.210		
4.2	14.42.220		
4.3 (New Section)	14.42.225		
5	14.42.230		
6.1	14.42.240		
6.2	14.42.250		
6.3	14.42.260		
6.4	14.42.270		
6.5	14.42.280		
6.6	14.42.290		
6.7	14.42.300		
6.8	14.42.310		
6.9	14.42.320		
7.1	14.42.330		
7.2	14.42.340		
7.3	14.42.350		
7.4	14.42.360		
8.1	14.42.370		
8.2	14.42.380		

Reference Table for Codification			
Section Number in Attached Text	Code Section Number for Codification		
8.3	14.42.390		
8.4	14.42.400		
8.5	14.42.410		
8.6	14.42.420		
9.1	14.42.430		
9.2	Not Used		
9.3	14.42.450		
10.1	14.42.460		
10.2	14.42.470		

### **Grease Control**

	DN 1 - GENERAL PROVISIONS	
1.1	Purpose and Policy	
1.2	Administration	
1.3	Definitions	
1.4	Abbreviations	
SECTIO	ON 2 - GENERAL REQUIREMENTS	10
2.1	Prohibited Discharge Standards	10
2.2	State Requirements	13
2.3	Local Limits	13
2.4	City's Right of Revision	13
2.5	Special Agreement	14
2.6	Dilution	
2.7	Pretreatment Facilities	
2.8	Additional Pretreatment Measures	
	ON 3 - GREASE REMOVAL SYSTEM REQUIREMENTS	16
3.1	New Construction	
3.2	Existing Users	
3.3	Grease Removal System Maintenance	
3.4	Grease Removal System Additives	
3.5	Solids Interceptor	
3.6	Grease Removal System Sizing	
3.7	Flow Controls	
	DN 4 - RECORD KEEPING/REPORTING REQUIREMENTS	
4.1	Record Keeping	
4.2	Reporting	
4.3	Electronic Reporting	
	DN 5 - CONFIDENTIAL INFORMATION	
	ON 6 - ADMINISTRATIVE ENFORCEMENT REMEDIES	
6.1	Notification of Violation (Notice of Violation, NOV)	
6.2	Consent Orders	
6.3	Show Cause Hearing	
6.4	Compliance Orders	
6.4 6.5	Cease and Desist Orders	
6.6	Administrative Fines	
6.7	Emergency Suspensions (Non Emergency)	
6.8	Termination of Discharge (Non-Emergency)	
6.9	Appeal Procedures	23
	DN 7 - JUDICIAL ENFORCEMENT REMEDIES	
7.1	Injunctive Relief	
7.2	Civil Penalties	
7.3	Criminal Prosecution	
7.4	Remedies Non-exclusive	
	ON 8 - SUPPLEMENTAL ENFORCEMENT ACTION	
8.1	Performance Bonds	
8.2	Financial Assurances	_
8.3	Service Severance	
8.4	Public Nuisances	
8.5	Contractor Listing	
8.6	Publication of Violations and/or Enforcement Actions.	
SECTION 9 - AFFIRMATIVE DEFENSES TO DISCHARGE VIOLATIONS		
9.1	Upset	
9.2	Not Used	
9.2	Bypass	34

10.1 Pretreatment Charges and Fees	SECTION	10 - MISCELLANEOUS PROVISIONS	.36
	10.1	Pretreatment Charges and Fees	.36
		Non-Liability	

#### **SECTION 1 - GENERAL PROVISIONS**

#### 1.1 Purpose and Policy

This ordinance sets forth uniform requirements for Food Service Establishments (FSEs) and Non-FSE FOG Dischargers (NFDs), collectively Fats, Oils and Greases (FOG) dischargers, that discharge wastewater to the City of Everett Publicly Owned Treatment Works (POTW). The objectives of this ordinance are:

- A. To prevent the introduction of pollutants into the POTW by FOG dischargers that will interfere with the operation of the POTW;
- B. To prevent the introduction of pollutants into the POTW by FOG dischargers that will cause Combined Sewer Overflows (CSOs) or Sanitary Sewer Overflows (SSOs);
- C. To protect the environment from pollution caused by CSOs and SSOs;
- D. To adequately regulate the sale and use of additives added to Grease Removal Systems (GRSs).
- E. <u>To assist users with efficient operation and maintenance of pretreatment equipment and facilities to maintain consistent compliance with the ordinance.</u>

This ordinance shall apply to all non-residential users as defined herein as FSEs or NFDs, except those defined as Significant Industrial Users under 40CFR403, that discharge or may discharge excess Fats, Oils and Greases (FOG) to the POTW. The ordinance authorizes inspection and regulation of these users; authorizes monitoring, compliance, and enforcement activities; establishes administrative review procedures; requires user record keeping; provides for the issuance of control mechanisms if deemed necessary by the Director; and provides for the setting of fees for the equitable distribution of costs resulting from the program established herein if deemed necessary by the Director.

#### 1.2 Administration

Except as otherwise provided herein, the Director shall administer, implement, and enforce the provisions of this ordinance. Any powers granted to or duties imposed upon the Director may be delegated by the Director to other City of Everett personnel. The Director may create administrative guidelines to implement the provisions of this ordinance.

#### 1.3 Definitions

Unless a provision explicitly states otherwise, the following terms and phrases, as used in this ordinance, shall have the meanings hereinafter designated.

- A. <u>Accessible</u>. Accessible, when applied to required pretreatment monitoring or treatment equipment, shall mean direct access without the necessity of removing any panel, door, landscaping, vehicle, equipment, materials, use of a ladder, or other similar obstruction.
- B. <u>Additive</u>. Any material, in any physical form, put into a Grease Removal System (GRS) or any drain lines or appurtenances discharging to a GRS intended in any way to modify the operation of the GRS or its captured and retained FOG.
- C. <u>Administrative Penalty (fine)</u>. A punitive monetary charge unrelated to treatment cost, which is assessed by the Director rather than a court.
- D. <u>Approval Authority</u>. The state of Washington Department of Ecology.
- E. Authorized Representative of the User.
  - (1) If the user is a corporation:
    - (a) The president, secretary, treasurer, or a vice-president of the corporation in charge of a principal business function, or any other person who performs similar policy or decision-making functions for the corporation; or
    - (b) The manager of one or more facilities, provided the manager is authorized to make management decisions which govern the operation of the regulated facility; initiate and direct measures to assure long-term compliance with environmental laws and regulations; can ensure that the necessary systems are established or actions taken to gather complete and accurate information for required reports; and where authority to sign documents has been assigned or delegated to the manager in accordance with corporate procedures;
  - (2) If the user is a partnership or sole proprietorship: a general partner or proprietor, respectively;
  - (3) If the user is a Federal, State, or local governmental facility: a director or highest official appointed or designated to oversee the operation and performance of the activities of the government facility, or authorized designee.
  - (4) The individuals described in paragraphs 1 through 3 above may designate another authorized representative if the authorization is in writing, the authorization specifies the individual or position responsible for the overall operation of the facility from which the discharge originates or having overall responsibility for environmental matters for the company, and the written authorization is submitted to the City.
- F. <u>Automatic Grease Removal System (AGRS)</u>. A GRS that has provision to automatically remove separated FOG and/or settled solids from the tank and collect them for disposal.

- G. <u>Best Management Practices (BMPs).</u> means schedules of activities, prohibitions of practices, maintenance procedures, and other management practices to prevent or reduce the pollution of "waters of the United States." BMPs also include treatment requirements, operating procedures, and practices to control plant site runoff, spillage or leaks, sludge or waste disposal, or drainage from raw material storage.
- H. <u>Biochemical Oxygen Demand (BOD)</u>. The quantity of oxygen utilized in the biochemical oxidation of organic matter amenable to measurement by the methods described in <u>Standard Methods for the examination of Water and Wastewater</u>, latest approved edition or other methods approved by 40CFR136.
- I. <u>Chemical Oxygen Demand (COD)</u>. A measure of the oxygen consuming capacity of inorganic and organic matter present in wastewater amenable to measurement by the methods described in <u>Standard Methods for the examination of Water and Wastewater</u>, latest approved edition or other methods approved by 40CFR136. COD is expressed as the amount of oxygen consumed from a chemical oxidant in mg/L during a specific test.
- J. <u>City</u>. The City of Everett, Washington.
- K. Commissary Kitchen. A commercial kitchen facility used by third parties that provides potable water supply, safe food storage and preparation equipment, and fixtures and equipment required for proper wastewater disposal.
- L. <u>Combined Sewer Overflow (CSO)</u>. Any unplanned discharge from the combined sewer system.
- M. <u>Composite Sample</u>. The sample resulting from the combination of individual wastewater samples taken at selected intervals based on an increment of either flow or time.
- N. Day. Day shall be defined as a calendar day.
- O. Director. The Public Works Director, or the Director's duly authorized representative.
- P. <u>Discharge Authorization</u>. A wastewater discharge permit authorizing users to discharge wastewater to the Everett POTW. These permits are for non-Significant and/or non-Categorical Industrial Users requiring a control mechanism.
- Q. <u>Domestic Sewage</u>. Domestic sewage means the liquid and water borne wastes derived from ordinary living processes, free from industrial wastes, and of such character to permit satisfactory disposal, without special treatment, into the POTW.
- R. <u>Domestic User (Residential User)</u>. Any person who contributes, causes, or allows the contribution of wastewater into the City POTW that is of a similar volume and/or chemical make-up as that of a residential dwelling unit. Discharges from a residential dwelling unit typically include up to 100 gallons per capita per day at 300 mg/L of BOD and TSS.

- S. <u>Environmental Protection Agency (EPA)</u>. The US Environmental Protection Agency or, where appropriate, the Regional Water Management Division Director, or other duly authorized official of said agency.
- T. <u>Existing User</u>. An existing user is defined as any user which is discharging wastewater prior to the effective date of this ordinance.
- U. <u>Fats, Oils and Grease (FOG)</u>. The term fats, oils, and grease shall mean those components of wastewater amenable to measurement by the methods described in <u>Standard Methods for the examination of Water and Wastewater</u>, latest approved edition or other methods approved by 40CFR136 (i.e., EPA Method 1664 (HEM; SGT/HEM)). For the purposes of this ordinance, the term Fats, Oils and Grease shall include polar fats, oils, and grease and other components extracted from wastewater by these methods, excluding the non-polar fraction.
- V. <u>Food Service Establishment (FSE)</u>. Any establishment, commercial or noncommercial, primarily engaged in the preparing, serving, or otherwise making available for consumption foodstuffs in or on a receptacle that requires washing more than two (2) days per week and that discharges to the POTW.
- W. Ghost Kitchen. A functioning FSE where food is prepared for another FSE, MFU, or NFD.
- X. <u>Grab Sample</u>. A sample which is taken from a wastestream on a one-time basis without regard to the flow in the wastestream and without consideration of time.
- Y. Gravity <u>Grease Interceptor/Interceptor/Interceptor-style GRS</u>. Any relatively large in ground or above-ground tank, generally, but not always, of precast concrete, with internal plumbing and baffling intended to act as a GRS or AGRS to serve one or more fixtures and which shall be remotely located. GGIs shall meet the IAPMO/ANSI Z1001 Standard.
- Z. Grease/Greases. See FOG.
- AA. <u>Grease Impact Area</u>. Any area of the POTW collection system where grease deposits originating at FSEs and/or NFDs create maintenance requirements exceeding normal sewer maintenance. These areas will be identified by the Director and updated as needed.
- AB. <u>Grease Removal System (GRS)</u>. Any device designed for, and intended for, separating, collecting, and removing waterborne FOG and settleable solids prior to discharging to the POTW. This includes any AGRS.
- AC. <u>High Strength Waste</u>. Any waters or wastewater having a concentration of BOD or Total Suspended Solids in excess of 300 mg/L, or having a concentration of Fats, Oil and Grease in excess of 50 mg/L.

.

- AD. Hydromechanical Grease Removal System/<u>Grease Trap//Trap-style GRS</u>. Any relatively small appurtenance, generally, but not always, made of cast iron, poly/plastic, or fabricated steel, with internal configuration and internal or external flow control, intended to function as a GRS or AGRS. All hydromechanical trap-style grease removal systems shall be PDI or IAPMO approved.
- AE. <u>Interference</u>. A discharge which alone or in conjunction with a discharge or discharges from other sources, either: (1) inhibits or disrupts the POTW, its treatment processes or operations; (2) inhibits or disrupts its biosolids (sludge) processes, use or disposal; or (3) is a cause of a violation of the City's NPDES permit or of the prevention of sewage sludge use or disposal in compliance with any of the following statutory/ regulatory provisions or permits issued thereunder: Section 405 of the Clean Water Act; the Solid Waste Disposal Act (SWDA), including Title II commonly referred to as the Resource Conservation and Recovery Act (RCRA); any State regulations contained in any State sludge management plan prepared pursuant to Subtitle D of the SWDA; the Clean Air Act; the Toxic Substances Control Act; and the Marine Protection, Research, and Sanctuaries Act.
- AF. <u>Liquid Waste</u>. Liquid waste is the discharge from any fixture, appliance, or appurtenance in connection with a plumbing system which does not receive fecal matter.
- AG. <u>Maximum Allowable Discharge Limit</u>. The maximum concentration (or loading) of a pollutant allowed to be discharged at any time.
- AH. <u>Mobile Food Unit (MFU)</u>. An operation where the food service establishment is readily movable; this includes mobile food trucks, trailers, and carts.
- AI. North American Industry Classification System (NAICS). A classification pursuant to the North American Industry Classification System Manual issued by the United States Office of Management and Budget.
- AJ. New User. A new user is defined as a user that applies to the City for a new building permit or any person who occupies an existing building and plans to discharge wastewater to the City's collection system after the effective date of this ordinance. Any person that buys an existing facility that is discharging non-domestic wastewater will be considered an existing user if no substantial remodel is made in the operation.
- AK. <u>Non-FSE FOG Discharger (NFD)</u>. Any establishment, such as a church, synagogue, worship hall, banquet facility, or meeting space, with a commercial-style kitchen that is used for preparing, serving, or otherwise making available for consumption foodstuffs in or on a receptacle that requires washing two days a week or less and that discharges to the POTW.
- AL. <u>Pass Through</u>. A discharge which exits the POTW into waters of the United States in quantities or concentrations which, alone or in conjunction with a discharge or discharges from other sources, is a cause of a violation of any requirement of the City's NPDES permit (including an increase in the magnitude or duration of a violation).

- AM. <u>Permittee</u>. A person or user issued a wastewater discharge permit, or discharge authorization.
- AN. <u>Person</u>. Any individual, partnership, co-partnership, firm, company, corporation, association, joint stock company, trust, estate, governmental entity, or any other legal entity, or their legal representatives, agents, or assigns. This definition includes all Federal, State, or local governmental entities.
- AO. <u>pH.</u> A measure of the acidity or alkalinity of a substance, expressed in standard units measured using the methods described in <u>Standard Methods for the Examination of Water</u> and Wastewater, current approved edition.
- AP. <u>Pollutant</u>. Any dredged spoil, solid waste, incinerator residue, sewage, garbage, sewage sludge, munitions, medical wastes, chemical wastes, biological materials, radioactive materials, heat, wrecked or discharged equipment, rock, sand, cellar dirt, agricultural and industrial wastes, and the characteristics of the wastewater [i.e., pH, temperature, TSS, turbidity, color, BOD, Chemical Oxygen Demand (COD), toxicity, or odor].
- AQ. Pretreatment. The reduction of the amount of pollutants, the elimination of pollutants, or the alteration of the nature of pollutant properties in wastewater prior to (or in lieu of) introducing such pollutants into the POTW. This reduction or alteration can be obtained by physical, chemical, or biological processes; by process changes; or by other means (except by diluting the concentration of the pollutants unless allowed by an applicable pretreatment standard).
- AR. <u>Pretreatment Requirements</u>. Any substantive or procedural requirement related to pretreatment imposed on a user, other than a pretreatment standard.
- AS. <u>Pretreatment Standards or Standards</u>. Pretreatment standards shall mean prohibited discharge standards, categorical pretreatment standards, and local limits established by the City (POTW).
- AT. <u>Prohibited Discharge Standards or Prohibited Discharges</u>. Absolute prohibitions against the discharge of certain substances; these prohibitions appear in Sections 2.1 (A) and (B) of this ordinance.
- AU. Publicly Owned Treatment Works (POTW). A treatment works, as defined by Section 212 of the Act (33 USC 1292) which is owned by the City. This definition includes all devices facilities, or systems used in the collection, storage, treatment, recycling, and reclamation of sewage or industrial wastes of a liquid nature and any conveyances which convey wastewater to a treatment plant. The term also means the City of Everett Water Pollution Control Facility.
- AV. Sanitary Flow: Sewage.

- AW. <u>Sanitary Sewer Overflow (SSO)</u>. Any unplanned discharge from the separate sanitary sewer system.
- AX. <u>Septic Tank Waste</u>. Any domestic and/or residential sewage from holding tanks such as vessels, chemical toilets, campers, trailers, and septic tanks.
- AY. <u>Settleable Solids</u>. The solids that sinks to the bottom of water, wastewater, or other liquid, and which is amenable to measurement by the methods described in <u>Standard Methods for the examination of Water and Wastewater</u>, latest approved edition or other methods approved by 40CFR136.
- AZ. <u>Sewage</u>. Human excrement and gray water (household showers, dishwashing operations, food preparation, etc.)
- BA. <u>Sewer</u>. Any pipe, conduit, ditch, or other device used to collect and transport sewage from the generating source.
- BB. Shall, May. "Shall" is mandatory, "may" is permissive.
- BC. <u>Solids Interceptor</u>. Any device designed for, and intended for, separating, collecting, and removing waterborne solids prior to being discharged to a GRS. A Solids Interceptor must remove solids greater than 1/8 inch.
- BD. <u>Standard Industrial Classification (SIC) Code</u>. A classification pursuant to the Standard Industrial Classification Manual issued by the United States Office of Management and Budget.
- BE. State. The State of Washington.
- BF. <u>Storm Water</u>. Any flow occurring during or following any form of natural precipitation, and resulting from such precipitation, including snowmelt.
- BG. <u>Substantial Remodel</u>. Any modification to an existing FSE or NFD kitchen that involves changes to 50% or more of the floor area of the kitchen.
- BH. Total Suspended Solids. The total suspended matter that floats on the surface of, or is suspended in, water, wastewater, or other liquid, and which is removable by laboratory filtering and is amenable to measurement by the methods described in Standard Methods for the examination of Water and Wastewater, latest approved edition or other methods approved by 40CFR136.
- BI. <u>Toxic Pollutant</u>. One of the pollutants, or combination of those pollutants, listed as toxic in regulations promulgated by EPA under Section 307 (33 USC 1317) of the Act, or other pollutants as may be promulgated.
- BJ. Treatment Plant Effluent. The discharge from the POTW into waters of the United States.

- BK. <u>User</u>. For the purposes of this ordinance a user is any FSE or NFD. This shall not include domestic user as defined herein.
- BL. <u>Wastewater</u>. Liquid and water-carried industrial wastes and sewage from residential dwellings, commercial buildings, industrial and manufacturing facilities, and institutions, whether treated or untreated, which are contributed to the POTW.
- BM. Wastewater Discharge Permit (Industrial Wastewater Discharge Permit, Discharge Permit, Discharge Authorization). An authorization or equivalent control document issued by the City to users discharging wastewater to the POTW. The permit may contain appropriate pretreatment standards and requirements as set forth in this ordinance.
- BN. <u>Wastewater Treatment Plant or Treatment Plant or Pollution Control Facility</u>. That portion of the POTW which is designed to provide treatment of municipal sewage and industrial waste.

The use of the singular shall be construed to include the plural and the plural shall include the singular as indicated by the context of its use.

# 1.4 Abbreviations

The following abbreviations shall have the designated meanings:

AGRS	Automatic Grease Removal System
AKART	All Known Available and Reasonable Technology
BMPs	.Best Management Practices
BOD	.Biochemical Oxygen Demand
CFR	Code of Federal Regulations
COD	Chemical Oxygen Demand
CSO	Combined Sewer Overflow
EPA	.US Environmental Protection Agency
FOG	Fats, Oils, and Grease
FSE	Food Service Establishment
GPD	gallons per day
GGI	. Gravity Grease Interceptor
GRS	Grease Removal System
HGI	. Hydromechanical Grease Interceptor (aka "Trap-style" GRS)
L	liter
mg	.milligrams
mg/L	.milligrams per liter
NFD	Non-FSE FOG Discharger
O&M	Operation and Maintenance
POTW	Publicly Owned Treatment Works
SSO	Sanitary Sewer Overflow
TSS	Total Suspended Solids

## **SECTION 2 - GENERAL REQUIREMENTS**

## 2.1 Prohibited Discharge Standards

- A. General Prohibitions: No user shall introduce or cause to be introduced into the POTW any pollutant or wastewater which causes pass through or interference. These general prohibitions apply to all users of the POTW whether or not they are subject to categorical pretreatment standards or any other National, State, or local pretreatment standards or requirements.
- B. Specific Prohibitions: No user shall introduce or cause to be introduced into the POTW the following pollutants, substances, or wastewater:
  - (1) Pollutants which create a fire or explosive hazard in the POTW, including, but not limited to, wastestreams with a closed-cup flashpoint of less than 140°F (60°C) using the test methods specified in 40 CFR 261.21;
  - (2) Wastewater having a pH less than 5.0 s.u. or more than 11.0 s.u. (unless in compliance with Section 2.5 of this Ordinance) or otherwise causing corrosive structural damage to the POTW or equipment;
  - (3) Solid or viscous substances in amounts which will cause obstruction of the flow in the POTW resulting in interference but in no case solids greater than one quarter inch (1/4");
  - (4) Pollutants, including oxygen-demanding pollutants (BOD, COD, etc.), released in a discharge at a flow rate and/or pollutant concentration which, either singly or by interaction with other pollutants, will cause interference with the POTW;
  - (5) Wastewater having a temperature which will inhibit biological activity in the treatment plant resulting in interference, but in no case wastewater which causes the temperature at the introduction into the treatment plant to exceed 104°F (40°C) unless the Approval Authority, upon the request of the POTW, approves alternate temperature limits;
  - (6) Petroleum oil, non-biodegradable cutting oil, or products of mineral oil origin, in amounts that will cause interference or pass through;
  - (7) Pollutants which result in the presence of toxic gases, vapors, or fumes within the POTW in a quantity that may cause acute worker health and safety problems;
  - (8) Trucked or hauled pollutants (except Domestic Sewage or Septic Tank Wastes) unless authorized by the director, and at discharge points designated by the City.

- (9) Noxious or malodorous liquids, gases, solids, or other wastewater which, either singly or by interaction with other wastes, are sufficient to create a public nuisance or a hazard to life, or to prevent entry into the sewers for maintenance or repair;
- (10) Wastewater which imparts color which cannot be removed by the treatment process, such as, but not limited to, dye wastes and vegetable tanning solutions, which consequently imparts color to the treatment plant's effluent, thereby violating the City's NPDES permit. Color (in combination with turbidity) shall not cause the treatment plant effluent to reduce the depth of the compensation point for photosynthetic activity by more than ten percent (10%) from the seasonably established norm for aquatic life;
- (11) Wastewater containing any radioactive wastes or isotopes except as specifically approved by the Director in compliance with applicable State or Federal regulations;
- (12) Storm water, surface water, ground water, artesian well water, roof runoff, subsurface drainage, swimming pool drainage, condensate, deionized water, non-contact cooling water, and unpolluted wastewater, unless specifically authorized by the Director;
- (13) Any sludges, screenings, or other residues from the pretreatment of industrial or commercial wastes or from industrial or commercial processes, or residues from cleaning wetwells or sewers, except as authorized by the Director;
- (14) Medical wastes, except as specifically authorized by the Director;
- (15) Wastewater causing, alone or in conjunction with other sources, the treatment plant's effluent to fail a toxicity test;
- (16) Detergents, surface-active agents, or other substances in amounts which may cause excessive foaming in the POTW;
- (17) Any liquid, solids, or gases which by reason of their nature or quantity are, or may be, sufficient either alone or by interaction with other substances to cause fire or explosion or be injurious in any other way to the POTW or to the operation of the POTW. At no time shall two (2) successive readings on an explosion meter, at the point of discharge into the system (or at any point in the system), be more than five (5%) per cent nor any single reading over ten (10%) per cent of the lower explosive limit (LEL) of the meter.

- (18) Animal guts or tissues, paunch manure, bones, hair, hides or fleshings, entrails, whole blood, feathers, ashes, cinders, sand, spent lime, stone or marble dusts, metal, glass, straw, shavings, grass clippings, rags, spent grains, spent hops, waste paper, wood, plastics, gas, tar asphalt residues, residues from refining or processing of fuel or lubricating oil, mud, glass grinding or polishing wastes, or other similar wastes in amounts that cause Interference in the collections system and at the POTW.
- (19) Any substance which will cause the POTW to violate its NPDES and/or other disposal system permits.
- (20) Any wastewater, which in the opinion of the Director can cause harm either to the sewers, sewage treatment process, or equipment; have an adverse effect on the receiving stream; or can otherwise endanger life, limb, public property, or constitute a nuisance, unless allowed under special agreement by the Director (except that no special waiver shall be given from categorical pretreatment standards).
- (21) The contents of any tank or other vessel owned or used by any person in the business of collecting or pumping sewage, effluent, septage, or other wastewater unless said person has first obtained testing and approval as may be generally required by the City of Everett and paid all fees assessed for the privilege of said discharge.
- (22) Any hazardous or dangerous wastes as defined in rules published by the State of Washington (WAC 173-303), in EPA rules 40 CFR Part 261, or Hazardous Waste Pharmaceuticals.
- (23) Persistent pesticides and/or pesticides regulated by the Federal Insecticide Fungicide Rodenticide Act (FIFRA)
- (24) Any slug load.
- (25) Any substance which may cause the POTW's effluent or treatment residues, sludges, or scums to be unsuitable for reclamation and reuse, or to interfere with the reclamation process.
- (26) Fats, oils and grease in amounts that may cause obstructions or maintenance problems in the collection/conveyance system, or interference in the POTW.
- (27) The use of the treatment and controls located at the POTW for wastewater treatment required by a National Emission Standards for Hazardous Air Pollutants for Source Categories (NESHAP) under 40 CFR Part 63 is prohibited. The discharge of any untreated wastewater regulated by a NESHAP also is prohibited. The POTW does not and will not accept a NESHAP regulated waste stream nor provide treatment or controls as an

agent for any Industrial User within the meaning of 40 CFR Part 63, including but not limited to 40 CFR § 63.1595.

(28) Any toxic substances in amounts exceeding standards promulgated by the United States Environmental Protection Agency pursuant to Section 307(a) of the Act and the EPA's priority pollutant list.

Pollutants, substances, or wastewater prohibited by this section shall not be processed or stored in such a manner that they are likely to be discharged to the POTW unless the user has in place an accidental spill prevention plan (ASPP)/slug control plan.

## 2.2 State Requirements

State requirements and limitations on discharges to the POTW shall be met by all users which are subject to such standards in any instance in which they are more stringent than federal requirements and limitations, or those in this ordinance or other applicable ordinances.

### 2.3 Local Limits

The following pollutant limits are established to protect against pass through and interference. No person shall discharge wastewater containing in excess of the following daily maximum allowable discharge limit.

nonpolar fats, oils, 200 mg/L and grease (NPFOG)

The above limit applies at the point where the wastewater is discharged to the POTW (end of the pipe).

The City Council authorizes the Director to revise local limits.

# 2.4 City's Right of Revision

The City reserves the right to establish, by ordinance or in wastewater discharge permits, more stringent standards or requirements on discharges to the POTW.

## 2.5 Special Agreement

The City reserves the right to enter into special agreements with permitted users setting out special terms under which they may discharge to the POTW. In no case will a special agreement waive compliance with a categorical pretreatment standard or federal pretreatment requirement. However, the user may request a net gross adjustment to a categorical standard in accordance with 40 CFR 403.15. They may also request a variance from the categorical pretreatment standard from the Approval Authority in accordance with 403.13.

## 2.6 Dilution

A user shall not increase the use of process water, or in any way attempt to dilute a discharge, as a partial or complete substitute for adequate treatment to achieve compliance with an applicable pretreatment standard or requirement unless expressly authorized by an applicable pretreatment standard or requirement. The Director may impose mass limitations on users that the Director believes may be using dilution to meet applicable pretreatment standards or requirements, or in other cases when the imposition of mass limitations is appropriate.

### 2.7 Pretreatment Facilities

A. General: Users shall provide all known, available, and reasonable methods of prevention, control, and treatment (AKART) as required to comply with this ordinance and shall achieve compliance with all applicable pretreatment standards and requirements set out in this ordinance within the time limitations specified by the EPA, the State, or the Director, whichever is more stringent.

Any facilities required to pretreat wastewater to a level acceptable to the City shall be provided, operated, and maintained at the user's expense.

In addition, the Director may establish Best Management Practices (BMPs) for particular groups of users. These BMPs may include but are not limited to types or methods of pretreatment technology to be used, methods of source control, minimum maintenance requirements, spill prevention, pollution prevention, and housekeeping practices (e.g. in trash enclosures), or other requirements as deemed necessary.

B. If a failure to maintain any GRS results in partial or complete blockage of the building sewer, private sewer system discharging to the City Sewer System, or other parts of the City Sewer System, or adversely affects the treatment or transmission capabilities of the POTW, or requires excessive maintenance by the City, or poses a possible health hazard, the discharger responsible for the facilities

shall be subject to the remedies herein, including cost recovery, enforcement and penalties.

## 2.8 Additional Pretreatment Measures

- A. When deemed necessary, the Director may require users to restrict their discharge during peak flow periods, designate that certain wastewater be discharged only into specific sewers, relocate and/or consolidate points of discharge, separate sewage wastestreams from industrial wastestreams, and such other conditions as may be necessary to protect the POTW and determine the user's compliance with the requirements of this ordinance.
- B. When determined necessary by the Director, each user discharging into the POTW, shall install and maintain, on the user's property and at the user's expense, a suitable storage and flow-control facility to ensure equalization of flow. The Director may require the facility to be equipped with alarms and a rate of discharge controller, the regulation of which shall be determined by the Director. A wastewater Discharge Authorization (DA) may be issued solely for flow equalization.

# **SECTION 3 – GREASE REMOVAL SYSTEM REQUIREMENTS**

All FSEs and NFDs shall have an adequate grease removal system installed and exercise proper kitchen best management practices to ensure that excess concentrations of FOG are not discharged to the POTW.

All Mobile Food Units (MFUs) which, in the judgement of the Director, generate FOG-bearing wastewater requiring pretreatment, shall use a commissary kitchen facility equipped with an approved grease removal system, and exercise proper best management practices to ensure that excess amounts of FOG are not discharged to the POTW.

#### 3.1 New Construction

All new construction FSEs and NFDs shall submit kitchen fixture plan views and kitchen waste plans showing all potential grease discharging lines, all GRSs, and connecting piping to the Director or designee for approval prior to construction. The plumbing shall be installed in accordance with the approved plans. Failure to submit plans or construct in accordance with approved plans is a violation of this ordinance.

- A. All new single occupancy Food Service Establishment buildings shall be constructed with properly sized interceptor-style grease removal systems. Where bona-fide space or gradient limitations make the use of an interceptor-style GRS impractical, and with prior approval of the Director, a FSE may install properly sized hydromechanical trap-style GRSs. All kitchen drains and any other drains that may carry grease-laden waste shall be connected to this GRS (except the dishwasher if a hydromechanical trap-style GRS is installed.) If a hydromechanical trap-style GRS is installed, the kitchen may not have a garbage disposal/garbage grinder/macerator or similar unit installed. See Sec. 3.5.
- B. All new construction multi-tenant buildings (strip centers/malls) shall include a separate waste line for all leasable spaces that discharges to a common 2000 gallon or larger interceptor (See Sec. 3.6.E.). This waste line shall be permanently marked to identify it as required by the Director. When a space is leased, sold, or rented to a FSE or NFD, all kitchen drains and any other drains that may carry grease-laden waste shall be connected to this waste line; no domestic sewage may be connected to this line. The property owner shall be responsible for proper maintenance of this interceptor in accordance with the provisions of this ordinance.
- C. All new single occupancy Non-FSE FOG Discharger buildings shall install a properly sized GRS. Interceptor-style GRSs are recommended, but hydromechanical trap-style GRSs are permissible. All kitchen drains and any other drains that may carry grease-laden waste shall be connected to this GRS (except the dishwasher if a hydromechanical trap-style GRS is installed.) If an hydromechanical trap-style GRS is installed, the kitchen may not have a garbage disposal/garbage grinder/macerator or similar unit installed.

D. Any FSE or NFD undertaking a Substantial Remodel, as defined herein (Sec. 1.3.BG.), will be considered new construction for the purposes of this ordinance.

## 3.2 Existing Users

- A. Any existing FSE without a functional GRS shall be required to install one. The type of GRS required will be determined by the Director, taking into account cost, available space and gradient, whether the user is in a grease impact area, and any other pertinent information. Unless otherwise approved by the Director, all kitchen drains and any other drains that may carry grease-laden waste shall be connected to this GRS (except the dishwasher if a hydromechanical trap-style GRS is installed.) If a hydromechanical trap-style GRS is installed, the kitchen may not have a garbage disposal/garbage grinder/macerator or similar unit installed.
- B. Any existing NFD without a functional GRS may be required to install one. The type of GRS required will be determined by the Director, taking into account cost, available space and gradient, whether the user is in a grease impact area, and any other pertinent information. Unless otherwise approved by the Director, all kitchen drains and any other drains that may carry grease-laden waste shall be connected to this GRS (except the dishwasher if a hydromechanical trap-style GRS is installed.) If a hydromechanical trap-style GRS is installed, the kitchen may not have a garbage disposal/garbage grinder/macerator or similar unit installed.

# 3.3 Grease Removal System Maintenance

- A. All grease removal systems shall be maintained to ensure proper operation. At a minimum, interceptor-style GRSs shall be cleaned at least once every 90 days and hydromechanical trap-style GRSs cleaned at least once per week. These required frequencies may be extended with the approval of the Director. Grease Removal Systems must be thoroughly cleaned whenever the combined thickness of the floating greases and settled solids is equal to, or greater than, 25% of the total liquid depth in the GRS.
- B. When cleaned, an interceptor-style GRS must be completely pumped out, all solids removed, solidified grease scraped or washed from the interior and the structure and all internal plumbing inspected for damage and corrosion. The GRS shall be refilled with water prior to being placed back into operation. If repairs are required, they shall be performed within 7 days, or within the shortest practicable time frame depending on repairs required, and as approved by the Director or designee.
- C. Hydromechanical trap-style GRSs may be maintained by the FSE or NFD. When cleaned, the trap must have surface grease and oil removed, settled solids removed, all sides scraped, removable parts removed and cleaned, be inspected for damage and corrosion, and be properly reassembled. If the GRS is equal to or greater than 25% full, it must be thoroughly cleaned as noted in 3.3.A. If repairs are required,

they shall be performed within 7 days, or within the shortest practicable time frame as approved by the Director.

- D. The material that is removed in the process of cleaning a GRS shall not be discharged back into the GRS, any part of the POTW, any private sewer, any drainage piping, waters of the state, or storm sewer system. The practices of stacking and decanting are strictly prohibited. All materials removed shall be handled and disposed of in accordance with Federal, State, County and Local laws, rules and regulations.
- E. In addition to the maintenance required above, Automatic Grease Removal Systems shall be maintained in accordance with the manufacturers' guidelines.

# 3.4 Grease Removal System Additives

No user may use an additive of any type for the GRS without the approval of the Director. The Director will adopt procedures for the approval of additives.

No vendor may sell, attempt to sell or otherwise distribute any additive in the City of Everett without prior approval of the additive by the Director. Any vendor selling, or attempting to sell, or otherwise distributing any additive intended for use in a GRS that has not been approved for use by the Director is in violation of this ordinance and is subject to all enforcement actions contained herein.

# 3.5 Solids Interceptor

If a garbage disposal/garbage grinder/macerator or similar unit is installed in a kitchen, it must discharge to the GRS through a solids interceptor plumbed immediately after the garbage disposal/garbage grinder/macerator or similar unit. The solids interceptor shall be maintained in proper operating condition at all times there is flow through it.

# 3.6 Grease Removal System Sizing

- A. Hydromechanical Trap-style grease removal systems shall be sized in accordance with the standards in the currently adopted Plumbing Code. But in no case shall they be rated at less than 20 GPM without express approval by the Director.
- B. The minimum size for an interceptor-style grease removal system shall be determined using the following table:

Drainage Fixture Units <sup>1</sup>	Nominal Interceptor Volume (gallons)
21	750
35	1000
90	1250
172	1500
216	2000
307	2500
342	3000
428	4000
576	5000 <sup>D</sup>
724	6000 <sup>D</sup>
880	7000 <sup>D</sup>
1036	8000 <sup>D</sup>
>1036	8000 <sup>D</sup>

#### Note:

- Drainage Fixture Units (as defined in the currently adopted Uniform Plumbing Code) for installed equipment plumbed to kitchen drain lines connected to the Grease Removal System.
  - C. No conventional interceptor smaller than 750 gallons shall be installed without prior approval of the Director. High-Efficiency or High-Capacity Interceptors not meeting minimum volume criteria shall be reviewed and approved on a case-by-case basis. If a garbage disposal/garbage grinder/macerator or similar unit is installed, the minimum size is 1500 gallons to adequately retain the added settleable solids loading.
  - D. The maximum size for an interceptor shall be 4000 gallons. If the calculated minimum size is larger than this, two interceptors of approximately equal size shall be installed in series. If the calculated minimum size is 8000 gallons, two 4000-gallon interceptors shall be installed in series.
  - E. Shared Grease Interceptors (Multi-Tenant/Strip Center). The number or volume of interceptors shared by multiple tenants/users may be adjusted if deemed necessary by the Director or Director's Designee, based on an assessment of square footage, size of water service, and/or the nature of the FSEs or NFDs served.

#### 3.7 Flow Controls

All hydromechanical trap-style grease removal systems shall have an internal or external flow control installed to ensure that wastewater flow through the trap does not exceed the manufacturer's design flow rating. This flow control shall be maintained in operating condition at all times.

# SECTION 4 – RECORD KEEPING/REPORTING REQUIREMENTS

## 4.1 Record Keeping

Users subject to this ordinance shall document all cleaning and maintenance activities performed on their GRS. These records shall be maintained for a minimum of three (3) years and be available onsite for inspection and copying by the Director or the Director's representative. This period shall be automatically extended for the duration of any litigation concerning the user or the POTW, or where the user has been specifically notified of a longer retention period required by the Director.

# 4.2 Reporting

Where the Director has determined that a user must provide written reports, these reports shall be submitted in accordance with the requirement of the Director. Written reports will be deemed to have been submitted on the date postmarked. For reports which are not mailed, postage prepaid, into a mail facility serviced by the United States Postal Service, the date of receipt of the report by the City shall govern.

# 4.3 Electronic Reporting

The City may implement electronic reporting for users. Until the electronic reporting system is approved under the Cross-Media Electronic Reporting Rule (CROMERR) and the individual user has complied with identity certification requirements, electronic reports must be followed up with original paper reports. Electronically submitted reports will be deemed to have been submitted on the date they are posted to the electronic reporting system. Electronic reporting does not preclude the City from requiring a user to continue submitting paper reports, if the City deems it is still necessary.

When service providers (contractors) submit electronic copies of trap or interceptor-style GRS cleaning, maintenance, or repairs, such service records should include sufficient information to assess the condition and compliance of the serviced GRS. This includes an estimate of combined thickness of the floating greases and settled solids, tank damage or corrosion, and the condition of the trap or interceptor's internal plumbing (See Section 3.3).

## **SECTION 5 - CONFIDENTIAL INFORMATION**

Information and data on a user obtained from reports, surveys, wastewater discharge permit applications, wastewater discharge permits, and monitoring programs, and from City inspection and sampling activities, shall be available to the public without restriction, unless the user specifically requests, and is able to demonstrate to the satisfaction of the City, that the release of such information would divulge information, processes or methods of production that are exempt from disclosure under the Public Records at Chapter 42.56 RCW, or as hereafter amended.

When requested and demonstrated by the user furnishing a report that such information should be held confidential, the City shall make reasonable efforts to protect the portions of a report which might disclose trade secrets or secret processes from inspection by the public, but shall be made available immediately upon request to governmental agencies for uses related to the NPDES program or pretreatment program, in enforcement proceedings involving the person furnishing the report, or upon presentation of a subpoena or other court order.

Wastewater constituents, characteristics, and other effluent data as defined by 40 CFR 2.302, or the identity and address of the user will not be recognized as confidential information and will be available to the public without restriction.

## **SECTION 6 - ADMINISTRATIVE ENFORCEMENT REMEDIES**

# 6.1 Notification of Violation (Notice of Violation, NOV)

When the Director finds that a user has violated (or continues to violate) any provision of this ordinance, a wastewater discharge permit or order issued hereunder, or any other pretreatment standard or requirement, the Director may serve upon that user a written Notice of Violation. The Director may select any means of service which is reasonable under the circumstances.

Within seven calendar (7) days of the receipt of this notice, an explanation of the violation and a plan for the satisfactory correction and prevention thereof, to include specific required actions, shall be submitted by the user to the Director. Submission of this plan in no way relieves the user of liability for any violations occurring before or after receipt of the Notice of Violation. Nothing in this section shall limit the authority of the City to take any action, including emergency actions or any other enforcement action, without first issuing a Notice of Violation.

### 6.2 Consent Orders

The Director may enter into Consent Orders, assurances of voluntary compliance in "good faith," or other negotiated settlements between the City and the user establishing an agreement with any user responsible for noncompliance. Such documents will include specific action to be taken by the user to correct the noncompliance within a time period specified by the document. Such documents shall have the same force and effect as the administrative orders issued pursuant to Sections 6.4 and 6.5 of this ordinance and shall be judicially enforceable. Use of a Consent Order shall not be a bar against, or prerequisite for, taking any other action against the user.

# 6.3 Show Cause Hearing

The Director may order a user which has violated or continues to violate, any provision of this ordinance, a wastewater discharge permit or order issued hereunder, or any other pretreatment standard or requirement, to appear before the Director and show cause why a proposed enforcement action should not be taken. Notice shall be served on the user specifying the time and place for the hearing, the proposed enforcement action, the reasons for such action, and a request that the user show cause why the proposed enforcement action should not be taken. The notice of the hearing shall be served personally or by registered or certified mail (return receipt requested) at least ten business (10) days prior to the hearing. Such notice may be served on any authorized representative of the user. A show cause hearing shall not be a bar against, or prerequisite for, taking any other action against the user.

## 6.4 Compliance Orders

When the Director finds that a user has violated or continues to violate any provision of this ordinance, a wastewater discharge permit or order issued hereunder, or any other pretreatment standard or requirement, the Director may issue an order to the user responsible for the discharge directing that the user come into compliance within a time specified in the order. Compliance Orders may require users to refrain from certain activities, install additional pretreatment equipment, increase self-monitoring, use best management practices designed to minimize the amount of pollutants discharged to the sewer. A compliance order may be issued unilaterally, and the terms of the order need not be discussed with the user in advance. If the user does not come into compliance within the time specified in the order, sewer service may be discontinued. Issuance of a compliance order shall not be a bar against, or a prerequisite for, taking any other action against the user.

### 6.5 Cease and Desist Orders

When the Director finds that a user has violated (or continues to violate) any provision of this ordinance, a wastewater discharge permit or order issued hereunder, or any other pretreatment standard or requirement, or that the user's past violations are likely to recur, the Director may issue an order to the user directing it to cease and desist all such violations and directing the user to:

- A. Immediately comply with all requirements; and
- B. Take such appropriate remedial or preventive action as may be needed to properly address a continuing or threatened violation, including halting operations and/or terminating the discharge.

Issuance of a cease and desist order shall not be a bar against, or a prerequisite for, taking any other action against the user.

#### 6.6 Administrative Fines

- A. When the Director finds that a user has violated or continues to violate any provision of this ordinance, a wastewater discharge permit or order issued hereunder, or any other pretreatment standard or requirement, the Director may fine such user in an amount not less than \$100 and not to exceed \$10,000. Such fines shall be assessed on a per violation, per day basis. Said Administrative fines shall constitute a sewer service surcharge, and upon assessment, shall be subject to collection in the same manner as all other sewer utility rates, charges and penalties.
- B. Unless other arrangements have been made with, and authorized by the Director, unpaid charges, fines, and penalties shall accrue thereafter at a rate of one percent (1%) per month. After 90 days, if charges, fines, and penalties have not been paid, the City may suspend the user's sewer service.

C. Users desiring to appeal and dispute such fines must file a written request for the Director to reconsider the fine along with full payment of the fine amount within ten calendar (10) days of being notified of the fine. Upon receipt of a timely appeal, the Director shall set a date and time for an appeal hearing, but in no case shall the hearing be set more than sixty business (60) days from the receipt of the timely notice of appeal. The appellant shall be notified in writing of the date, time, and place for the appeal hearing. The Director or Director's designee shall serve as the Hearing Examiner. In the event the user's appeal is successful, any payments made shall be returned to the user. Affirmation or modification of an administrative fine by the Public Works Director shall relate back to the original date of assessment.

The City shall recover the costs of preparing administrative enforcement actions, such as notices and orders, including the cost of additional inspections, sampling and analysis, and may add them to the fine. The City may also add the monetary value of any economic benefit the user gained through noncompliance to the fine.

- D. Issuance of an administrative fine shall not be a bar against, or a prerequisite for, taking any other action against the user.
- E. Users seeking judicial review of administrative fines must do so by filing a Petition for Review in the Snohomish County Superior Court within thirty calendar (30) days of the decision of the Director.

# 6.7 Emergency Suspensions

The Director may immediately suspend a user's discharge (after informal notice to the user) whenever such suspension is necessary to stop an actual or threatened discharge which reasonably appears to present or cause an imminent or substantial endangerment to the health or welfare of persons, threatens to interfere with the operation of the POTW, or which presents or may present an endangerment to the environment.

A. Any user notified of a suspension of its discharge shall immediately stop or eliminate its contribution. In the event of a user's failure to immediately comply voluntarily with the suspension order, the Director shall take such steps as deemed necessary, including immediate severance of the sewer connection, to prevent or minimize damage to the POTW, its receiving stream, or endangerment to any individuals.

The Director may allow the user to recommence its discharge when the user has demonstrated to the satisfaction of the City that the period of endangerment has passed. If the Director does not allow the user to recommence its discharge within 15 days of the emergency suspension, the Director shall initiate termination proceedings pursuant to Section 6.8 of this ordinance.

B. A user that is responsible, in whole or in part, for any discharge presenting imminent endangerment shall submit a detailed written statement, describing the causes of the harmful contribution and the measures taken to prevent any future occurrence, to the Director prior to the date of any show cause or termination hearing under Sections 6.3 and 6.8 of this ordinance.

Nothing in this section shall be interpreted as requiring a hearing prior to any emergency suspension under this section.

# 6.8 Termination of Discharge (Non-Emergency)

Any user that violates the following conditions is subject to discharge termination:

- A. Violation of wastewater discharge permit conditions;
- B. Failure to accurately report the wastewater constituents and characteristics of its discharge;
- C. Failure to report significant changes in operations or wastewater volume, constituents and characteristics prior to discharge;
- D. Refusal of reasonable access to the user's premises for the purpose of inspection, monitoring or sampling; or
- E. Violation of the pretreatment standards in Section 2 of this ordinance.

Such user will be notified of the proposed termination of its discharge and be offered an opportunity for hearing pursuant to Section 6.9 of this ordinance to dispute the proposed termination action. Initiation of a termination proceeding by the City shall not be a bar to, or a prerequisite for, taking any other action against the user.

# 6.9 Appeal Procedures

- A. Appeals.
  - 1. Any user who has been subject to an enforcement action by the City and who seeks to dispute a Notice of Violation, order, fine, or other action of the Director may file an appeal. No other person may appeal an enforcement action.
  - 2. The notice of appeal must be filed in writing and received by the Director, in writing, within ten (10) calendar days of the receipt of the disputed action or proposed action. If the notice of appeal is not received by the Director within ten (10) days, the right to an appeal is waived. The notice of appeal shall state with particularity the basis upon which the appellant is disputing the action taken or proposed to be taken Including:

- (a) The name and the address of the appellant;
- (b) The nature of the decision or action being appealed;
- (c) Why appellant believes the decision or action is unwarranted; and
- (d) What the appellant believes the appropriate decision or action should be.
- 3. Upon receipt of a timely notice of appeal, the Director shall set a date and time for an appeal hearing, but in no case shall the hearing be set more than sixty (60) days from the receipt of the timely notice of appeal. The appellant shall be notified in writing of the date, time, and place for the appeal hearing. The Director or Director's designee shall serve as the hearing examiner and be the presiding officer at the hearing.

#### B. Appeal Hearing.

- 1. Content of Notice of hearing. The notice of hearing shall include:
  - (a) Names and mailing addresses of all parties to whom notice is being given, and if known, the names and addresses of their representatives;
  - (b) If the City intends to appear, the mailing address and telephone number of the office designated to represent the City in the proceeding;
  - (c) The official file or other reference number and name of proceeding;
  - (d) The name, official title, mailing address and telephone number of the presiding officer, if known;
  - (e) A statement of the time, place and nature of the proceeding;
  - (f) A statement of the legal authority and jurisdiction under which the hearing is to be held;
  - (g) A reference to the particular section of the ordinance or regulations involved;
  - (h) A short and plain statement of the matters asserted by the agency; and
  - (i) A statement that a party who fails to attend or participate in a hearing or other stage of an appeal hearing may be held in default.
- 2. Procedures at hearing. The hearing examiner/presiding officer, who may be the Director or the Director's designee, shall regulate the course of the hearing. The presiding officer shall afford to all parties the opportunity to respond, present evidence and argument, conduct cross-examination, and submit rebuttal evidence; provided, however, the presiding officer may control the manner and extent of cross-examinations and rebuttal. In the discretion of the presiding officer, all or part of the hearing may be conducted by telephone or other electronic means as long as each party in the hearing has an opportunity to effectively participate and hear.

- 3. Rules of evidence. Evidence, including hearsay evidence, is admissible if in the judgment of the presiding officer it is the kind of evidence on which reasonably prudent persons are accustomed to rely in the conduct of their affairs. The presiding officer shall exclude evidence that is excludable on constitutional or statutory grounds or on the basis of evidentiary privilege recognized in the courts of this state. The presiding officer may exclude evidence that is irrelevant, immaterial, or unduly repetitious. All testimony of parties and witnesses shall be made under oath or affirmation. Documentary evidence may be received in the form of copies or excerpts or by incorporation by reference. Official notice may be taken of (a) any judicially cognizable facts, (b) technical or scientific facts within the City's specialized knowledge, and (c) codes or standards that have been adopted by an agency of the United States, this state or another state, or by a nationally recognized organization or association. A party proposing that official notice be taken may be required to produce a copy of the material to be noticed.
- 4. Default. If a party fails to attend or participate in any stage of a hearing, the presiding officer may serve upon all parties a default or other dispositive order, which shall include a statement of grounds for the order. Within seven (7) calendar days after service of a default order, the party against whom it was entered may file a written motion requesting that the order be vacated, stating the grounds for the motion. If the party against whom the default order is entered fails to timely file a motion to vacate or the motion to vacate is not granted, the default order will be the final decision of the City.
- 5. Burden of proof. The appellant shall have the burden of proof by a preponderance of the evidence.

#### C. Appeal Conclusion.

After the conclusion of the hearing, the hearing examiner shall determine if the disputed action was proper, and shall approve, modify, or rescind the disputed action. The final determination of the hearing examiner shall be in writing, and all parties shall be provided a copy of the final determination within ten (10) days (or such longer time as the hearing examiner may determine in writing). This decision will include findings of fact that are supported by and based on the record. These findings will be entitled to deference on any judicial review.

- D. Judicial Review of Appeal.
  - 1. Any party, including the City, the Washington State Department of Ecology, the United States Environmental Protection Agency, or the user/appellant, is entitled to review of the final determination of the hearing examiner in the Snohomish County Superior Court. Provided, that any petition for review shall be filed no later than thirty (30) calendar days after date of the final determination.
  - 2. Copies of the petition for review shall be served as in all civil actions.
  - 3. The filing of the petition shall not stay enforcement of the final determination except by order of the superior court and on posting of a bond to be determined by the court naming the City as beneficiary.
  - 4. The review shall be conducted by the court without a jury. The record shall be satisfied by a narrative report certified by the hearing examiner and no verbatim record of proceedings before the hearing examiner shall be required to be presented to the superior court.
  - 5. The court may affirm the final determination or remand the matter for further proceedings before the hearing examiner; or the court may reverse the final determination if the substantial rights of the petitioners may have been prejudiced because the final determination was:
    - (a) In violation of constitutional provisions; or
    - (b) In excess of the authority or jurisdiction of the hearing examiner; or
    - (c) Arbitrary and capricious.

## **SECTION 7 - JUDICIAL ENFORCEMENT REMEDIES**

# 7.1 Injunctive Relief

When the Director finds that a user has violated or continues to violate any provision of this ordinance, a wastewater discharge permit, or order issued hereunder, or any other pretreatment standard or requirement, the Director may petition the Snohomish County Superior Court through the City's Attorney for the issuance of a temporary or permanent injunction, as appropriate, which restrains or compels the specific performance of the wastewater discharge permit, order, or other requirement imposed by this ordinance on activities of the user.

The City may also seek such other action as is appropriate for legal and/or equitable relief, including a requirement for the user to conduct environmental remediation. A petition for injunctive relief shall not be a bar against, or a prerequisite for, taking any other action against a user. Injunctive relief shall be nonexclusive to other remedies available to the City.

#### 7.2 Civil Penalties

- A. A user which has violated or continues to violate any provision of this ordinance, a wastewater discharge permit, or order issued hereunder, or any other pretreatment standard or requirement shall be liable to the City for a maximum civil penalty of \$10,000 per violation, per day. In the case of a monthly or other long-term average discharge limit, penalties shall accrue for each day during the period of the violation.
- B. The Director may recover reasonable attorneys' fees, court costs, and other expenses associated with enforcement activities, including sampling and monitoring expenses, and the cost of any actual damages incurred by the City. The City may also add the monetary value of any economic benefit the user gained through noncompliance to the penalty.
- C. In determining the amount of civil liability, the Court shall take into account all relevant circumstances, including, but not limited to, the extent of harm caused by the violation, the magnitude and duration, any economic benefit gained through the user's violation, corrective actions by the user, the compliance history of the user, and any other factor as justice requires.
- D. Filing a suit for civil penalties shall not be a bar against, or a prerequisite for taking any other action against a user.

#### 7.3 Criminal Prosecution

- A. A user who violates any provision of this ordinance, a wastewater discharge permit, or order issued hereunder, or any other pretreatment standard or requirement shall, upon conviction, be guilty of a gross misdemeanor, punishable by a fine of not more than \$5,000 and/or one year in jail. Each day a violation occurs shall constitute a separate offense.
- B. A user who introduces any substance into the POTW which causes personal injury or property damage shall, upon conviction, be guilty of a gross misdemeanor and be subject to a penalty of not more than \$5,000 and/or one year in jail. Each day a violation occurs shall constitute a separate offense. This penalty shall be in addition to any other cause of action for personal injury or property damage available under State law.
- C. A user who knowingly makes any false statements, representations, or certifications in any application, record, report, plan, or other documentation filed, or required to be maintained, pursuant to this ordinance, wastewater discharge permit, or order issued hereunder, or who falsifies, tampers with, or knowingly renders inaccurate any monitoring device or method required under this ordinance shall, upon conviction, be guilty of a gross misdemeanor, and punished by a fine of not more than \$5,000 and/or one year in jail. Each day a violation occurs shall constitute a separate offense.

In addition, the user shall be subject to:

- 1) the provisions of 18 USC Section 1001 relating to fraud and false statements;
- 2) the provisions of Sections 309 (c) 4 of the Clean Water Act, as amended governing false statements, representation, or certification; and
- 3) the provision of Section 309 (c) (6) of the Clean Water Act, regarding responsible corporate officers.

### 7.4 Remedies Non-exclusive

The provisions in Sections 6 through 9 of this ordinance are not exclusive remedies. The City reserves the right to take any, all, or any combination of these actions against a non-compliant user. Enforcement of pretreatment violations will generally be in accordance with the City's enforcement response plan. However, the City reserves the right to take other action against any user when the circumstances warrant. Further, the City is empowered to take more than one enforcement action against any non-compliant user. These actions may be taken concurrently.

## **SECTION 8 - SUPPLEMENTAL ENFORCEMENT ACTION**

#### 8.1 Performance Bonds

The Director may decline to issue or reissue a wastewater discharge permit to any permitted user which has failed to comply with any provision of this ordinance, a previous wastewater discharge permit or order issued hereunder, or any other pretreatment standard or requirement unless such user first files a satisfactory bond, payable to the City, in a sum not to exceed a value determined by the Director to be necessary to achieve consistent compliance.

#### 8.2 Financial Assurances

The Director may decline to issue or reissue a wastewater discharge permit to any permitted user which has failed to comply with any provision of this ordinance, a previous wastewater discharge permit or order issued hereunder, or any other pretreatment standard or requirement, unless the user first submits proof that it has obtained financial assurances sufficient to meet pretreatment requirements, and/or restore or repair damage to the POTW caused by its discharge.

#### 8.3 Service Severance

Whenever a user has violated or continues to violate any provision of this ordinance, fails to respond as required to an enforcement action or order issued hereunder, or any other pretreatment standard or requirement, water and/or sewer service to the user may be severed. Service will only recommence, at the user's expense, after it has satisfactorily demonstrated its ability to comply.

## 8.4 Public Nuisances

A violation of any provision of this ordinance, wastewater discharge permit, or order issued hereunder, or any other pretreatment standard or requirement, is hereby declared a public nuisance and shall be corrected or abated as directed by the Director.

# 8.5 Contractor Listing

Users which have not achieved compliance with applicable pretreatment standards and requirements are not eligible to receive a contractual award for the sale of goods or services to the City. Existing contracts for the sale of goods or services to the City held by a user found to be in significant noncompliance with pretreatment standards or requirements may be terminated at the discretion of the City.

## 8.6 Publication of Violations and/or Enforcement Actions.

The Director may publish violations and/or enforcement actions at any time, where monetary fines may be inappropriate in gaining compliance, or in addition to monetary fines. Violations and/or enforcement actions may also be published when the Director feels that public notice should be made, or at other appropriate times. The cost of such publications will be recovered from the user.

# SECTION 9 - AFFIRMATIVE DEFENSES TO DISCHARGE VIOLATIONS

## 9.1 Upset

- A. For the purposes of this section, upset means an exceptional incident in which there is unintentional and temporary noncompliance with applicable pretreatment standards because of factors beyond the reasonable control of the user. An upset does not include noncompliance to the extent caused by operational error, improperly designed treatment facilities, inadequate treatment facilities, lack of preventive maintenance, or careless or improper operation.
- B. An upset shall constitute an affirmative defense to an action brought for noncompliance with applicable pretreatment standards if the requirements of paragraph (C) are met.
- C. A user who wishes to establish the affirmative defense of upset shall demonstrate, through properly signed or initialed contemporaneous operating logs, or other relevant evidence that:
  - (1) An upset occurred and the user can identify the cause(s) of the upset;
  - (2) The facility was at the time being operated in a prudent and workman-like manner and in compliance with applicable operation and maintenance procedures; and
  - (3) The user has submitted the following information to the POTW and treatment plant operator within twenty four (24) hours of becoming aware of the upset (if this information is provided orally, a written submission must be provided within five (5) days):
    - (a) A description of the indirect discharge and cause of noncompliance;
    - (b) The period of noncompliance, including exact dates and times or, if not corrected, the anticipated time the noncompliance is expected to continue; and
    - (c) Steps being taken and/or planned to reduce, eliminate, and prevent recurrence of the noncompliance.
- D. In any enforcement proceeding, the user seeking to establish the occurrence of an upset shall have the burden of proof.
- E. A User may appeal any enforcement action due to an upset as provided under Section 6.9 of this ordinance.

F. Users shall control production of all discharges to the extent necessary to maintain compliance with applicable pretreatment standards upon reduction, loss, or failure of its treatment facility until the facility is restored or an alternative method of treatment is provided. This requirement applies in the situation where, among other things, the primary source of power of the treatment facility is reduced, lost, or fails.

#### 9.2 Not Used

# 9.2 Bypass

- A. For the purposes of this section,
  - (1) Bypass means the intentional diversion of wastestreams from any portion of a user's treatment facility.
  - (2) Severe property damage means substantial physical damage to property, damage to the treatment facilities which causes them to become inoperable, or substantial and permanent loss of natural resources which can reasonably be expected to occur in the absence of a bypass. Severe property damage does not mean economic loss caused by delays in production.
- B. A user may allow any bypass to occur which does not cause applicable pretreatment standards or requirements to be violated, but only if it also is for essential maintenance to assure efficient operation. These bypasses are not subject to the provision of paragraphs (C) and (D) of this section.
- C. (1) If a user knows in advance of the need for a bypass, it shall submit prior notice to the Director, at least ten (10) days before the date of the bypass, if possible.
  - (2) A user shall submit oral notice to the City of an unanticipated bypass that exceeds applicable pretreatment standards within twenty four (24) hours from the time it becomes aware of the bypass. A written submission shall also be provided within five (5) days of the time the user becomes aware of the bypass. The written submission shall contain a description of the bypass and its cause; the duration of the bypass, including exact dates and times, and, if the bypass has not been corrected, the anticipated time it is expected to continue; and steps taken or planned to reduce, eliminate, and prevent reoccurrence of the bypass. The Director may waive the written report on a case-by-case basis if the oral report has been received within twenty four (24) hours.
- D. (1) Bypass is prohibited, and the Director may take an enforcement action against a user for a bypass, unless

- (a) Bypass was unavoidable to prevent loss of life, personal injury, or severe property damage;
- (b) There were no feasible alternatives to the bypass, such as the use of auxiliary treatment facilities, retention of untreated wastes, or maintenance during normal periods of equipment downtime. This condition is not satisfied if adequate back-up equipment should have been installed in the exercise of reasonable engineering judgment to prevent a bypass which occurred during normal periods of equipment downtime or preventive maintenance; and
- (c) The user submitted notices as required under paragraph (C) of this section.
- (2) The Director may approve an anticipated bypass, after considering its adverse effects, if the Director determines that it will meet the three conditions listed in paragraph (D)(1) of this section.

## **SECTION 10 - MISCELLANEOUS PROVISIONS**

## 10.1 Pretreatment Charges and Fees

The Director may adopt reasonable fees for reimbursement of costs of setting up and operating the City's FOG Control Program.

These fees relate solely to the matters covered by this ordinance and are separate from all other rates or charges for sewer service, provided that the City shall collect said charges in the same manner as other sewer utility rates are collected, including but not limited to the sewer lien procedures provided under 35.67 RCW.

#### Fees may include:

- A. Fees for wastewater discharge permits, including the cost of processing the permit applications, public noticing, issuing and administering the permit, and reviewing monitoring reports submitted by users;
- B. Fees for modifying or transferring permits;
- C. Fees for monitoring, inspection, surveillance and enforcement procedures including the cost of collection and analyzing a user's discharge;
- D. Fees for reviewing accidental discharge procedures and construction;
- E. Fees for preparing and executing enforcement action;
- F. Fees for filing appeals;
- G. Fees for High Strength Waste and non-domestic flow; and
- H. Other fees as the City may deem necessary to carry out the requirements contained herein.
- I. <u>Permit Fees</u>: Should the Director determine that adequate FOG control activities require issuance of control mechanism to FSEs and NFDs, the fees shall be set by the Director.
  - J. <u>Monitoring Fees:</u> Any user establishing a pattern of noncompliance, or having a history of noncompliance, or suspected of being in noncompliance, may require additional monitoring visits as deemed appropriate by the Director. Any additional inspections, sampling, surveillance monitoring activities, and analysis performed which detect noncompliance will be billed directly to the user.
  - K. <u>Enforcement Actions</u>: All expenses in preparing enforcement actions will be billed directly to the User.

L. <u>Cost Recovery</u>: The City may recover any expenses incurred due to noncompliance by a user, including, but not limited to; costs of opening a plugged sewer, costs of repairing a damaged sewer, costs of determining the cause of a SSO or CSO, and costs of cleaning up and mitigating a SSO or CSO.

All fees or charges will be collected via direct billing. Unless the Director has been made aware of extenuating circumstances that would prevent prompt payment, all fees are payable within 30 days of the billing. Fees past due will be considered a violation of this ordinance. Users not paying fees within 60 days of the billing period will be subject to termination of service. The Director may change existing or adopt new fees.

# 10.2 Non-Liability

It is expressly the purpose of this ordinance to provide for and promote the health, safety and welfare of the general public. It is not the intent of this ordinance to create or otherwise establish or designate any particular class or group of persons who will or should be especially protected or benefited by the terms or requirements of this ordinance.

It is the specific intent of this ordinance to place the obligation of complying with these regulations upon the applicant or Discharger and no provision nor any term used in this ordinance is intended to impose any duty whatsoever upon the City or any of its officers, employees or agents, except as provided under the Act or other related statutes of the United States or the State of Washington.

Nothing contained in this ordinance is intended to be nor shall be construed to create or form the basis for any tort liability on the part of the City or its officers, employees or agents, for any injury or damage resulting from the failure of an applicant or Discharger to comply with the provisions of this ordinance, or by reason or in consequence of any inspection, notice, order, certificate, permission or approval authorized or issued or done in connection with the implementation or enforcement of this ordinance, or inaction on the part of the City related in any manner to the implementation or the enforcement of this ordinance by its officers, employees or agents.

# **EVERETT** City Council Agenda Item Cover Sheet

**Project title:** ORDINANCE Repealing and Replacing Chapter 14.40 of the Everett Municipal Code

Council Bill # CB 2404-07	Project: 2024 Adoption of the new Chapter 14.40 EMC related to wastewater pretreatment regulations	
A d - d - d d - d -	Partner/Supplier: n/a	
Agenda dates requested:	Location: n/a	
Briefing 1 <sup>st</sup> Reading 5/08/24	Preceding action: 2008 Adoption of Chapter 14.40 EMC related to wastewater pretreatment regulations under Ord. 3070-08	
2 <sup>nd</sup> Reading 5/15/24 Consent	Fund: n/a	
Action 5/22/24 Ordinance X Public hearing Yes x No	Fiscal summary statement: This action item has no fiscal impact associated at this time.	
Budget amendment: Yes x No	Project summary statement:	
PowerPoint presentation: Yes x No  Attachments: Ordinance	In 2019 our Water Pollution Control Facility (WPCF) biosolids exceeded Molybdenum (Mo) levels for beneficial use. This led to an investigation of our incoming wastewater influent into the source of the contamination. The industrial source was identified and corrected, however, during the investigation it became clear that the City needed a Mo limit.	
Department(s) involved: Public Works  Contact person: Jeff Marrs	The City's current Local Limits were established in 1993 and there have since been significant changes to both the plant and water quality regulations prompting the reevaluation of all limits. Chapter 14.40 of the Everett Municipal Code authorizes enforcement of Local Limits.	
Phone number: (425) 257-8967	The new Chapter 14.40 EMC reflect current practices, regulations, city policies, coding and revises the established Local Limits.	
Email: jmarrs@everettwa.gov	The proposed Ordinance was available for a 30-day public comment period ending Apri 18, 2024.	
	Recommendation (exact action requested of Council):	
Initialed by:	Adopt an Ordinance relating to Uniform Requirements for Users of the Publicly Owned Treatment Works, repealing and replacing Ordinance 3070-08 (Chapter 14.40 EMC).	
Department head		
Administration		
Council President		



ORDINANCE NO.	
---------------	--

An ORDINANCE relating to Wastewater Pretreatment Regulations, repealing Everett Municipal Code Chapter 14.40 and adopting a New Everett Municipal Code Chapter 14.40

#### WHEREAS,

- A. The City Council deems it necessary to set forth uniform requirements for users of the City's publicly owned treatment works in order to comply with all applicable State and Federal laws and regulations.
- B. The City's wastewater pretreatment regulations, codified at Chapter EMC 14.40, were last amended in 2008, and the City recognizes the need to update these regulations.
- C. Washington State's Department of Ecology and the United States Environmental Protection Agency have reviewed and approved the requirements for users of the City's publicly owned treatment works contained in this Ordinance; and

#### NOW, THEREFORE, THE CITY OF EVERETT DOES ORDAIN:

**Section 1.** Chapter 14.40 EMC is hereby repealed in its entirety. For purposes of reference, chapter 14.40 EMC was adopted under Ord. 3070-08, 2008.

<u>Section 2.</u> A new chapter 14.40 EMC is hereby added to the Everett Municipal Code as set forth in <u>Exhibit A</u>. Permits issued under chapter 14.40 EMC adopted under Ord. 3070-08 shall hereafter be subject to the provisions of the new chapter 14.40 EMC attached as Exhibit A hereto.

<u>Section 3.</u> The City Clerk and the codifiers of this Ordinance are authorized to make necessary corrections to this Ordinance including, but not limited to, the correction of scrivener's/clerical errors, references, ordinance numbering, section/subsection numbers, and any internal references.

<u>Section 4</u>. The City Council hereby declares that should any section, paragraph, sentence, clause or phrase of this ordinance be declared invalid for any reason, it is the intent of the City Council that it would have passed all portions of this ordinance independent of the elimination of any such portion as may be declared invalid.

Section 5. The enactment of this Ordinance shall not affect any case, proceeding, appeal or other

matter currently pending in any court or in any way modify any right or liability, civil or criminal, which may be in existence on the effective date of this Ordinance.

<u>Section 6</u>. It is expressly the purpose of this Ordinance to provide for and promote the health, safety and welfare of the general public and not to create or otherwise establish or designate any particular class or group of persons who will or should be especially protected or benefited by the terms of this Ordinance. It is the specific intent of this Ordinance that no provision or any term used in this Ordinance is intended to impose any duty whatsoever upon the City or any of its officers or employees. Nothing contained in this Ordinance is intended nor shall be construed to create or form the basis of any liability on the part of the City, or its officers, employees or agents, for any injury or damage resulting from any action or inaction on the part of the City related in any manner to the enforcement of this Ordinance by its officers, employees or agents.

Cassie Franklin, Mayor	
ATTEST:	
City Clerk	
PASSED:	
VALID:	
PUBLISHED:	
EFFECTIVE DATE:	

# EXHIBIT A TO WASTEWATER PRETREATMENT REGULATIONS ORDINANCE

## **CHAPTER 14.40 EMC**

Reference Table for Codification				
Section Number in Attached Text	Code Section Number for Codification			
1.1	14.40.010			
1.2	14.40.020			
1.3	14.40.030			
1.4	14.40.040			
2.1	14.40.050			
2.2	14.40.060			
2.3	14.40.070			
2.4	14.40.080			
2.5	14.40.090			
2.6	14.40.0100			
2.7	14.40.0110			
2.8	14.40.0120			
2.9	14.40.0130			
2.10	14.40.0140			
2.11	14.40.0150			
2.12	14.40.0160			
2.13	14.40.0170			
2.14 (New Section)	14.40.0175			
3.0	14.40.0180			
3.1	14.40.0190			
3.2	14.40.0200			
3.3	14.40.0210			
3.4	14.40.0220			
3.5	14.40.0230			
3.6	14.40.0240			
3.7	14.40.0250			
3.8	14.40.0260			
3.9	14.40.0270			
3.10	14.40.0280			
3.11	14.40.0290			
3.12	14.40.0300			
3.13	14.40.0310			
3.14	14.40.0320			
3.15	14.40.0330			
3.16	14.40.0340			
4.1	14.40.0350			
4.2	14.40.0360			
4.3	14.40.0370			
4.4	14.40.0380			
4.5	14.40.0390			
4.6	14.40.0400			
4.7	14.40.0410			

i

Reference Table for Codification		
Section Number in Attached Text Code Section Number for Codification		
4.8	14.40.0420	
4.9	14.40.0430	
4.10	14.40.0440	
4.11	14.40.0450	
4.12	14.40.0460	
4.13	14.40.0470	
4.14 (New Section)	14.40.0475	
5.1	14.40.0480	
5.2	14.40.0490	
5.3 (Repealed)	14.40.0500 (Repealed)	
6.1	14.40.0500	
6.2	14.40.0510	
6.3	14.40.0520	
6.4	14.40.0530	
7	14.40.0540	
8	14.40.0550	
9.1	14.40.0560	
9.2	14.40.0570	
9.3	14.40.0580	
9.4	14.40.0590	
9.5	14.40.0600	
9.6	14.40.0610	
9.7	14.40.0620	
9.8	14.40.0630	
9.9	14.40.0640	
10.1	14.40.0650	
10.2	14.40.0660	
10.3	14.40.0670	
10.4	14.40.0680	
11.1	14.40.0690	
11.2	14.40.0700	
11.3	14.40.0710	
11.4	14.40.0720	
11.5	14.40.0730	
11.6	14.40.0740	
11.7 (New Section)	14.40.0751	
11.8 (New Section)	14.40.0752	
12.1	14.40.0750	
12.2	14.40.0760	
12.3	14.40.0770	
13.1	14.40.0780	
13.2	14.40.0790	

## **TABLE OF CONTENTS**

SECTION 1	L - GENERAL PROVISIONS	. 1
1.1	Purpose and Policy	1
1.2	Administration	. 2
1.3	Definitions	. 2
1.4	Abbreviations	15
SECTION 2	2 - GENERAL REQUIREMENTS	16
2.1	Prohibited Discharge Standards	16
2.2	Federal Categorical Pretreatment Standards	19
2.3	State Requirements	19
2.4	Local Limits	19
2.5	pH Effluent Limitations Under Continuous Monitoring	
2.6	City's Right of Revision	22
2.7	Special Agreement	22
2.8	Dilution	23
2.9	Pretreatment Facilities	23
2.10	Deadline for Compliance with Applicable Pretreatment Requirements	25
2.11	Additional Pretreatment Measures	
2.12	Accidental Discharge/Slug Control Plans	26
2.13	Septic Tank Wastes	27
2.14	High Strength Wastes	28
SECTION 3	3 - WASTEWATER DISCHARGE PERMIT REQUIREMENTS	
3.0	Permit Required	29
3.1	Wastewater Discharge Permitting: Existing SIU	29
3.2	Wastewater Discharge Permitting: New Source and New User	
3.3	Wastewater Discharge Permitting: Extrajurisdictional Users	
3.4	Wastewater Discharge Permitting: Zero Discharge Permits	30
3.5	Wastewater Discharge Permitting: Middle Tier Categorical Industrial User	30
3.6	Wastewater Discharge Permitting: Non-Significant Categorical Industrial User	30
3.7	Wastewater Discharge Permit Application Contents	31
<i>3.8</i>	Signatory and Certification Requirement	34
<i>3.9</i>	Wastewater Discharge Permit Decisions	34
3.10	Wastewater Discharge Permit Contents	35
3.11	Wastewater Discharge Permit Appeals	37
3.12	Wastewater Discharge Permit Duration	38
3.13	Wastewater Discharge Permit Modification	38
3.14	Wastewater Discharge Permit Transfer	39
3.15	Wastewater Discharge Permit Revocation	39
3.16	Wastewater Discharge Permit Re-issuance	40
	_	

4.2	Final Compliance Report (Initial Compliance Report)	41
4.3	Periodic Compliance Report	42
4.4	Compliance Schedules for Meeting Applicable Pretreatment Standards	43
4.5	Notification of Significant Production Changes	44
4.6	Dangerous and Hazardous Waste Notification	44
4.7	Notice of Potential Problems, Including Accidental Spills, Slug Loadings	45
4.8	Noncompliance Reporting	45
4.9	Notification of Changed Discharge	45
4.10	TTO Reporting	46
4.11	Reports from Unpermitted Users	46
4.12	Record Keeping	46
4.13	Timing	47
4.14	Electronic Reporting	47
SECTION 5	5 - SAMPLING AND ANALYTICAL REQUIREMENTS	48
5.1	Sampling Requirements for Users	48
<i>5.2</i>	Analytical Requirements	
SECTION 6	5 - COMPLIANCE MONITORING	50
6.1	Inspection and Sampling	50
6.2	Monitoring Facilities	
6.3	Search Warrants	51
6.4	Vandalism	
SECTION 7	7 - CONFIDENTIAL INFORMATION	52
SECTION 8	3 - PUBLICATION OF USERS IN SIGNIFICANT NONCOMPLIANCE	53
SECTION 9	- ADMINISTRATIVE ENFORCEMENT REMEDIES	55
9.1	Notification of Violation (Notice of Violation, NOV)	55
9.2	Consent Orders	
9.3	Show Cause Hearing	55
9.4	Compliance Orders	56
9.5	Cease and Desist Orders	
9.6	Administrative Fines	56
9.7	Emergency Suspensions	57
9.8	Termination of Discharge (Non-Emergency)	58
9.9	Appeal Procedures	59
SECTION 1	LO - JUDICIAL ENFORCEMENT REMEDIES	63
10.1	Injunctive Relief	63
10.2	Civil Penalties	63
10.3	Criminal Prosecution	64
10.4	Remedies Non-Exclusive	64
SECTION 1	1 - SUDDIEMENTAL ENFORCEMENT ACTION	66

11.1	Performance Bonds	66
11.2	Financial Assurances	
11.3	Service Severance	66
11.4	Public Nuisances	66
11.5	Contractor Listing	66
11.6	Publication of Violations and/or Enforcement Actions	67
11.7	Penalties for Late Reports	
11.8	Payment of Outstanding Fees and Penalties	67
SECTION	N 12 - AFFIRMATIVE DEFENSES TO DISCHARGE VIOLATIONS	68
12.1	Upset	68
12.2	Prohibited Discharge Standards	
12.3	Bypass	
SECTION	N 13 - MISCELLANEOUS PROVISIONS	71
13.1	Pretreatment Charges and Fees	71
13.2	Non-Liability	72

#### **SECTION 1 - GENERAL PROVISIONS**

## 1.1 Purpose and Policy

This ordinance sets forth uniform requirements for users of the Publicly Owned Treatment Works (POTW) for the City of Everett, and enables the City to comply with all applicable State and Federal laws, including the Clean Water Act (33 USC 1251 et seq.) and the General Pretreatment Regulations (40 CFR Part 403). The objectives of this ordinance are:

- A. To prevent the introduction of pollutants into the POTW that will interfere with the operation of the POTW;
- B. To prevent the introduction of pollutants into the POTW which will pass through the POTW, inadequately treated, into receiving waters or otherwise be incompatible with the POTW;
- C. To ensure that the quality of the wastewater treatment plant biosolids is maintained at a level which allows its use and disposal in compliance with applicable statutes and regulations;
- D. To protect POTW personnel who may be affected by wastewater, wastewater solids, and biosolids in the course of their employment and to protect the general public;
- E. To improve the opportunity to recycle and reclaim wastewater and biosolids from the POTW;
- F. To enable the City to comply with its National Pollutant Discharge Elimination System permit conditions, sludge use and disposal requirements, and any other Federal or State laws to which the POTW is subject.

This ordinance shall apply to all users of the POTW. The ordinance authorizes the issuance of wastewater discharge permits and discharge authorizations; authorizes monitoring, compliance, and enforcement activities; establishes administrative review procedures; requires user reporting; and provides for the setting of fees for the equitable distribution of costs resulting from the program established herein.

#### 1.2 Administration

Except as otherwise provided herein, the Director shall administer, implement, and enforce the provisions of this ordinance. Any powers granted to or duties imposed upon the Director may be delegated by the Director to other City of Everett personnel. The Director may create administrative guidelines to implement the provisions of this ordinance.

## 1.3 Definitions

Unless a provision explicitly states otherwise, the following terms and phrases, as used in this ordinance, shall have the meanings hereinafter designated.

- A. <u>Accessible</u>. Accessible, when applied to required pretreatment monitoring or treatment equipment, shall mean direct access without the necessity of removing any panel, door, vehicle, equipment, materials, or other similar obstruction.
- B. <u>Act or "the Act"</u>. The Federal Water Pollution Control Act, also known as the Clean Water Act, as amended, 33 USC 1251 et seq.
- C. <u>Administrative Penalty (Administrative Fine)</u>. A punitive monetary charge unrelated to treatment cost, which is assessed by the Director rather than a court.
- D. <u>Applicable Pretreatment Standards</u>. For any specified pollutant, Everett prohibitive standards, Everett specific pretreatment standards (local limits), State of Washington pretreatment standards, or EPA's Categorical Pretreatment Standards (when effective); whichever standard is appropriate and most stringent.
- E. Approval Authority. The state of Washington Department of Ecology.
- F. <u>Authorized Representative of the User</u>.
  - (1) If the user is a corporation:
    - (a) The president, secretary, treasurer, or a vice-president of the corporation in charge of a principal business function, or any other person who performs similar policy or decision-making functions for the corporation; or
    - (b) The manager of one or more manufacturing, production, or operating facilities, provided, the manager is authorized to make management decisions which govern the operation of the

regulated facility including having the explicit or implicit duty of making major capital investment recommendations, and initiate and direct other comprehensive measures to assure long-term environmental compliance with environmental laws and regulations; can ensure that the necessary systems are established or actions taken to gather complete and accurate information for control mechanism requirements; and where authority to sign documents has been assigned or delegated to the manager in accordance with corporate procedures;

- (2) If the user is a partnership or sole proprietorship: a general partner or proprietor, respectively;
- (3) If the user is a Federal, State, or local governmental facility: a director or highest official appointed or designated to oversee the operation and performance of the activities of the government facility, or such director's authorized designee.
- (4) The individuals described in paragraphs 1 through 3 above may designate another authorized representative if the authorization is in writing, the authorization specifies the individual or position responsible for the overall operation of the facility from which the discharge originates or having overall responsibility for environmental matters for the company, and the written authorization is submitted to the City.
- G. <u>Average Daily Flow.</u> Average Daily Flow shall be defined as the arithmetical mean of the total Process Wastewater flow over a one (1) year period. This mean shall be calculated based on days when a discharge occurs.
- H. <u>Best Management Practices (BMPs).</u> The term Best Management Practices or BMPs means schedules of activities, prohibitions of practices, maintenance procedures, and other management practices to implement the prohibitions listed in § 403.5(a)(1) and (b). BMPs also include treatment requirements, operating procedures, and practices to control plant site runoff, spillage or leaks, sludge or waste disposal, or drainage from raw materials storage.
- I. <u>Biochemical Oxygen Demand (BOD)</u>. The quantity of oxygen utilized in the biochemical oxidation of organic matter amenable to measurement by the methods described in <u>Standard Methods for the examination of Water and Wastewater</u>, current approved edition; under standard laboratory procedures for five (5) days at 20° centigrade, usually expressed as a concentration [milligrams per liter (mg/L)].
- J. <u>Categorical Pretreatment Standard or Categorical Standard</u>. Any regulation containing pollutant discharge limits promulgated by the US EPA in accordance

- with Sections 307(b) and (c) of the Act (33 USC 1317) which apply to a specific category of users and which appear in 40 CFR Chapter I, Subchapter N, Parts 405-471.
- K. <u>Categorical User (Categorical Industrial User)</u>. A user covered by one of EPA's Categorical Pretreatment Standards.
- L. <u>Chemical Oxygen Demand (COD)</u>. A measure of the oxygen consuming capacity of inorganic and organic matter present in wastewater amenable to measurement by the methods described in <u>Standard Methods for the examination of Water and Wastewater</u>, current approved edition. COD is expressed as the amount of oxygen consumed from a chemical oxidant in mg/L during a specific test.
- M. <u>City</u>. The City of Everett, Washington.
- N. <u>Cooling Water/Non-Contact Cooling Water</u>. Water used for cooling which does not come into direct contact with any raw material, intermediate product, waste product, or finished product. Cooling water may be generated from any use, such as air conditioning, heat exchangers, cooling or refrigeration to which the only pollutant added is heat.
- O. <u>Color</u>. The optical density at the visual wave length of maximum absorption, relative to distilled water. One hundred percent (100%) transmittance is equivalent to zero (0.0) optical density.
- P. <u>Combined Sewer</u>. A sewer receiving both surface runoff and sewage.
- Q. <u>Commercial User</u>. A Commercial User is any discharger of Industrial Waste that does not meet the definition of a Significant Industrial User.
- R. <u>Composite Sample</u>. The sample resulting from the combination of individual wastewater samples taken at selected intervals based on an increment of either flow or time.
- S. <u>Daily Maximum Limit</u>. The maximum allowable discharge limit of a pollutant during a calendar day. Where Daily Maximum Limits are expressed in units of mass, the daily discharge is the total mass discharged over the course of the day (lb/day). Where Daily Maximum Limits are expressed in terms of a concentration, the daily discharge is the arithmetic average measurement of the pollutant concentration derived from all measurements taken that day
- T. <u>Day</u>. Day shall be defined as a calendar day.
- U. <u>Director</u>. The Public Works Director, or such Director's duly authorized representative.

- V. <u>Discharge Authorization</u>. A wastewater discharge permit authorizing users to discharge wastewater to the Everett POTW. These permits would be for users other than Significant Industrial Users, or Categorical Industrial Users, but still requiring a control mechanism.
- W. <u>Domestic Sewage</u>. Domestic sewage means the liquid and water borne wastes derived from ordinary living processes, free from industrial wastes, and of such character to permit satisfactory disposal, without special treatment, into the POTW.
- X. <u>Domestic User (Residential User)</u>. Any person who contributes, causes, or allows the contribution of wastewater into the City POTW that is of a similar volume and/or chemical make-up as that of a residential dwelling unit. Discharges from a residential dwelling unit typically include up to 100 gallons per capita per day at 300 mg/L of BOD and TSS.
- Y. <u>Environmental Protection Agency (EPA)</u>. The US Environmental Protection Agency or, where appropriate, the Regional Water Management Division Director, or other duly authorized official of said agency.
- Z. <u>Existing Source</u>. For a categorical industrial user, an existing source is any source of discharge, the construction or operation of which commenced prior to the publication by EPA of proposed categorical pretreatment standards, which will be applicable to such source if the standard is thereafter promulgated in accordance with Section 307 of the Act.
- AA. <u>Existing User</u>. For non-categorical users an existing user is defined as any user which is discharging wastewater prior to the effective date of this ordinance.
- AB. Fats, Oils and Grease, Total (Total FOG). The term fats, oils, and grease shall mean those components of wastewater amenable to measurement by the method 1664 described in <u>Standard Methods for the examination of Water and Wastewater</u>, current approved edition. The term Fats, Oils and Grease shall include polar and non-polar (petroleum-based) fats, oils, and grease and other components extracted from wastewater by this method.
- AC. <u>General Permit</u>. At the discretion of the Director, groups of users may be regulated under general control mechanisms if the following conditions are met. All of the facilities to be covered must:
  - (1) Involve the same or substantially similar types of operations;
  - (2) Discharge the same types of wastes;
  - (3) Require the same effluent limitations;

- (4) Require the same or similar monitoring; and
- (5) In the opinion of the Director, are more appropriately controlled under a general control mechanism than under individual control mechanisms.
- AD. <u>Grab Sample</u>. A sample which is taken from a wastestream on a one-time basis without regard to the flow in the wastestream and without consideration of time.
- AE. <u>Gravity Grease Interceptor</u>. a plumbing appurtenance identified by volume and intended to be installed in a sanitary drainage system to intercept, using gravity only, nonpetroleum fats, oils, and greases (FOG) from a wastewater discharge.
- AF. <u>Hexane Extractable Materials (HEM)</u>. Hexane extractable material (HEM, oil and grease): Material that is extracted from a sample using n-hexane and determined by this method (i.e., EPA Method 1664, Revision B: n-Hexane Extractable Material (HEM; Oil and Grease) and Silica Gel Treated n-Hexane Extractable Material (SGT-HEM; Non-polar Material) by Extraction and Gravimetry. This material includes relatively non- volatile hydrocarbons, vegetable oils, animal fats, waxes, soaps, greases, and related matter.
- AG. <u>High Strength Waste</u>. Any waters or wastewater having a concentration of BOD or Total Suspended Solids in excess of 300 mg/L, or having a concentration of Total Fats, Oil and Grease in excess of 50 mg/L.
- AH. <u>Hydromechanical Grease Removal Device (Grease Trap)</u>. a plumbing appurtenance or appliance installed in a sanitary drainage system to intercept nonpetroleum fats, oils, and greases (FOG) from a wastewater discharge; rated by flow. The design incorporates air entrainment, hydromechanical separation, interior baffling, and/or barriers in combination or separately, and one or more of the following: external flow control, with air intake (vent), directly connected
  - (1) external flow control, without air intake (vent), directly connected
  - (2) without external flow control, directly connected
  - (3) without external flow control, indirectly connected
- Al. <u>Indirect Discharge or Discharge</u>. The introduction of pollutants into the POTW from any non-domestic source regulated under Section 307(b), (c), or (d) of the Act. The discharge into the POTW is normally by means of pipes, conduits, pumping stations, force mains, constructed drainage ditches, surface water intercepting ditches, and all constructed devices and appliances appurtenant thereto.
- AJ. Industrial User. An Industrial User is any discharger of non-domestic waste.

- AK. <u>Industrial User-Specific (case-by-case basis) Local Limit:</u> MAHL protective mass based local limit not evenly distributed among users but allocated individually based upon user need.
- AL. <u>Industry specific limits</u>. Discharge limits for pollutants, applied on a case-by-case basis, for which local limits, as described in Section 2.4, have not been developed.
- AM <u>Industrial Wastewater</u>. Any and all liquid or water borne waste from industrial or commercial processes, except domestic sewage.
- AN. <u>Interceptor</u>. An interceptor is a device designed and installed to separate and retain deleterious or undesirable matter from normal wastes and permit normal liquid wastes to discharge by gravity.
- AO. Interference. A discharge which alone or in conjunction with a discharge or discharges from other sources, either: (1) inhibits or disrupts the POTW, its treatment processes or operations; (2) inhibits or disrupts its biosolids (sludge) processes, use or disposal; or (3) is a cause of a violation of the City's NPDES permit or of the prevention of sewage sludge use or disposal in compliance with any of the following statutory/ regulatory provisions or permits issued thereunder: Section 405 of the Clean Water Act; the Solid Waste Disposal Act (SWDA), including Title II commonly referred to as the Resource Conservation and Recovery Act (RCRA); any State regulations contained in any State sludge management plan prepared pursuant to Subtitle D of the SWDA; the Clean Air Act; the Toxic Substances Control Act; and the Marine Protection, Research, and Sanctuaries Act.
- AP. <u>Liquid Waste</u>. Liquid waste is the discharge from any fixture, appliance or appurtenance in connection with a plumbing system which does not receive fecal matter.
- AQ. <u>Local Limit</u>. Specific discharge limits developed and enforced by the City upon industrial or commercial facilities to implement the general and specific discharge prohibitions listed in 40 CFR 403.5(a)(1) and (b).
- AR. <u>Lower explosive limit (LEL)</u>. Minimum concentration of a vapor-air mixture, which if ignited propagates flame independent of an external source of heat.
- AS. <u>Maximum Allowable Discharge Limit (instantaneous)</u>. The maximum concentration (or loading) of a pollutant allowed to be discharged at any time.
- AT. <u>Medical Wastes</u>. Isolation wastes, infectious agents, human blood and blood products, pathological wastes, sharps, body parts, contaminated bedding, surgical wastes, potentially contaminated laboratory wastes, dialysis wastes,

unsterilized wastes generated in the rooms of patients who are isolated because of a suspected or diagnosed communicable disease, recognizable portions of the human anatomy, and untreated dental amalgam wastes containing silver and mercury.

- AU. <u>Middle Tier Categorical Industrial User</u>. At the Director's discretion, a Categorical Industrial User may be deemed a Middle Tier Categorical Industrial User (MTCIU) where the Industrial User meets all of the following conditions:
  - (1) The Industrial User's total categorical wastewater flow does not exceed any of the following:
    - (a) 0.01 percent of the design dry weather hydraulic capacity of the POTW, or 5,000 gallons per day, whichever is smaller, as measured by a continuous effluent flow monitoring device unless the Industrial User discharges in batches;
    - (b) 0.01 percent of the design dry weather organic treatment capacity of the POTW; and
    - (c) 0.01 percent of the maximum allowable headworks loading for any pollutant regulated by the applicable categorical Pretreatment Standard for which approved local limits were developed by the POTW;
  - (2) The Industrial User has not been in significant noncompliance, as defined in § 403.8(f)(2)(viii), for any time in the past two years;
  - (3) The Industrial User does not have daily flow rates, production levels, or pollutant levels that vary so significantly that decreasing the reporting requirement for this Industrial User would result in data that are not representative of conditions.
- AV. <u>North American Industry Classification System (NAICS)</u>. A classification pursuant to the North American Industry Classification System Manual issued by the United States Office of Management and Budget.

#### AW. New Source.

- (1) Any building, structure, facility, or installation from which there is (or may be) a discharge of pollutants, the construction of which commenced after the publication of proposed pretreatment standards under Section 307(c) of the Act which will be applicable to such source if such standards are thereafter promulgated in accordance with that section, provided that:
  - (a) The building, structure, facility, or installation is constructed at a site at which no other source is located; or

- (b) The building, structure, facility, or installation totally replaces the process or production equipment that causes the discharge of pollutants at an existing source; or
- (c) The production or wastewater generating processes of the building, structure, facility, or installation are substantially independent of an existing source at the same site. In determining whether these are substantially independent factors such as the extent to which the new facility is integrated with the existing plant, and the extent to which the new facility is engaged in the same general type of activity as the existing source should be considered.
- (2) Construction on a site at which an existing source is located results in a modification rather than a new source, if the construction does not create a new building, structure, facility, or installation meeting the criteria of Section (1)(b) or (c) above but otherwise alters, replaces, or adds to existing process or production equipment.
- (3) Construction of a new source as defined under this paragraph has commenced if the owner or operator has:
  - (a) Begun, or caused to begin as part of a continuous on-site construction program;
    - (i) any placement, assembly, or installation of facilities or equipment; or
    - ii) significant site preparation work including clearing, excavation, or removal of existing buildings, structures, or facilities which is necessary for the placement, assembly, or installation of new source facilities or equipment; or;
  - (b) Entered into a binding contractual obligation for the purchase of facilities or equipment which are intended to be used in its operation within a reasonable time. Options to purchase or contracts which can be terminated or modified without substantial loss, and contracts for feasibility, engineering, and design studies do not constitute a contractual obligation under this paragraph.
- AX. New User. A new user is not necessarily a new source and is defined as a user that applies to the City for a new building permit or any person who occupies an existing building and plans to discharge wastewater to the City's collection system after the effective date of this ordinance. Any person that buys an

- existing facility that is discharging non-domestic wastewater will be considered an existing user if no significant changes are made in the operation.
- AY <u>Noncontact Cooling Water</u>. Water used for cooling that does not come into direct contact with any raw material, intermediate product, waste product, or finished product.
- AZ <u>Non-Significant Categorical Industrial User</u>. A Non-Significant Categorical User is a Categorical Industrial User that meets the restrictions laid out in Section 1.3 BV (3) of this Ordinance.
- BA. <u>Pass Through</u>. A discharge which exits the POTW into waters of the United States in quantities or concentrations which, alone or in conjunction with a discharge or discharges from other sources, is a cause of a violation of any requirement of the City's NPDES permit (including an increase in the magnitude or duration of a violation).
- BB. <u>Permittee</u>. A person or user issued a wastewater Discharge Permit, Discharge Authorization, Zero Discharge Permit, or General Permit.
- BC. <u>Person</u>. Any individual, partnership, co-partnership, firm, company, corporation, association, joint stock company, trust, estate, governmental entity, or any other legal entity, or their legal representatives, agents, or assigns. This definition includes all Federal, State, or local governmental entities.
- BD. <u>pH</u>. A measure of the acidity or alkalinity of a substance, expressed in standard units, measured using the methods described in <u>Standard Methods for the Examination of Water and Wastewater</u>, current approved edition.
- BE. <u>Pollutant</u>. Any dredged spoil, solid waste, incinerator residue, sewage, garbage, sewage sludge, munitions, medical wastes, chemical wastes, biological materials, radioactive materials, heat, wrecked or discharged equipment, rock, sand, cellar dirt, agricultural and industrial wastes, and the characteristics of the wastewater [i.e., pH, temperature, TSS, turbidity, color, BOD, Chemical Oxygen Demand (COD), toxicity, or odor].
- BF. <u>Pollution</u>. The human-made or human-induced alteration of the chemical, physical, biological, and/or radiological integrity of water.
- BG. <u>Pollution Prevention</u>. Any method, action or activity that eliminates or reduces pollution prior to its creation. Pollution prevention involves the practice of source reduction. Source reduction is any practice which eliminates or reduces the amount of any hazardous substance, pollutant, or contaminant entering any waste stream or otherwise released into the environment prior to recycling, treatment, or disposal.

- BH. <u>POTW Protection Criteria</u> The Protection Criteria are set at the average uncontrollable background concentrations as determined by background sampling plus 1 standard deviation except for BOD, TSS, and Fats, Oils, and Grease (i.e., HEM), which are set at surcharge trigger levels.
- BI. <u>Pretreatment</u>. The reduction of the amount of pollutants, the elimination of pollutants, or the alteration of the nature of pollutant properties in wastewater prior to (or in lieu of) introducing such pollutants into the POTW. This reduction or alteration can be obtained by physical, chemical, or biological processes; by process changes; or by other means (except by diluting the concentration of the pollutants unless allowed by an applicable pretreatment standard).
- BJ. <u>Pretreatment Requirements</u>. Any substantive or procedural requirement related to pretreatment imposed on a user, other than a pretreatment standard.
- BK. <u>Pretreatment Standards or Standards</u>. Pretreatment standards shall mean prohibited discharge standards, categorical pretreatment standards, and local limits established by the City (POTW).
- BL. <u>Process Wastewater.</u> Process Wastewater is Industrial Waste minus Cooling Water/Non-Contact Cooling Water.
- BM. <u>Prohibited Discharge Standards or Prohibited Discharges</u>. Absolute prohibitions against the discharge of certain substances; these prohibitions appear in Sections 2.1 (A) and (B) of this ordinance.
- BN. <u>Public Sewer</u>. A sewer provided by or subject to the jurisdiction of the City. It shall also include sewers inside or outside the City boundaries that ultimately discharge into the City sanitary or combined sewer system.
- BO. <u>Publicly Owned Treatment Works (POTW)</u>. A treatment works, as defined by Section 212 of the Act (33 USC 1292) which is owned by the City. This definition includes all devices facilities, or systems used in the collection, storage, treatment, recycling, and reclamation of sewage or industrial wastes of a liquid nature and any conveyances which convey wastewater to a treatment plant. The POTW includes the City of Everett Water Pollution Control Facility.
- BP. Sanitary Flow: Sewage.
- BQ. <u>Sanitary Sewer</u>. A sewer which carries domestic and industrial waste and to which storm, surface, and ground waters are not intentionally admitted.
- BR. <u>Septic Tank Waste (Septage)</u>. Any domestic and/or residential sewage from holding tanks such as vessels, chemical toilets, campers, trailers, and septic tanks.

- BS. <u>Sewage</u>. All wastes conveyed through the collections system, includes both domestic and industrial wastewater.
- BT. <u>Sewer</u>. Any pipe, conduit, ditch, or other device used to collect and transport sewage from the generating source.
- BU. Shall, May. "Shall" is mandatory, "may" is permissive.
- BV. <u>Significant Industrial User (SIU)</u>. Except as provided in paragraphs (BV)(3) and (BV)(4) of this section, the term Significant Industrial User means:
  - (1) All Industrial Users subject to Categorical Pretreatment Standards under 40 CFR 403.6 and 40 CFR chapter I, subchapter N; and
  - (2) Any other Industrial User that:
    - (a) discharges an average of 25,000 gallons per day or more of process wastewater to the POTW (excluding sanitary, noncontact cooling and boiler blowdown wastewater);
    - (b) contributes a process wastestream which makes up 5 percent or more of the average dry weather hydraulic or organic capacity of the POTW Treatment plant; or
    - (c) is designated as such by the Director on the basis that the Industrial User has a reasonable potential for adversely affecting the POTW's operation or for violating any Pretreatment Standard or requirement (in accordance with 40 CFR 403.8(f)(6)).
  - (3) The Director may determine that an Industrial User subject to categorical Pretreatment Standards under § 403.6 and 40 CFR chapter I, subchapter N is a Non-Significant Categorical Industrial User rather than a Significant Industrial User on a finding that the Industrial User never discharges more than 100 gallons per day (gpd) of total categorical wastewater (excluding sanitary, noncontact cooling and boiler blowdown wastewater, unless specifically included in the Pretreatment Standard) and the following conditions are met:
    - (a) The Industrial User, prior to the Director's finding, has consistently complied with all applicable categorical Pretreatment Standards and Requirements;
    - (b) The Industrial User annually submits the certification statement required in § 403.12(q) together with any additional information necessary to support the certification statement; and

- (c) The Industrial User never discharges any untreated concentrated wastewater.
- (4) Upon a finding that an Industrial User meeting the criteria in paragraph (BV)(2)(b) of this section has no reasonable potential for adversely affecting the POTW's operation or for violating any Pretreatment Standards or requirement, the Director may at any time, on his own initiative or in response to a petition received from an Industrial User or POTW, and in accordance with 40 CFR 403.8(f)(6), determine that such Industrial User is not a Significant Industrial User.
- BW. <u>Sludge</u>. Residual solids resulting from the treatment of wastewater.
- BX. <u>Slug Load or Slug Discharge</u>. Any discharge at a flow rate or concentration which could cause a violation of the discharge standards in Section 2.1 through 2.4 of this ordinance. A Slug Discharge is any Discharge of a non-routine, episodic nature, including but not limited to an accidental spill or a non-customary batch Discharge, which has a reasonable potential to cause Interference or Pass Through, or in any other way violate the POTW's regulations, Local Limits or Permit conditions.
- BY. <u>Standard Industrial Classification (SIC) Code</u>. A classification pursuant to the Standard Industrial Classification Manual issued by the United States Office of Management and Budget.
- BZ. <u>State</u>. The State of Washington.
- CA. <u>Storm Water</u>. Any flow occurring during or following any form of natural precipitation, and resulting from such precipitation, including snowmelt.
- CB. <u>Total Suspended Solids (TSS)</u>. The total suspended matter that floats on the surface of, or is suspended in, water, wastewater, or other liquid, and which is removable by laboratory filtering and is amenable to measurement by the methods described in <u>Standard Methods for the examination of Water and Wastewater</u>, current approved edition.
- CC. <u>Toxic Pollutant</u>. One of the pollutants, or combination of those pollutants, listed as toxic in regulations promulgated by EPA under Section 307 (33 USC 1317) of the Act, or other pollutants as may be promulgated.
- CD. <u>Treatment Plant Effluent</u>. The discharge from the POTW into waters of the United States.
- CE. User. Any source of discharge into the sewer.
- CF. Wastewater. Liquid and water-carried industrial wastes and sewage from

residential dwellings, commercial buildings, industrial and manufacturing facilities, and institutions, whether treated or untreated, which are contributed to the POTW.

- CG. <u>Wastewater Discharge Permit (Industrial Wastewater Discharge Permit, Discharge Authorization, General Permit)</u>. An authorization or equivalent control document issued by the City to users discharging wastewater to the POTW. The permit may contain appropriate pretreatment standards and requirements as set forth in this ordinance.
- CH. <u>Wastewater Treatment Plant or Treatment Plant or Pollution Control Facility</u>. That portion of the POTW which is designed to provide treatment of municipal sewage and industrial waste.
- CI. <u>Zero Discharge Permit.</u> A Permit for a Categorical User that operates its processes so that no Industrial Waste is discharged to the POTW.

The use of the singular shall be construed to include the plural and the plural shall include the singular as indicated by the context of its use.

# 1.4 Abbreviations

The following abbreviations shall have the designated meanings:

AKART	All Known Available and Reasonable Technology
ASPP	Accidental Spill Prevention Plan
BMPs	Best Management Practices
BOD	Biochemical Oxygen Demand
CIUCateg	orical Industrial User
CFR	Code of Federal Regulations
COD	Chemical Oxygen Demand
CSO	Combined Sewer Overflow
CWAClean	Water Act
EPA	.US Environmental Protection Agency
FOG	Fats, Oils, and Grease
GPD	Gallons Per Day
HEM	.Hexane Extractable Material
L	liter
LEL	Lower Explosive Limit
mg	_
mg/L	milligrams per liter.
MGDMilli	on Gallons per Day
	Non-Significant Categorical Industrial User
NPDES	National Pollutant Discharge Elimination System
O&M	Operation and Maintenance
POTW	Publicly Owned Treatment Works
	Resource Conservation and Recovery Act
SIC	Standard Industrial Classifications
SIU	Significant Industrial User
SCP	Slug Control Plan
SSO	Sanitary Sewer Overflow
	Solid Waste Disposal Act (42 USC 6901, et seq.)
TSS	Total Suspended Solids
USC	United States Code

## **SECTION 2 - GENERAL REQUIREMENTS**

## 2.1 Prohibited Discharge Standards

- A. General Prohibitions: No user shall introduce or cause to be introduced into the POTW any pollutant or wastewater which causes pass through or interference. These general prohibitions apply to all users of the POTW whether or not they are subject to categorical pretreatment standards or any other National, State, or local pretreatment standards or requirements.
- B. Specific Prohibitions: No user shall introduce or cause to be introduced into the POTW the following pollutants, substances, or wastewater:
  - (1) Pollutants which create a fire or explosive hazard in the POTW, including, but not limited to, wastestreams with a closed-cup flashpoint of less than 140°F (60°C) using the test methods specified in 40 CFR 261.21;
  - (2) Wastewater having a pH less than 5.0 s.u. or more than 11.0 s.u. (unless in compliance with Section 2.5 of this Ordinance) or otherwise causing corrosive structural damage to the POTW or equipment;
  - (3) Solid or viscous substances in amounts which will cause obstruction of the flow in the POTW resulting in interference but in no case solids greater than one quarter inch (1/4");
  - (4) Pollutants, including oxygen-demanding pollutants (BOD, COD, etc.), released in a discharge at a flow rate and/or pollutant concentration which, either singly or by interaction with other pollutants, will cause interference with the POTW;
  - (5) Wastewater having a temperature which will inhibit biological activity in the treatment plant resulting in interference, but in no case wastewater which causes the temperature at the introduction into the treatment plant to exceed 104°F (40°C) unless the Approval Authority, upon the request of the POTW, approves alternate temperature limits;
  - (6) Petroleum oil, non-biodegradable cutting oil, or products of mineral oil origin, in amounts that will cause interference or pass through;
  - (7) Pollutants which result in the presence of toxic gases, vapors, or fumes within the POTW in a quantity that may cause acute worker health and safety problems;
  - (8) Trucked or hauled pollutants (except Domestic Sewage or Septic Tank

- Wastes) unless authorized by the director, and at discharge points designated by the City.
- (9) Noxious or malodorous liquids, gases, solids, or other wastewater which, either singly or by interaction with other wastes, are sufficient to create a public nuisance or a hazard to life, or to prevent entry into the sewers for maintenance or repair;
- (10) Wastewater which imparts color which cannot be removed by the treatment process, such as, but not limited to, dye wastes and vegetable tanning solutions, which consequently imparts color to the treatment plant's effluent, thereby violating the City's NPDES permit. Color (in combination with turbidity) shall not cause the treatment plant effluent to reduce the depth of the compensation point for photosynthetic activity by more than ten percent (10%) from the seasonably established norm for aquatic life;
- (11) Wastewater containing any radioactive wastes or isotopes except as specifically approved by the Director in compliance with applicable State or Federal regulations;
- (12) Storm water, surface water, ground water, artesian well water, roof runoff, subsurface drainage, swimming pool drainage, condensate, deionized water, non-contact cooling water, and unpolluted wastewater, unless specifically authorized by the Director;
- (13) Any sludges, screenings, or other residues from the pretreatment of industrial or commercial wastes from industrial or commercial processes, or residues from cleaning wetwells or sewers, except as authorized by the Director;
- (14) Medical wastes, except as specifically authorized by the Director;
- (15) Wastewater, alone or in conjunction with other sources, that causes the treatment plant's effluent to fail a toxicity test;
- (16) Detergents, surface-active agents, or other substances in amounts which may cause excessive foaming in the POTW;
- (17) Any liquid, solids, or gases which by reason of their nature or quantity are, or may be, sufficient either alone or by interaction with other substances to cause fire or explosion or be injurious in any other way to the POTW or to the operation of the POTW. At no time shall two (2) successive readings on an explosion meter, at the point of discharge into the system (or at any point in the system), be more than five (5%) per

- cent nor any single reading over ten (10%) per cent of the lower explosive limit (LEL) of the meter.
- (18) Animal guts or tissues, paunch manure, bones, hair, hides or fleshings, entrails, whole blood, feathers, ashes, cinders, sand, spent lime, stone or marble dusts, metal, glass, straw, shavings, grass clippings, rags, spent grains, spent hops, waste paper, wood, plastics, gas, tar asphalt residues, residues from refining or processing of fuel or lubricating oil, mud, glass grinding or polishing wastes, or other similar wastes in amounts that cause interference in the POTW or obstruction to the sewer system.
- (19) Any substance which will cause the POTW to violate its NPDES and/or other disposal system permits.
- (20) Any wastewater, which in the opinion of the Director can cause harm either to the sewers, sewage treatment process, or equipment; have an adverse effect on the receiving stream; or can otherwise endanger life, limb, public property, or constitute a nuisance, unless allowed under special agreement by the Director (except that no special waiver shall be given from categorical pretreatment standards).
- (21) The contents of any tank or other vessel owned or used by any person in the business of collecting or pumping sewage, effluent, septage, or other wastewater unless said person has first obtained testing and approval as may be generally required by the City of Everett and paid all fees assessed for the privilege of said discharge.
- (22) Any hazardous or dangerous wastes as defined in rules published by the State of Washington (WAC 173-303), in EPA rules 40 CFR Part 261, or Hazardous Waste Pharmaceuticals.
- (23) Persistent pesticides and/or pesticides regulated by the Federal Insecticide Fungicide Rodenticide Act (FIFRA)
- (24) Any slug load.
- (25) Any substance which may cause the POTW's effluent or treatment residues, sludges, or scums to be unsuitable for reclamation and reuse, or to interfere with the reclamation process.
- (26) Fats, oils and grease in amounts that may cause obstructions or maintenance problems in the collection/conveyance system, or interference in the POTW.
- (27) The use of the treatment and controls located at the POTW for wastewater treatment required by a National Emission Standards for

Hazardous Air Pollutants for Source Categories (NESHAP) under 40 CFR Part 63 is prohibited. The discharge of any untreated wastewater regulated by a NESHAP also is prohibited. The POTW does not and will not accept a NESHAP regulated waste stream nor provide treatment or controls as an agent for any Industrial User within the meaning of 40 CFR Part 63, including but not limited to 40 CFR § 63.1595.

(28) Any toxic substances in amounts exceeding standards promulgated by the United States Environmental Protection Agency pursuant to Section 307(a) of the Act and the EPA's priority pollutant list.

Pollutants, substances, or wastewater prohibited by this section shall not be processed or stored in such a manner that they are likely to be discharged to the POTW unless the user has in place an accidental spill prevention plan (ASPP)/slug control plan.

## 2.2 Federal Categorical Pretreatment Standards

The National Categorical Pretreatment Standards found at 40 CFR Chapter I, Subchapter N, Parts 405-471 are incorporated herein by reference as if set forth in full in this ordinance.

- A. Where a categorical Pretreatment Standard is expressed only in terms of either the mass or the concentration of a pollutant, the Director may impose equivalent concentration or mass limits.
- B. Where a categorical Pretreatment Standard is expressed only in terms of mass of pollutant per unit of production, the Director may impose equivalent concentration or mass limits.

#### 2.3 State Requirements

State requirements and limitations on discharges to the POTW shall be met by all users which are subject to such standards in any instance in which they are more stringent than federal requirements and limitations, or those in this ordinance or other applicable ordinances.

#### 2.4 Local Limits

- A. The director is authorized to establish Local Limits pursuant to 40 CFR 403.5(c).
- B. The following technically based Maximum Allowable Headworks Loadings [MAHL] have been established to protect against pass through and interference. The MAHLs apply to all wastewater entering the WPCF (industrial, commercial, and residential). The Maximum Allowable Industrial Loading (MAIL) is calculated based

upon the MAHL minus the background (commercial and residential) loadings and a safety factor. The City uses a combination of uniform concentration and industry specific mass based limits to establish local limits that ensure MAHLs are not exceeded. Any applicable local limits shall be included in individual Industrial User wastewater discharge permits or other control mechanisms. At no time shall the City allocate loading to permitted users that causes these loadings to be exceeded:

Parameter	lb/day
Arsenic (total)	5.29
Cadmium (total)	2.01
Chromium (total)	173
Copper (total)	5.91
Cyanide (total)	0.87
Lead (total)	10.4
Mercury (total)	0.11
Molybdenum (total)	7.63
Nickel (total)	7.59
Selenium (total)	7.48
Silver (total)	0.93
Zinc (total)	108
Biochemical Oxygen Demand <sup>1</sup>	83,000
Total Suspended Solids <sup>1</sup>	89,000

#### Established by NPDES Permit

Local limits apply at the point where the wastewater is discharged to the POTW (end of the pipe). Categorical pretreatment standards apply at the end of the process. However, the Director may elect to have local limits apply after pretreatment and/or prior to mixing with dilution flows.

All concentrations for metallic substances are for total metal unless indicated otherwise.

Where a user is subject to a categorical pretreatment standard and a local limit for a given pollutant, the more stringent limit or applicable pretreatment standard shall apply.

The City Council authorizes the Director to revise local limits.

#### C. POTW Protection Criteria

The City has established the following criteria (in mg/L) based upon technical evaluation of the background loading. Any user proposing to discharge wastewater that contains the following pollutants at or below the listed concentration shall not be required to discharge under an industrial wastewater discharge permit unless the discharge meets

other SIU requirements or is otherwise required by the director at the director's discretion. If the proposed discharge exceeds the following concentration criteria, the user shall submit a completed Permit Application to the Industrial Pretreatment Section. These concentrations apply at the point where the wastewater is discharged to the POTW collection system. The City reserves the right to modify, add or delete Protection Criteria as the City deems necessary:

Parameter	mg/l
Arsenic (total)	0.001
Cadmium (total)	0.0002
Chromium (total)	0.002
Copper (total)	0.034
Cyanide (total)	0.003
Lead (total)	0.003
Mercury (total)	0.00007
Molybdenum (total)	0.0009
Nickel (total)	0.003
Selenium (total)	0.0005
Silver (total)	0.0003
Zinc (total)	0.139
Biochemical Oxygen Demand	300
Total Suspended Solids	300
HEM (total)	50

#### D. Local Pollutant Allocations

The City is also authorized to develop and issue Local Pollutant Allocations for the control of pollutants of concern other than those listed in Section 2.4.B. These Local Pollutant Allocations are also based on the Maximum Allowable Headworks Loadings, Pass-Through, Interference, or protection of health and are implemented through the issuance of a industrial wastewater discharge permits or other control mechanisms. A Local Pollutant Allocation is not a local limit, it is an industry specific limit for a pollutant that is not included with the local limits or Categorical Pretreatment Standards but still have the potential to adversely affect the plant or collections system and therefore must be monitored.

These limits apply at the point where the wastewater is discharged to the POTW (end of the pipe).

#### E. Best Management Practices

The Director may develop Best Management Practices (BMPs), for inclusion in an ordinance or in individual wastewater discharge permits, to implement Local Limits and the requirements of Section 2.1. These BMPs may include but are not limited to types or

methods of pretreatment technology to be used, methods of source control, minimum maintenance requirements, dragout, spill, and pollution prevention practices, or other requirements as deemed necessary. The Director may establish BMPs in lieu of numerical limits for non-categorical users.

## 2.5 pH Effluent Limitations Under Continuous Monitoring

- A. Where a permittee continuously monitors the pH of wastewater discharged to the City's sewer system pursuant to a requirement in their discharge permit, the permittee shall maintain the pH of such wastewater within the range set forth in the permit, except excursions from the range are permitted subject to the following limitations:
  - (1). The total time during which the pH values are outside the required range of pH values shall not exceed two (2) hours in any calendar month; and
  - (2). No individual excursion from the allowable range of pH values shall exceed 15 minutes.
- B. At no time will the pH value be outside the allowable range of pH values by more than one pH unit.
- C. At no time shall the pH be less than 5.0 pH units.
- D. All batch discharges shall be in compliance with the allowable pH range.
- E. For the purposes of this Ordinance, an excursion is an unintentional and temporary incident in which the pH value of the discharged wastewater exceeds the range set forth in the user's discharge permit.
- F. Temporary pH value excursions that comply with the provisions or this section of the Ordinance will not be considered violations of the user's discharge permit but shall be reported in the pretreatment self-monitoring report.

# 2.6 City's Right of Revision

The City reserves the right to establish, by ordinance or in wastewater discharge permits, differing standards or requirements on discharges to the POTW.

# 2.7 Special Agreement

The City reserves the right to enter into special agreements with users setting out special terms under which they may discharge to the POTW. In no case will a special

agreement waive compliance with a categorical pretreatment standard or federal pretreatment requirement. However, the user may request a net gross adjustment to a categorical standard in accordance with 40 CFR 403.15. They may also request a variance from the categorical pretreatment standard from the Approval Authority in accordance with 403.13.

#### 2.8 Dilution

A user shall not increase the use of process water, or in any way attempt to dilute a discharge, as a partial or complete substitute for adequate treatment to achieve compliance with an applicable pretreatment standard or requirement unless expressly authorized by an applicable pretreatment standard or requirement. The Director may impose mass limitations on users which the Director believes may be using dilution to meet applicable pretreatment standards or requirements, or in other cases when the imposition of mass limitations is appropriate.

#### 2.9 Pretreatment Facilities

A. General: Users shall provide all known, available, and reasonable methods of prevention, control, and treatment (AKART) as required to comply with this ordinance and shall achieve compliance with all applicable pretreatment standards and requirements set out in this ordinance within the time limitations specified by the EPA, the State, or the Director, whichever is more stringent.

Any facilities required to pretreat wastewater to a level acceptable to the City shall be provided, operated, and maintained at the user's expense.

The Director may establish Best Management Practices (BMPs) for particular groups of users.

When required by the Director, an engineering report as required by, and complying with, WAC 173-240, including detailed plans showing the pretreatment facilities and operating procedures shall be submitted to the City for review, and shall be acceptable to the City before construction of the facility. The review of such plans and operating procedures will in no way relieve the user from the responsibility of modifying the facility as necessary to produce an acceptable discharge to the City under the provisions of this ordinance.

Within ninety (90) days after the completion of the wastewater pretreatment facility, the discharger shall furnish as built drawings and its operations and maintenance procedures. Any subsequent significant changes in the pretreatment facility or method of operation shall be reported to and approved by the Director prior to the initiation of the changes.

- New sources, and new users determined to be SIU's must have pretreatment facilities installed and operating, if required, prior to discharge.
- B. Grease Control/Pretreatment: Measures to control, recycle, remove or pretreat for grease that are applicable to food service establishments, other than SIUs, are subject to the Grease Control Ordinance, chapter 14.42 of the Everett Municipal Code, as may be amended.
- C. Other Interceptors: Dischargers who operate automatic and coin-operated laundries, car washes, filling stations, commercial garages or similar businesses having any type of washing facilities (including pressure washing and steam cleaning) or any other dischargers producing grit, sand, oils, lint, or other materials which have the potential of causing partial or complete obstruction of the building side sewer or other areas in the POTW shall, upon order of the Director, install approved interceptors, oil/water separators, or tanks in accordance with specifications adopted by the City of Everett such that excessive amounts of oil, sand and inert solids are effectively prevented from entering the POTW.
- D. Amalgam Separators: Users who generate dental amalgam wastes containing silver or mercury shall install amalgam separators on all equipment that might carry amalgam waste to the sanitary sewer and service in accordance with 40 CFR Part 441.
- E. Installation and Maintenance: All pretreatment devices including oil/water separators, settling tanks, separators and grit traps shall be properly installed, maintained and operated by the discharger at the discharger's own expense. The installation shall be kept in continuous operation at all times and shall be maintained to provide efficient operation. Cleaning must be performed by a service contractor qualified to perform such cleaning, or in a manner approved by the Director. All material removed shall be disposed of in accordance with all state and federal regulations. Records and certification of maintenance shall be made readily available to the Director for review and inspection, and must be maintained onsite, accessible for review by the Director, for a minimum of three (3) years.

If a failure to maintain settling tanks, grit traps, grease interceptors, or oil/water separators results in partial or complete blockage of the building sewer, private sewer system discharging to the City Sewer System, or other parts of the City Sewer System, or adversely affects the treatment or transmission capabilities of the POTW, or requires excessive maintenance by the City, or poses a possible health hazard, the discharger responsible for the facilities shall be subject to the remedies herein, including cost recovery, enforcement and penalties.

## 2.10 Deadline for Compliance with Applicable Pretreatment Requirements

Compliance by existing sources (categorical users) covered by Categorical Pretreatment Standards shall be within three (3) years of the date the Standard is effective unless a shorter compliance time is specified in the appropriate Standard. The City shall establish a final compliance deadline date for any categorical user when the local limits for said user are more restrictive than EPA's Categorical Pretreatment Standards. The City may establish a final compliance deadline date for any existing user not covered by Categorical Pretreatment Standards.

New source dischargers, and new users that are determined to be SIU's, are required to comply with applicable pretreatment standards within the shortest feasible time (not to exceed 90 days from the beginning of discharge). New Sources, and new users that are determined to be SIU's, shall install and have in operating condition, and shall start-up all pollution control equipment required to meet applicable pretreatment standards before beginning to discharge.

Any wastewater discharge permit issued to a categorical user shall not contain a compliance date beyond any deadline date established in EPA's Categorical Pretreatment Standards. Any other existing user that is considered to be an SIU, or a categorical user that must comply with a more stringent local limit, which is in noncompliance with any local limits shall be provided with a compliance schedule to ensure compliance within the shortest time feasible.

#### **2.11** Additional Pretreatment Measures

- A. When deemed necessary, the Director may require users to restrict their discharge during peak flow periods, designate that certain wastewater be discharged only into specific sewers, relocate and/or consolidate points of discharge, separate sewage wastestreams from industrial wastestreams, and such other conditions as may be necessary to protect the POTW and determine the user's compliance with the requirements of this ordinance.
- B. When determined necessary by the Director, a user discharging into the POTW shall install and maintain, on such user's property and at such user's expense, a suitable storage and flow-control facility to insure equalization of flow. The Director may require the facility to be equipped with alarms and a rate of discharge controller, the regulation of which shall be determined by the Director. A wastewater discharge permit or Discharge Authorization may be issued solely for flow equalization.

- C. Grease, oil, and sand interceptors shall be provided when, in the opinion of the Director, they are necessary for the proper handling of wastewater containing excessive amounts of grease and oil, or sand; except that such interceptors shall not be required for residential users. All interception units shall be of type and capacity approved by the Director and shall be so located to be easily accessible for cleaning and inspection. Such interceptors shall be inspected, cleaned, and repaired regularly, as needed, by the user at the user's expense.
- D. Users with the potential to discharge flammable substances may be required to install and maintain an approved combustible gas detection system.

## 2.12 Accidental Discharge/Slug Control Plans

The Director may require any user to develop and implement an accidental discharge/slug control plan. Where deemed necessary by the City, facilities to prevent accidental discharge or slug discharges of pollutants shall be provided and maintained at the user's cost and expense.

An accidental spill prevention plan (ASPP)/slug control plan showing facilities and operating procedures to provide this protection shall be submitted to the City for review and approval before implementation. The City shall determine which user is required to develop a plan and require said plan to be submitted within 90 days after notification by the City. Each user shall implement its ASPP as submitted or as modified after such plan has been reviewed and approved by the City. Review and approval of such plans and operating procedures by the City shall not relieve the user from the responsibility to modify its facility as necessary to meet the requirements of this Section.

- A. Any user required to develop and implement an accidental discharge/control slug plan shall submit a plan which addresses, at a minimum, the following:
  - (1) Description of discharge practices, including non-routine batch discharges;
  - (2) Description of stored chemicals;
  - (3) Procedures for immediately notifying the Director of any accidental or slug discharge. Such notification must also be given for any discharge which would violate any of the standards in Sections 2.1 through 2.4 of this ordinance; and
  - (4) Procedures to prevent adverse impact from any accidental or slug discharge. Such procedures include, but are not limited to, inspection and maintenance of storage areas, handling and transfer of materials, loading

and unloading operations, control of plant site runoff, worker training, building of containment structures or equipment, measures for containing toxic organic pollutants (including solvents), and/or measures and equipment for emergency response.

- B. Users shall notify the Everett Water Pollution Control Facility immediately upon the occurrence of a slug or accidental discharge of substances regulated by this ordinance. The notification shall include location of discharge, date and time thereof, type of waste, concentration and volume, and corrective actions. Any affected user shall be liable for any expense, loss, or damage to the POTW, in addition to the amount of any fines imposed on the City on account thereof under state or federal law.
- C. Within five (5) days following an accidental discharge, the user shall submit to the Director a detailed written report describing the cause of the discharge and the measures to be taken by the user to prevent similar future occurrences. Such notification shall not relieve the user of any expense, loss, damage, or other liability which may be incurred as a result of damage to the POTW, fish kills, or any other damage to person or property; nor shall such notification relieve the user of any fines, civil penalties, or other liability which may be imposed by this Ordinance or other applicable law.
- D. When required by the Director, signs shall be permanently posted in conspicuous places on the user's premises advising employees whom to call in the event of a slug or accidental discharge. Contact procedure shall include after hours, holidays, and weekends. Employers shall instruct all employees who may cause or discover such a discharge with respect to emergency notification procedures.

# 2.13 Septic Tank Wastes

- A. Hauled residential/domestic septic tank waste may be introduced into the POTW only at a designated receiving structure within the treatment plant area, or at a site authorized by the Director and at such times as are established by the Director. Only residential/domestic septic waste shall be accepted and such wastes shall not violate any requirements established or adopted by the City, except as authorized by the Director.
- B. Permits for individual vehicles to use such facilities shall be issued by the City. Wastewater discharge permits may be issued to each septage hauling business and may encompass more than one vehicle. Permits may be revoked at the director's discretion if the permittee violates any condition found therein. Permits may be transferred in accordance with the conditions outlined within the permit.

- C. No load may be discharged without prior consent of the Director. The Director may collect samples of each hauled load to ensure compliance with applicable pretreatment standards. The Director may require the hauler to provide a waste analysis of any load prior to discharge. Hauled waste that is not strictly domestic in origin shall be refused and the permit may be revoked until such time the permittee provides disposal documentation of the refused load to the Director.
- D. Septage haulers shall utilize a waste tracking form (manifest) provided by the Director for every load discharged. Any discharge without a manifest form is an unauthorized discharge and the hauler will receive an administrative fine of not less than \$5,000, and may have their discharge permit revoked or suspended. Violations of permit conditions or this ordinance shall subject the permittee to the enforcement actions detailed in the permit and this ordinance.
- E. Wastewater discharge permit fees for liquid waste haulers shall be established and charged in addition to other license and volume fees charged under the Everett Municipal Code, as amended. License and volume fees shall be established as part of the user fee system utilizing the principles established in Section 13 of this ordinance.

## 2.14 High Strength Wastes

The discharge of Industrial wastewater with compatible pollutants shall be subject to the review and approval of the Director. If the Director determines that the mass loading and/or volume of flow is excessive, and the wastewater will not cause interference, pass through, or reduce the efficiency of the wastewater treatment system, the wastewater may be allowed subject to additional charges. Permitted industrial user discharges exceeding the BOD, TSS, and HEM concentrations listed in Sec 2.4.C shall be subject to high strength surcharges. These additional charges will be based on the average treatment cost of the pollutant and flow. The additional charge shall be applied to the total load and flow discharged.

## **SECTION 3 - WASTEWATER DISCHARGE PERMIT REQUIREMENTS**

#### 3.0 Permit Required

No SIU shall discharge wastewater into the POTW without first obtaining a wastewater discharge permit from the Director. Any violation of the terms and conditions of a wastewater discharge permit shall be deemed a violation of this ordinance and subjects the wastewater discharge permittee to the sanctions set out in this ordinance. Obtaining a wastewater discharge permit does not relieve a permittee of its obligation to comply with all Federal and State pretreatment standards or requirements or with any other requirements of Federal, State, and local law.

The Director may require other users, including liquid waste haulers, to obtain wastewater discharge permits (as necessary) to carry out the purposes of this ordinance.

## 3.1 Wastewater Discharge Permitting: Existing SIU

Any SIU that was discharging wastewater into the POTW prior to the effective date of this ordinance without an effective wastewater discharge permit and that wishes to continue such discharges in the future shall, within 60 days after notification by the Director submit a permit application to the City in accordance with Section 3.7 of this ordinance, and shall not cause or allow discharges to the POTW to continue after 90 days of the effective date of this ordinance except in accordance with a wastewater discharge permit issued by the Director.

The City's notification to SIUs covered by categorical pretreatment standards will be in reasonable time to ensure that the SIUs complies with the 180-day submittal deadline date established in 40 CFR §403.12 (b).)

## 3.2 Wastewater Discharge Permitting: New Source and New User

At least 90 days prior to the anticipated start-up, new sources, sources that become a user subsequent to the promulgation of an applicable categorical pretreatment standard, and new users that are determined to be SIU, shall apply for a wastewater discharge permit and will be required to submit to the City at least the information listed in paragraphs (A)-(E) of Section 3.7 of this ordinance. A new source, or new user that is determined to be a SIU, cannot discharge without first receiving a wastewater discharge permit from the City. New sources, and new users that are determined to be SIU's, shall also be required to include in their application information on the method of

pretreatment the user intends to use to meet applicable pretreatment standards. New Sources, and new users that are determined to be SIU's, shall give estimates of the information requested in paragraphs (D) and (E) of Section 3.7 of this ordinance.

## 3.3 Wastewater Discharge Permitting: Extrajurisdictional Users

Any existing user located beyond the City limits required to obtain a wastewater discharge permit shall submit a wastewater discharge permit application as outlined in Section 3.1 of this ordinance.

New Sources and new users that are determined to be SIU's, located beyond the City limits required to obtain a wastewater discharge permit shall comply with Section 3.2 of this ordinance.

## 3.4 Wastewater Discharge Permitting: Zero Discharge Permits

Any categorical user that operates its regulated processes so that no industrial waste is discharged to the POTW shall request that a zero-discharge permit be issued by the City. To be eligible for a zero discharge permit the user shall demonstrate to the Director's satisfaction that no industrial waste will be discharged and shall either permanently seal all accesses to the POTW other than those required for disposal of domestic sewage or install shutoff devices that will accept City installed, tamper evident seals. Breaking this seal without prior authorization by the Director shall be a violation of the zero-discharge permit and this ordinance.

# 3.5 Wastewater Discharge Permitting: Middle Tier Categorical Industrial User

Any categorical user that operates its regulated processes so that it complies with the requirements in Section 1.3 AU of this Ordinance may request that it be designated a Middle Tier Categorical Industrial User. If the Director agrees with that request, the User's discharge permit will be modified to incorporate the applicable provisions of 40 CFR 403. If, at any time, the User no longer complies with the requirements in Section 1.3 AU of this Ordinance, it shall immediately notify the Director and comply with the backup monitoring and reporting requirements contained in the permit.

# 3.6 Wastewater Discharge Permitting: Non-Significant Categorical Industrial User

Any categorical user that operates its regulated processes so that it complies with the

requirements in Section 1.3 BV (3) of this Ordinance may request that it be designated a Non-Significant Categorical Industrial User. If the Director agrees with that request, the User will be issued a Non-Significant Categorical Industrial User permit and shall comply with the requirements of that permit. If, at any time, the User no longer complies with the requirements in Section 1.3 BV (3) of this Ordinance, it shall immediately notify the Director and comply with the backup reporting requirements contained in the permit.

## 3.7 Wastewater Discharge Permit Application Contents

All users required to obtain a wastewater discharge permit must submit, at a minimum, the following information. The Director shall approve a form to be used as a permit application. Categorical users submitting the following information shall have complied with 40 CFR 403.12(b).

- A. <u>Identifying information.</u> The user shall submit the name and address of the facility including the name of the operator and owners, contact information, description of activities, facilities, and plant production processes on the premises;
- B. <u>Permits.</u> The user shall submit a list of any environmental control permits held by or for the facility;
- C. Description of operations. The user shall submit a brief description of the nature, average rate of production, and Standard Industrial Classification of the operation(s) carried out by such Industrial User, including a list of all raw materials and chemicals used or stored at the facility which are, or could accidentally or intentionally be, discharged to the POTW; number and type of employees; hours of operation; each product produced by type, amount, process or processes, and rate of production; type and amount of raw materials processed (average and maximum per day) and the time and duration of discharges. This description should also include a schematic process diagram which indicates points of discharge to the POTW from the regulated or manufacturing processes. Disclosure of site plans, floor plans, mechanical and plumbing plans and details to show all sewers, sewer connections, inspection manholes, sampling chambers and appurtenances by size, location and elevation.

#### D. <u>Flow Measurement.</u>

#### (1) Categorical User:

The user shall submit information showing the measured average daily and maximum daily flow, in gallons per day, to the POTW from each of

#### the following:

- (a) Regulated or manufacturing process streams; and
- (b) Other streams as necessary to allow use of the combined wastestream formula of 40 CFR 403.6(e) if required.

#### (2) Non-Categorical User

The user shall submit information showing the measured average daily and maximum daily flow, in gallons per day, to the POTW from each of the following:

(a) Total process flow, wastewater treatment plant flow, total plant flow or individual manufacturing process flow as required by the Director.

The City may allow for verifiable estimates of these flows where justified by cost or feasibility considerations.

#### E. <u>Measurements of pollutants.</u>

- (1) Categorical User:
  - (a) The user shall identify the applicable pretreatment standards for each regulated or manufacturing process.
  - (b) In addition, the user shall submit the results of sampling and analysis identifying the nature and concentration (and/or mass) where required by the Categorical Pretreatment Standard or as required by the City of regulated pollutants (including standards contained in Sections 2.1 through 2.4 of this ordinance, as appropriate) in the discharge from each regulated or manufacturing process. Both daily maximum and average concentration (and/or mass, where required) shall be reported. The sample shall be representative of daily operations and shall conform to sampling and analytical procedures outlined in Section 5.
  - (c) The user shall take a minimum of one representative sample to compile that data necessary to comply with the requirements of this paragraph.
  - (d) Where an alternate concentration or mass limit has been

calculated in accordance with 40 CFR 403.6(e) for a categorical user covered by a categorical pretreatment standard this adjusted limit along with supporting data shall be submitted as part of the application.

#### (2) Non-Categorical User

- (a) The user shall identify the applicable pretreatment standards for its wastewater discharge.
- (b) In addition, the user shall submit the results of sampling and analysis identifying the nature and concentration (or mass where required by the City) of regulated pollutants contained in Sections 2.1 through 2.4 of this ordinance, as appropriate in the discharge. Both daily maximum and average concentration (or mass, where required) shall be reported. The sample shall be representative of daily operations and shall conform to sampling and analytical procedures outlined in Section 5 of this ordinance.
- (c) The user shall take a minimum of one representative sample to compile that data necessary to comply with the requirements of this paragraph.
- (d) Where the Director developed alternate concentration or mass limits because of dilution this adjusted limit along with supporting data shall be submitted as part of the application.
- F. <u>Certification</u>. A statement, reviewed by an authorized representative of the user and certified by a qualified professional as outlined in Section 3.8 of this ordinance, indicating whether the applicable Pretreatment Standards are being met on a consistent basis, and, if not, whether additional operation and maintenance (O and M) and/or additional pretreatment is required for the user to meet the applicable Pretreatment Standards and Requirements;
- G. <u>Compliance Schedule</u>. If additional pretreatment and/or Operation & Maintenance activities will be required to meet the applicable Pretreatment Standards, the City will establish the shortest schedule by which the user will provide such additional pretreatment and/or Operation & Maintenance activities. The schedule shall conform with the requirements of Section 4.4. The completion date in this schedule shall not be later than the compliance date established pursuant to Section 2.10 of this ordinance.
  - (1) Where the user's categorical Pretreatment Standard has been modified by a removal allowance (40 CFR 403.7), the combined wastestream

formula (40 CFR 403.6(e)), and/or a Fundamentally Different Factors variance (40 CFR 403.13) at the time the user submits the report required by this paragraph, the information required by paragraphs (D) and (E) of this section shall pertain to the modified limits.

- (2) If the categorical Pretreatment Standard is modified by a removal allowance (40 CFR 403.7), the combined wastestream formula (40 CFR 403.6(e)), and/or a Fundamentally Different Factors variance (40 CFR 403.13) after the user submits the report required by paragraphs (D) and (E) of this section, then a new report shall be submitted by the user within 60 days after the modified limit is approved.
- H. Any other information as may be deemed necessary by the Director to evaluate the wastewater discharge permit application. Incomplete or inaccurate applications will not be processed and will be returned to the user for revision.

# 3.8 Signatory and Certification Requirement

All wastewater discharge permit applications and user reports must be signed by an authorized representative of the user and contain the following certification statement:

"I certify under penalty of law that this document and all attachments were prepared under my direction or supervision in accordance with a system designed to assure that qualified personnel properly gather and evaluate the information submitted. Based on my inquiry of the person or persons who manage the system, or those persons directly responsible for gathering the information, the information submitted is to the best of my knowledge and belief, true, accurate, and complete. I am aware that there are significant penalties for submitting false information, including the possibility of fine and imprisonment for knowing violations."

Depending upon the nature of a user's discharge permit, certification statements may require additional information (zero discharge, non-significant categorical, etc.). Refer to individual permits for the required certification statements.

# 3.9 Wastewater Discharge Permit Decisions

The Director will evaluate the data furnished by the user in the application and may require additional information. Within thirty (30) days of receipt of a complete wastewater discharge permit application and any additional information required, the Director will determine whether or not to issue a wastewater discharge permit based on full evaluation and acceptance of the data furnished. Upon a determination to issue a

wastewater discharge permit, the user shall fulfill the public notice requirements, and publish in the largest local daily newspaper, its application for, and the City's intent to issue, a wastewater discharge permit. The manner, format, content and length of time for the publication shall be as prescribed by the Director, but at no time shall be less stringent than the requirements in WAC 173-216-090. The cost of publication shall be paid for by the applicant. If there have been no adverse comments when the public comment period has elapsed, the application shall be considered complete and the permit will be drafted. Once drafted the permit will also be put out for public comment. The permit shall be issued within thirty [30] days of the end of the applicable public comment period if no comments are received, or completion of review and addressing of any public notice comments received. If addressing public comments results in significant changes to the draft permit, the public notice may need to be repeated with the modified permit. The Director may deny any application for a wastewater discharge permit.

# 3.10 Wastewater Discharge Permit Contents

Wastewater discharge permits shall include such conditions as are reasonably deemed necessary by the Director to prevent pass through or interference, protect the quality of the water body receiving the treatment plant's effluent, protect worker health and safety, facilitate sludge management and disposal, and protect against damage to the POTW.

- A. Wastewater discharge permits must contain the following conditions:
  - (1) A statement that indicates wastewater discharge permit duration, which in no event shall exceed five (5) years;
  - (2) A statement that the wastewater discharge permit is non-transferable without prior notification to and approval from the City, and provisions for furnishing the new owner or operator with a copy of the existing wastewater discharge permit;
  - (3) Applicable pretreatment standards: Effluent limits, including Best Management Practices, based on applicable general Pretreatment Standards, categorical Pretreatment Standards, local limits, and State and local law;
  - (4) Self-monitoring, and sampling as necessary, reporting, notification, submittal of technical reports, compliance schedules, and record-keeping requirements. These requirements shall include an identification of pollutants to be monitored, sampling location, sampling frequency, and sample type based on Federal, State, and local law;

- (5) Requirement for immediate notification to the City where self-monitoring results indicate noncompliance;
- (6) Requirement to report a by-pass or upset of a pretreatment facility;
- (7) Requirement to control and notify the City of slug loads and accidental discharges, if determined by the City to be necessary;
- (8) Requirement for the SIU who reports noncompliance to repeat the sampling and analysis as required in Section 4.8 and submit results to the City within 30 days after becoming aware of the violation;
- (9) A statement of applicable civil, criminal, and administrative penalties for violation of pretreatment standards and requirements, and any applicable compliance schedule.
- (10) Requirements for notification of the City of any new wastewater characteristics, change in volume, or constituents of the wastewater being discharged or change in production.
- B. Wastewater discharge permits may contain, but need not be limited to, the following conditions:
  - (1) Limits on the average and/or maximum rate of discharge, time of discharge, and/or requirements for flow regulation and equalization;
  - (2) Requirements for the installation of pretreatment technology, pollution control, or construction of appropriate containment devices, designed to reduce, eliminate, or prevent the introduction of pollutants into the treatment works;
  - (3) Requirements for the development and implementation of spill control plans or other special conditions including management practices necessary to adequately prevent accidental, unanticipated, or non-routine discharges;
  - (4) Development and implementation of waste minimization and pollution prevention plans to reduce the amount of pollutants discharged to the POTW;
  - (5) The unit charge or schedule of user charges and fees for the management of the wastewater discharged to the POTW;

- (6) Requirements for installation and maintenance of inspection and sampling facilities and equipment;
- (7) A statement that compliance with the wastewater discharge permit does not relieve the permittee of responsibility for compliance with all applicable Federal and State pretreatment standards, including those which become effective during the term of the wastewater discharge permit;
- (8) Any special agreements the Director chooses to continue or develop between the City and user;
- (9) Compliance schedules;
- (10) Other conditions as deemed appropriate by the Director to ensure compliance with this ordinance, and State and Federal laws, rules, and regulations.

# 3.11 Wastewater Discharge Permit Appeals

Any person, including the user, may petition the Director to reconsider the terms of a wastewater discharge permit within thirty (30) days of its issuance.

- A. Failure to submit a timely petition for review shall be deemed to be a waiver of the administrative appeal.
- B. In its petition, the appealing party must indicate the wastewater discharge permit provisions objected to, the reasons for this objection, and the alternative condition, if any, it seeks to place in the wastewater discharge permit.
- C. The effectiveness of the wastewater discharge permit shall not be stayed pending the appeal.
- D. If the Director fails to act within sixty (60) days of the receipt of an appeal, a request for reconsideration shall be deemed to be denied. Decisions not to reconsider a wastewater discharge permit, not to issue a wastewater discharge permit, or not to modify a wastewater discharge permit, shall be considered final administrative actions for purposes of judicial review.
- E. Aggrieved parties seeking judicial review of the final administrative wastewater discharge permit decision must do so by filing a petition for review with the Snohomish County Superior Court within thirty (30) days following the final administrative wastewater discharge permit decision.

#### 3.12 Wastewater Discharge Permit Duration

Wastewater discharge permits shall be issued for a specified time period, not to exceed five (5) years. A wastewater discharge permit may be issued for a period less than five (5) years, at the discretion of the Director. Each wastewater discharge permit will indicate a specific date upon which it will expire. The wastewater discharge permit will expire at 11:59 PM on the specified date. Wastewater discharge permits may be extended or renewed by the Director on the Director's own initiative or upon timely application by the permit holder.

## 3.13 Wastewater Discharge Permit Modification

The Director may modify the wastewater discharge permit for good cause including, but not limited to, the following:

- A. To incorporate any new or revised Federal, State, or local pretreatment standards or requirements;
- B. To address significant alterations or additions to the user's operation, processes, or wastewater volume or character since the time of wastewater discharge permit issuance;
- C. A change in the POTW that requires either a temporary or permanent reduction or elimination of the authorized discharge;
- D. Information indicating that the permitted discharge poses a threat to the City's POTW, City personnel, or the receiving waters;
- E. Violation of any terms or conditions of the wastewater discharge permit;
- F. Misrepresentations or failure to fully disclose all relevant facts in the wastewater discharge permit application or in any required reporting;
- G. Revision of or a grant of variance from categorical pretreatment standards pursuant to 40 CFR 403.13;
- H. To correct typographical or other errors in the wastewater discharge permit; or
- I. To reflect a transfer of the facility ownership and/or operation to a new owner/operator.
- J. Any Permit Modification may be appealed under Section 3.11 of this ordinance.

## 3.14 Wastewater Discharge Permit Transfer

Wastewater discharge permits may be reassigned or transferred to a new owner and/or operator only if the permittee gives at least ninety (90) days advance notice to the Director and the Director approves the wastewater discharge permit transfer. The notice to the Director must include a written certification by the new owner and/or operator which:

- A. States that the new owner and/or operator has no immediate intent to change the facility's operations and processes;
- B. Identifies the specific date on which the transfer is to occur;
- C. Acknowledges full responsibility for complying with the existing wastewater discharge permit; and
- D. States that a copy of the existing permit has been provided to the new owner or operator.

Provided that the above occurs and that there were no significant changes to the manufacturing operation or wastewater discharge, the new owner will be considered an existing user and be covered by the existing limits and requirements in the previous owner's permit. Failure to provide advance notice of a transfer renders the wastewater discharge permit voidable as of the date of facility transfer.

# 3.15 Wastewater Discharge Permit Revocation

Wastewater discharge permits may be revoked for the following reasons:

- A. Failure to notify the City of significant changes to the wastewater prior to the changed discharge;
- B. Failure to provide prior notification to the City of changed conditions;
- C. Misrepresentation or failure to fully disclose all relevant facts in the wastewater discharge permit application;
- D. Falsifying self-monitoring reports;
- E. Tampering with monitoring equipment;
- F. Refusing to allow the City timely access to the facility premises and records;

- G. Failure to meet discharge limitations;
- H. Failure to pay fines;
- Failure to pay sewer charges;
- J. Failure to meet compliance schedules;
- K. Failure to complete a wastewater survey or the wastewater discharge permit application;
- L. Failure to provide advance notice of the transfer of a permitted facility;
- M. If the City has to invoke its emergency provision as cited in Section 9.7 of the Ordinance;
- N. Violation of any pretreatment standard or requirement;
- O. Violation of any terms of the wastewater discharge permit;
- P. Violation of any provisions of this ordinance; or
- Q. Violation of any terms of an order of the Director issued under this ordinance.

Wastewater discharge permits shall be voidable upon cessation of operations or transfer of business ownership. All wastewater discharge permits issued to a particular user are void upon the issuance of a new wastewater discharge permit to that user.

## 3.16 Wastewater Discharge Permit Re-issuance

A user, required to have a wastewater discharge permit, shall apply for wastewater discharge permit re-issuance by submitting a complete wastewater discharge permit application or other data the Director deems adequate for renewal, in accordance with Section 3.7 of this ordinance, a minimum of ninety (90) days prior to the expiration of the user's existing wastewater discharge permit. A user, whose existing wastewater discharge permit has expired and has submitted its re-application in the time period specified herein, shall be deemed to have an effective wastewater discharge permit until the City issues or denies the new wastewater discharge permit. A user, whose existing wastewater discharge permit has expired and who failed to submit its reapplication in the time period specified herein, will be deemed to be discharging without a wastewater discharge permit.

## **SECTION 4 - REPORTING REQUIREMENTS**

## 4.1 Baseline Monitoring Reports

A. Within either one hundred and eighty (180) days after the effective date of a categorical pretreatment standard, or the final administrative decision on a category determination under 40 CFR 403.6(a)(4) (whichever is later) existing categorical users currently discharging to or scheduled to discharge to the POTW, shall be required to submit to the City a report which contains the information listed in Section 3.7 of this ordinance.

At least ninety (90) days prior to commencement of their discharge, new sources, and sources that become categorical users subsequent to the promulgation of an applicable categorical standard, shall be required to submit to the City a report which contains the information listed in Section 3.7 of this ordinance.

A new source may be required to submit an engineering report to the Director and comply with WAC 173-240; the report should explain the method of pretreatment a new source intends to use to meet applicable categorical standards. A new source shall give estimates of its anticipated flow and quantity of pollutants discharged.

# 4.2 Final Compliance Report (Initial Compliance Report)

- A. Within 90 days following the date for final compliance by the SIU with applicable pretreatment standards and requirements set forth in this ordinance, in a wastewater discharge permit, or within 30 days following commencement of the introduction of wastewater into the POTW by a new source or new users considered by the City to fit the definition of SIU, the affected user shall submit to the City a report containing the information outlined in Paragraph (D)-(F) of Section 3.7 of this ordinance.
- B. For users subject to equivalent mass or concentration limits established by the City in accordance with procedures established in 40 CFR 403.6 (c), this report shall contain a reasonable measure of the user's long term production rate. For all other users subject to categorical pretreatment standards expressed in terms of allowable pollutant discharge per unit of production (or other measure of operation), this report shall include the user's actual production during the appropriate sampling period.

## 4.3 Periodic Compliance Report

- A. Any user that is issued a permit under this ordinance and performs self-monitoring shall submit to the City during the months of June and December, unless required on other dates or more frequently by the City, a report indicating the nature of the effluent over the previous reporting period. The frequency of monitoring shall be as prescribed within the permit. At a minimum (except for zero discharge users, Middle Tier Categorical Industrial Users and Non-Significant Categorical Industrial Users) users shall sample their discharge at least twice per year. In cases where the Pretreatment Standard requires compliance with a Best Management Practice (BMP) or pollution prevention alternative, the user must submit documentation required by the Director or the Pretreatment Standard necessary to determine the compliance status of the user.
- B. The report shall include a record of the concentrations (and mass if specified in the wastewater discharge permit) of the pollutants listed in the wastewater discharge permit that were measured and a record of all flow measurements (average and maximum) taken at the designated sampling locations and shall also include any additional information required by this ordinance or the wastewater discharge permit. Production data shall be reported if required by the wastewater discharge permit. Both daily maximum and average concentration (or mass, where required) shall be reported when required by the control mechanism.
  - If a user sampled and analyzed more frequently than what was required by the City or by this ordinance, using methodologies in 40 CFR Part 136, it must submit all results of sampling and analysis of the discharge during the reporting period.
- C. Any user subject to equivalent mass or concentration limits established by the City or by unit production limits specified in the applicable categorical standards, shall report production data as outlined in Section 4.2 (B) of this ordinance.
- D. Zero discharge users shall submit periodic reports as required by the Director stating that no process waste has been discharged to the POTW.
- E Middle Tier Categorical Industrial Users and -Non-Significant Categorical Industrial Users shall submit reports as required in their control mechanisms.
- F. If the City calculated limits to factor out dilution flows or non-regulated flows, the user will be responsible for providing flows from the regulated process flows, dilution flows and non-regulated flows.
- G. Flows shall be reported on the basis of actual measurement; provided, however,

that the City may accept reports of average and maximum flows estimated by verifiable techniques if the City determines that an actual measurement is not feasible.

- H. Sampling shall be representative of the user's daily operations and shall be taken in accordance with the requirements specified in Section 5 of this ordinance.
- I. The City may require reporting by users that are not required to have an industrial wastewater discharge permit if information or data is needed to establish a sewer charge, determine the treatability of the effluent or determine any other factor that is related to the operation and maintenance of the sewer system.
- J. The City may require self-monitoring by the user or, if requested by the user, may agree to perform the periodic compliance monitoring needed to prepare the periodic compliance report required under this section. If the City agrees to perform such periodic compliance monitoring, it may charge the user for such monitoring, based upon the costs incurred by the City for the sampling and analyses. Any such charges shall be added to the normal sewer charge and shall be payable as part of the utility bills. The City is under no obligation to perform periodic compliance monitoring for a user.
- K. All periodic compliance reports must be signed and certified in accordance with section 3.8 and their applicable discharger classification.

# 4.4 Compliance Schedules for Meeting Applicable Pretreatment Standards

In any situation where a compliance schedule is issued by the City, the following conditions shall be met.

- A. The schedule shall contain increments of progress in the form of dates for the commencement and completion of major events leading to the construction and operation of additional pretreatment required for the user to meet the applicable pretreatment standards (e.g., hiring an engineer, completing preliminary plans, completing final plans, executing contract for major components, commencing construction, completing construction, etc.).
- B. No increment referred to in paragraph (A) of this section shall exceed nine (9) months.
- C. Not later than 14 days following each date in the schedule and the final date for compliance, the user shall submit a progress report to the City including, at a minimum, whether or not it complied with the increment of progress to be met

on such date and, if not, the date on which it expects to comply with this increment of progress, the reason for delay, and the steps being taken by the user to return the construction to the schedule established. In no event shall more than nine (9) months elapse between such progress reports.

# 4.5 Notification of Significant Production Changes

Any user operating under a wastewater discharge permit incorporating mass or concentration limits based on production levels shall notify the City within two (2) business days after the user has a reasonable basis to know that the production level will significantly change within the next calendar month. Any user not providing a notice of such anticipated change will be required to comply with the existing limits contained in its wastewater discharge permit.

# 4.6 Dangerous and Hazardous Waste Notification

Dangerous waste may potentially be discharged to the sewer under the Washington Department of Ecology Domestic Sewage Exclusion. Waste must not exhibit hazardous or dangerous waste characteristics and cannot exceed the dangerous waste concentration thresholds in WAC 173-303-090. All wastes discharged must be treatable by the WPCF. Any user that intends to discharge dangerous waste that meets the appropriate requirements (see WAC 173-303-070 through WAC 173-303-100) must contact the City to obtain an industrial waste discharge permit. Any dangerous waste discharged to the WPCF without a permit is prohibited and will be enforced as an unpermitted discharge. Pharmaceutical waste specifically cannot be sewered and cannot be approved under the domestic sewage exclusion.

The discharge of hazardous waste is strictly prohibited. If a user discharges hazardous waste then the user shall notify the City verbally immediately, within a minimum of 24 hours, by contacting Public Works during business hours or City of Everett Dispatch after hours and weekends. The user must follow up the initial notification with written notification to the City, the EPA RCRA Director and State Hazardous Waste and Toxics Reduction (HWTR) regional office, in writing within 90 days of the discharge of any substance, which, if otherwise disposed of, would be a hazardous waste under 40 CFR 261 or WAC 173-303. Such notification must include the name of the hazardous waste as set forth in 40 CFR 261, the EPA hazardous waste number, the quantity discharged, and the type of discharge (continuous, batch, or other).

Whenever the EPA or State of Washington publish new RCRA rules identifying additional dangerous or hazardous wastes, or new characteristics of existing dangerous and hazardous wastes, any affected user or Permittee must notify the City, EPA RCRA

Director and State HWTR regional office if any of these wastes are discharged to the City's treatment system. The notification must occur within 90 days of the effective date of the published regulation. To continue discharge, the wastes shall meet the requirements of Washington Department of Ecology Domestic Sewage Exclusion Guidance and be individually permitted by the City for suitability for discharge.

# 4.7 Notice of Potential Problems, Including Accidental Spills, Slug Loadings

Any user shall notify the City immediately of all discharges that could cause problems to the POTW, including any slug loadings, as defined in Section 1.3 of this ordinance. The notification shall include the concentration and volume and corrective action. Steps being taken to reduce any adverse impact should also be noted during the notification. Any user who discharges a slug (or slugs) of pollutants shall be liable for any expense, loss, or damage to the POTW, in addition to the amount of any fines imposed by the City or on the City under state or federal law.

# 4.8 Noncompliance Reporting

If monitoring or sampling performed by a user indicates a violation, the user shall notify the City within 24 hours of becoming aware of the violation. The user shall also repeat the sampling within five (5) days and submit the results of the repeat analysis to the City within 30 days after becoming aware of the violation, except the user is not required to resample if:

- A. The City performs sampling at the user at a frequency of at least once per month, or
- B. The City performs sampling at the user between the time when the user performs its initial sampling and the time when the user receives the results of this sampling, or
- C. The user's required sampling frequency is twice per month or more frequent. This includes parameters such as flow or pH that are continuously monitored.

# 4.9 Notification of Changed Discharge

All users shall promptly notify the City in advance of any substantial change in the volume or character of pollutants in their discharge, including significant manufacturing process changes, pretreatment modifications, or any other change that will substantially change their discharge. Formal written notification to the City shall be made 30 days

prior to such introduction and the user shall obtain approval from the City to do so.

# 4.10 TTO Reporting

Categorical users which are required by EPA to eliminate and/or reduce the levels of toxic organics (TTOs) discharged into the sewer system must follow the Categorical Pretreatment Standards for that industry. Those users must also meet the following requirements:

- A. Must sample, as part of the initial application requirements, for the organics listed under the TTO limit;
- B. May submit a statement that no TTOs are used at the facility and/or develop a solvent management plan in lieu of continued monitoring for TTO, if authorized by the Director:

If allowed to submit a statement or develop a solvent management plan, the user must routinely submit a certification statement as part of its self-monitoring report that there has been no dumping of concentrated toxic organic into the wastewater and that it is implementing a solvent management plan as approved by the City. The Director may require the development and implementation of a solvent management plan in addition to monitoring for TTO.

# 4.11 Reports from Unpermitted Users

Any user not required to obtain a wastewater discharge permit shall provide appropriate reports to the City as the Director may require.

## 4.12 Record Keeping

Users subject to the reporting requirements of this ordinance shall retain, and make available for inspection and copying, all records of information obtained pursuant to any monitoring activities required by this ordinance and any additional records of information obtained pursuant to monitoring activities undertaken by the user independent of such requirements. Records shall include the date, exact place, method, and time of sampling and the name of the person(s) taking the samples; the dates analyses were performed; who performed the analyses; the analytical techniques or methods used; and the results of such analyses. These records shall remain available for a period of at least three (3) years. This period shall be automatically extended for the duration of any litigation concerning the user or POTW, or where the user has been

specifically notified of a longer retention period by the Director.

# **4.13** *Timing*

Written reports will be deemed to have been submitted on the date postmarked. For reports which are not mailed, postage prepaid, into a mail facility serviced by the United States Postal Service, the date of receipt of the report shall govern.

# 4.14 Electronic Reporting

The City may implement electronic reporting for users. Until the electronic reporting system is approved under the Cross-Media Electronic Reporting Rule (CROMERR) and the individual user has complied with identity certification requirements, electronic reports must be followed up with original paper reports. Electronically submitted reports will be deemed to have been submitted on the date they are posted to the electronic reporting system. Electronic reporting does not preclude the City from requiring a user to continue submitting paper reports, if the City deems it is still necessary.

# **SECTION 5 - SAMPLING AND ANALYTICAL REQUIREMENTS**

## 5.1 Sampling Requirements for Users.

All samples shall be taken in a manner so they are representative of the wastestream being sampled.

- A. Where the discharge is continuous, a minimum of four (4) grab samples shall be taken for cyanide, total phenols, oil and grease, sulfide, and volatile organics. For all other pollutants, 24-hour composite samples shall be obtained through flowproportional composite sampling techniques where feasible. The City may waive flow-proportional composite sampling for any user that demonstrates that flowproportional is not feasible. In such cases, samples may be obtained through time-proportional composite sampling techniques or through a minimum of four (4) grab samples where the user demonstrates that this will provide a representative sample of the effluent being discharged. Multiple grab samples for cyanide, total phenols, and sulfides may be composited in the field, volatile organics and oil and grease should be composited in the laboratory. Composite samples for other parameters unaffected by the compositing procedures as documented in approved EPA methodologies may be authorized by the Director, as appropriate. In addition, grab samples may be required to show compliance with instantaneous limits.
- B. Samples shall be taken immediately downstream from pretreatment facilities if such exist or immediately downstream from the regulated or manufacturing process if no pretreatment exists or as determined by the City and/or contained in the user's wastewater discharge permit. For categorical users, if other wastewaters are mixed with the regulated wastewater prior to pretreatment the user shall measure the flows and concentrations necessary to allow use of the combined wastestream formula of 40 CFR 403.6(e) in order to evaluate compliance with the Applicable Categorical Pretreatment Standards. For other SIUs, for which the City has adjusted its local limits to factor out dilution flows, the user shall measure the flows and concentrations necessary to evaluate compliance with the adjusted pretreatment standard(s).
- C. In the case of batch discharges, the samples shall be obtained in a manner that ensures they are representative of the batch.
- D. All sample results shall indicate the time, date and place of sampling, and methods of analysis, and shall certify that such sampling and analysis is representative of normal work cycles and expected pollutant discharges from the user. If a user sampled and analyzed more frequently than what was

required in its wastewater discharge permit, using methodologies in 40 CFR Part 136, it must submit all results of sampling and analysis of the discharge as part of its self-monitoring report.

# 5.2 Analytical Requirements

All pollutant analyses, including sampling techniques, shall be performed in accordance with the techniques prescribed in 40 CFR Part 136, unless otherwise specified in an applicable categorical pretreatment standard. If 40 CFR Part 136 does not contain sampling or analytical techniques for the pollutant in question, sampling and analyses must be performed in accordance with procedures approved by the EPA.

All analyses performed to establish compliance and used in compliance reporting shall be performed by a laboratory accredited by the Washington State Department of Ecology, Quality Assurance Division in accordance with 173-50 WAC. Laboratories must be accredited for the analyses for which they are performing.

To ensure that the reported data is valid for determining compliance with requirements, all data shall have a detection level (DL) no greater than twenty-five percent (25%) of the regulatory limit included in this ordinance or applicable State or Federal regulation.

#### **SECTION 6 - COMPLIANCE MONITORING**

## 6.1 Inspection and Sampling

Continued connection and use of the Everett Municipal sewer system shall be contingent on the right of the City to inspect and sample all discharges into the system. The City shall have the right to enter the facilities of any user for the purpose of the enforcement of this ordinance and other provisions of the Everett Municipal Code, as amended, and to determine that any wastewater discharge permit or order issued hereunder, is being met and whether the user is complying with all requirements thereof. Users shall allow the Director ready access to all parts of the premises for the purposes of inspection, sampling, records examination and copying, and the performance of any additional duties.

- A. Where a user has security measures in force which require proper identification and clearance before entry into its premises, the user shall make necessary arrangements with its security guards so that, upon presentation of suitable identification, the Director will be permitted to enter without delay for the purposes of performing specific responsibilities.
- B. The Director shall have the right to set up on the user's property, or require installation of such devices as are necessary to conduct sampling and/or metering of the user's operations.
- C. Any temporary or permanent obstruction to safe and easy access to the facility to be inspected and/or sampled shall be promptly removed by the user at the written or verbal request of the Director and shall not be replaced. The costs of clearing such access shall be borne by the user.
- D. Unreasonable delays in allowing the Director access to the user's premises shall be a violation of this ordinance.

# **6.2** Monitoring Facilities

Each user may be required to provide and operate at its own expense a monitoring facility to allow inspection, sampling, and flow measurements of each sewer discharge to the City. Each monitoring facility shall be situated on the user's premises, except where such a location would be impractical or cause undue hardship on the user, the City may concur with the facility being constructed in the public street or sidewalk area, providing that the facility is located so that it will not be obstructed by landscaping or parked vehicles. The Director, whenever applicable, may require the construction and

maintenance of sampling facilities at other locations (for example, at the end of a manufacturing line, wastewater treatment system).

There shall be ample room in or near such facility to allow accurate sampling, flow measurement and preparation of samples for analysis. The facility, sampling, and measuring equipment shall be maintained at all times in a safe and proper operating condition at the expense of the user. All monitoring facilities shall be constructed and maintained in accordance with all applicable local construction standards and specifications. All monitoring facilities will remain unobstructed for ready access by City personnel.

The Director may require the user to install monitoring equipment as necessary. All devices used to measure wastewater flow and quality shall be maintained and calibrated in accordance with manufacturers' recommendations to ensure their accuracy.

#### 6.3 Search Warrants

If the Director has been refused access to a building, structure or property, or any part thereof, and is able to demonstrate probable cause to believe that there may be a violation of this ordinance, or that there is a need to inspect as part of a routine inspection program of the City designed to verify compliance with this ordinance or any wastewater discharge permit or order issued hereunder, or to protect the overall public health, safety and welfare of the community, then the Director shall seek issuance of a search and/or seizure warrant from the Everett Municipal Court, the Everett District Court, or the Snohomish County Superior Court. Such warrant shall be served at reasonable hours by the Director and may be accomplished in the in the company of a uniformed police officer of the City.

#### 6.4 Vandalism

No person shall willfully or negligently break, damage, destroy, uncover, deface, tamper with, or prevent access to any structure, appurtenance or equipment, or other part of the POTW along with associated conveyance system and sampling equipment. Any person found in violation of this requirement shall be subject to the sanctions set out in this ordinance.

#### **SECTION 7 - CONFIDENTIAL INFORMATION**

Information and data on a user obtained from reports, surveys, wastewater discharge permit applications, wastewater discharge permits, and monitoring programs, and from City inspection and sampling activities, shall be available to the public without restriction, unless the user specifically requests, and is able to demonstrate to the satisfaction of the City, that the release of such information would divulge information, processes or methods of production that are exempt from disclosure under the Public Records Act at Chapter 42.56 RCW, or as hereafter amended.

When requested and demonstrated by the user furnishing a report that such information should be held confidential, the City shall make reasonable efforts to protect the portions of a report which might disclose trade secrets or secret processes from inspection by the public; Such information, however, shall be made available immediately upon request to governmental agencies for uses related to the NPDES program or pretreatment program, in enforcement proceedings involving the person furnishing the report, or upon presentation of a duly issued subpoena or other court order.

Wastewater constituents, characteristics, and other effluent data as defined by 40 CFR 2.302, or the identity and address of the user will not be recognized as confidential information and will be available to the public without restriction.

# SECTION 8 - PUBLICATION OF USERS IN SIGNIFICANT NONCOMPLIANCE

The City shall publish annually, in the largest daily newspaper published in the municipality where the POTW is located, a list of the users that, during the previous twelve (12) months, were in significant noncompliance with applicable pretreatment standards and requirements. Significant noncompliance will be evaluated on a rolling quarter basis. The term significant noncompliance shall mean:

- A. Chronic violations of wastewater discharge limits, defined here as those in which sixty-six percent (66%) or more of wastewater measurements taken during a six (6) month period exceed the daily maximum limit or average limit for the same pollutant parameter by any amount;
- B. Technical Review Criteria (TRC) violations, defined here as those in which thirty-three percent (33%) or more of wastewater measurements taken for each pollutant parameter during a six (6) month period equals or exceeds the product of the daily maximum limit or the average limit multiplied by the applicable criteria (1.4 for BOD, TSS, fats, oils and grease, and 1.2 for all other pollutants except pH);
- C. Any other violation of a pretreatment standard or requirement that the City believes has caused, alone or in combination with other discharges, interference or pass through (including endangering the health of City personnel or the general public);
- D. Any discharge of pollutants that has caused imminent endangerment to the public or to the environment, or has resulted in the City's exercise of its emergency authority to halt or prevent such a discharge;
- E. Failure to meet, within ninety (90) days of the scheduled date, a compliance schedule milestone contained in a wastewater discharge permit or enforcement order for starting construction, completing construction, or attaining final compliance;
- F. Failure to provide within forty-five (45) days after the due date, any required reports, including baseline monitoring reports, reports on compliance with categorical pretreatment standard deadlines, periodic self-monitoring reports, and reports on compliance with compliance schedules;
- G. Failure to accurately report noncompliance; or
- H. Any other violation(s) that the City determines will adversely affect the

operation or implementation of the local pretreatment program.

#### **SECTION 9 - ADMINISTRATIVE ENFORCEMENT REMEDIES**

## 9.1 Notification of Violation (Notice of Violation, NOV)

When the Director finds that a user has violated (or continues to violate) any provision of this ordinance, a wastewater discharge permit or order issued hereunder, or any other pretreatment standard or requirement, the Director may serve upon that user a written Notice of Violation. The Director may select any means of service which is reasonable under the circumstances.

Within seven (7) calendar days of the receipt of this notice, an explanation of the violation and a plan for the satisfactory correction and prevention thereof, to include specific required actions, shall be submitted by the user to the Director. Submission of this plan in no way relieves the user of liability for any violations occurring before or after receipt of the Notice of Violation. Nothing in this section shall limit the authority of the City to take any action, including emergency actions or any other enforcement action, without first issuing a Notice of Violation.

#### 9.2 Consent Orders

The Director may enter into Consent Orders, assurances of voluntary compliance in "good faith," or other negotiated settlements between the City and the user establishing an agreement with any user who assumes responsibility for noncompliance. Such documents will include specific action to be taken by the user to correct the noncompliance within a time period specified by the document. Such documents shall have the same force and effect as the administrative orders issued pursuant to Sections 9.4 and 9.5 of this ordinance and shall be judicially enforceable. Use of a Consent Order shall not be a bar against, or prerequisite for, taking any other action against the user.

# 9.3 Show Cause Hearing

The Director may order a user which has violated or continues to violate, any provision of this ordinance, a wastewater discharge permit or order issued hereunder, or any other pretreatment standard or requirement, to appear before the Director and show cause why a proposed enforcement action should not be taken. Notice shall be served on the user specifying the time and place for the hearing, the proposed enforcement action, the reasons for such action, and a request that the user show cause why the proposed enforcement action should not be taken. The notice of the hearing shall be served personally or by registered or certified mail (return receipt requested) at least ten business (10) days prior to the hearing. Such notice may be served on any

authorized representative of the user. A show cause hearing shall not be a bar against, or prerequisite for, taking any other action against the user.

# 9.4 Compliance Orders

When the Director finds that a user has violated or continues to violate any provision of this ordinance, a wastewater discharge permit or order issued hereunder, or any other pretreatment standard or requirement, the Director may issue an order to the user responsible for the discharge directing that the user come into compliance within a time specified in the order. Compliance Orders may require users to refrain from certain activities, install additional pretreatment equipment, increase self-monitoring, use best management practices designed to minimize the amount of pollutants discharged to the sewer. A compliance order may be issued unilaterally, and the terms of the order need not be discussed with the user in advance. If the user does not come into compliance within the time specified in the order, sewer service may be discontinued. Issuance of a compliance order shall not be a bar against, or a prerequisite for, taking any other action against the user.

#### 9.5 Cease and Desist Orders

When the Director finds that a user has violated (or continues to violate) any provision of this ordinance, a wastewater discharge permit or order issued hereunder, or any other pretreatment standard or requirement, or that the user's past violations are likely to recur, the Director may issue an order to the user directing it to cease and desist all such violations and directing the user to:

- A. Immediately comply with all requirements; and
- B. Take such appropriate remedial or preventive action as may be needed to properly address a continuing or threatened violation, including halting operations and/or terminating the discharge.

Issuance of a cease and desist order shall not be a bar against, or a prerequisite for, taking any other action against the user.

#### 9.6 Administrative Fines

A. When the Director finds that a user has violated or continues to violate any provision of this ordinance, a wastewater discharge permit or order issued hereunder, or any other pretreatment standard or requirement, the Director

may fine such user in an amount not less than \$250 and not to exceed \$10,000. Such fines shall be assessed on a per violation, per day basis. In the case of monthly or other long-term average discharge limits, fines shall be assessed for each day during the period of violation. Said Administrative fines shall constitute a sewer service surcharge, and upon assessment, shall be subject to collection in the same manner as all other sewer utility rates, charges and penalties.

- B. Unless other arrangements have been made with, and authorized by the Director, unpaid charges, fines, and penalties shall accrue thereafter at a rate of one percent (1%) per month. After 90 days, if charges, fines, and penalties have not been paid, the City may revoke the user's discharge permit.
- C. Users desiring to appeal and dispute such fines must file a written request for the Director to reconsider the fine along with full payment of the fine amount within ten calendar (10) days of being notified of the fine. Upon receipt of a timely appeal, the Director shall set a date and time for an appeal hearing, but in no case shall the hearing be set more than sixty business (60) days from the receipt of the timely notice of appeal unless otherwise agreed upon in writing by the appellate and the Director. The appellant shall be notified in writing of the date, time, and place for the appeal hearing. The Director or the Director's designee shall serve as the Hearing Examiner. In the event the user's appeal is successful, any payments made shall be returned to the user. Affirmation or modification of an administrative fine by the Public Works Director shall relate back to the original date of assessment.
- D. The City shall recover the costs of preparing administrative enforcement actions, such as notices and orders, including the cost of additional inspections, sampling and analysis, and may add them to the fine. The City may also add to the fine the monetary value of any economic benefit the user gained through noncompliance to the fine.
- E. Issuance of an administrative fine shall not be a bar against, or a prerequisite for, taking any other action against the user.
- F. Users seeking judicial review of administrative fines must do so by filing a Petition for Review in the Snohomish County Superior Court within thirty calendar (30) days of the decision of the Director.

# 9.7 Emergency Suspensions

The Director may immediately suspend a user's discharge (after informal notice to the user) whenever such suspension is necessary to stop an actual or threatened discharge which reasonably appears to present or cause an imminent or substantial

endangerment to the health or welfare of persons, threatens to interfere with the operation of the POTW, or which presents or may present an endangerment to the environment.

A. Any user notified of a suspension of its discharge shall immediately stop or eliminate its contribution. In the event of a user's failure to immediately comply voluntarily with the suspension order, the Director shall take such steps as deemed necessary, including immediate severance of the sewer connection, to prevent or minimize damage to the POTW, its receiving stream, or endangerment to any individuals.

The Director may allow the user to recommence its discharge when the user has demonstrated to the satisfaction of the City that the period of endangerment has passed. If the Director does not allow the user to recommence its discharge within 15 days of the emergency suspension, the Director shall initiate termination proceedings pursuant to Section 9.8 of this ordinance.

B. A user that is responsible, in whole or in part, for any discharge presenting imminent endangerment shall submit a detailed written statement, describing the causes of the harmful contribution and the measures taken to prevent any future occurrence, to the Director prior to the date of any show cause or termination hearing under Sections 9.3 and 9.8 of this ordinance.

Nothing in this section shall be interpreted as requiring a hearing prior to any emergency suspension under this section.

# 9.8 Termination of Discharge (Non-Emergency)

In addition to the provisions in Section 3.15 of this ordinance, any user that violates the following conditions is subject to discharge termination:

- A. Violation of wastewater discharge permit conditions;
- B. Failure to accurately report the wastewater constituents and characteristics of its discharge;
- C. Failure to report significant changes in operations or wastewater volume, constituents and characteristics prior to discharge;
- D. Refusal of reasonable access to the user's premises for the purpose of inspection, monitoring or sampling; or
- E. Violation of the pretreatment standards in Section 2 of this ordinance.

Such user will be notified of the proposed termination of its discharge and be offered an opportunity for hearing pursuant to Section 9.9 of this ordinance to dispute the proposed termination action. Initiation of a termination proceeding by the City shall not be a bar to, or a prerequisite for, taking any other action against the user.

# 9.9 Appeal Procedures

#### A. Appeals.

- 1. Any user who has been subject to an enforcement action by the City and who seeks to dispute a Notice of Violation, order, fine, or other action of the Director may file an appeal. No other person may appeal an enforcement action.
- 2. The notice of appeal must be filed in writing and received by the Director, in writing, within ten (10) calendar days of the receipt of the disputed action or proposed action. If the notice of appeal is not received by the Director within ten (10) days, the right to an appeal is waived. The notice of appeal shall state with particularity the basis on which appellant is disputing the action taken or proposed to be taken, including:
  - (a) The name and the address of the appellant;
  - (b) The nature of the decision or action being appealed;
  - (c) Why appellant believes the decision or action is unwarranted; and
  - (d) What the appellant believes the appropriate decision or action should be.
- 3. Upon receipt of a timely notice of appeal, the Director shall set a date and time for an appeal hearing, but in no case shall the hearing be set more than sixty (60) days from the receipt of the timely notice of appeal unless otherwise agreed upon in writing by the appellant and Director. The appellant shall be notified in writing of the date, time, and place for the appeal hearing. The Director or the Director's designee shall serve as the hearing examiner and be the presiding officer at the hearing.

#### B. Appeal Hearing.

- 1. Content of Notice of hearing. The notice of hearing shall include:
  - (a) Names and mailing addresses of all parties to whom notice is being given, and if known, the names and addresses of their representatives;
  - (b) If the City intends to appear, the mailing address and telephone

- number of the office designated to represent the City in the proceeding;
- (c) The official file or other reference number and name of proceeding;
- (d) The name, official title, mailing address and telephone number of the presiding officer, if known;
- (e) A statement of the time, place and nature of the proceeding;
- (f) A statement of the legal authority and jurisdiction under which the hearing is to be held;
- (g) A reference to the particular section of the ordinance or regulations involved;
- (h) A short and plain statement of the matters asserted by the agency; and
- A statement that a party who fails to attend or participate in a hearing or other stage of an appeal hearing may be held in default.
- 2. Procedures at hearing. The hearing examiner/presiding officer, who may be the Director, or the Director's designee, shall regulate the course of the hearing. The presiding officer shall afford to all parties the opportunity to respond, present evidence and argument, conduct cross-examination, and submit rebuttal evidence; provided, however, the presiding officer may control the manner and extent of cross-examinations and rebuttal. In the discretion of the presiding officer, all or part of the hearing may be conducted by telephone or other electronic means as long as each party in the hearing has an opportunity to effectively participate and hear.
- 3. Rules of evidence. Evidence, including hearsay evidence, is admissible if in the judgment of the presiding officer it is the kind of evidence on which reasonably prudent persons are accustomed to rely in the conduct of their affairs. The presiding officer shall exclude evidence that is excludable on constitutional or statutory grounds or on the basis of evidentiary privilege recognized in the courts of this state. The presiding officer may exclude evidence that is irrelevant, immaterial, or unduly repetitious. All testimony of parties and witnesses shall be made under oath or affirmation. Documentary evidence may be received in the form of copies or excerpts or by incorporation by reference. Official notice may be taken of (a) any judicially cognizable facts, (b) technical or scientific facts within the City's specialized knowledge, and (c) codes or standards that have been adopted by an agency of the United States, this state or another state, or by a nationally recognized organization or association. A party proposing that official notice be taken may be required to produce a copy of the material to be noticed.

- 4. Default. If a party fails to attend or participate in any stage of a hearing, the presiding officer may serve upon all parties a default or other dispositive order, which shall include a statement of grounds for the order. Within seven (7) calendar days after service of a default order, the party against whom it was entered may file a written motion requesting that the order be vacated, stating the grounds for the motion. If the party against whom the default order is entered fails to timely file a motion to vacate or the motion to vacate is not granted, the default order will be the final decision of the City.
- 5. Burden of proof. The appellant shall have the burden of proof by a preponderance of the evidence.

#### C. Appeal Conclusion.

After the conclusion of the hearing, the hearing examiner shall determine if the disputed action was proper, and shall approve, modify, or rescind the disputed action. The final determination of the hearing examiner shall be in writing, and all parties shall be provided a copy of the final determination within ten (10) days (or such longer time as the hearing examiner may determine in writing). This decision will include findings of fact that are supported by and based on the record. These findings will be entitled to deference on any judicial review.

#### D. Judicial Review of Appeal.

- 1. Any party, including the City, the Washington State Department of Ecology, the United States Environmental Protection Agency, or the user/appellant, is entitled to review of the final determination of the hearing examiner in the Snohomish County Superior Court. Provided, that any petition for review shall be filed no later than thirty (30) calendar days after date of the final determination.
- 2. Copies of the petition for review shall be served as in all civil actions.
- 3. The filing of the petition shall not stay enforcement of the final determination except by order of the superior court and on posting of a bond to be determined by the court naming the City as beneficiary.
- 4. The review shall be conducted by the court without a jury. The record shall be satisfied by a narrative report certified by the hearing examiner and no verbatim record of proceedings before the hearing examiner shall be required to be presented to the superior court.
- 5. The court may affirm the final determination or remand the matter for

further proceedings before the hearing examiner; or the court may reverse the final determination if the substantial rights of the petitioners may have been prejudiced because the final determination was:

- (a) In violation of constitutional provisions; or
- (b) In excess of the authority or jurisdiction of the hearing examiner;

or

(c) Arbitrary and capricious.

#### **SECTION 10 - JUDICIAL ENFORCEMENT REMEDIES**

#### 10.1 Injunctive Relief

When the Director finds that a user has violated or continues to violate any provision of this ordinance, a wastewater discharge permit, or order issued hereunder, or any other pretreatment standard or requirement, the Director may petition the Snohomish County Superior Court through the City's Attorney for the issuance of a temporary or permanent injunction, as appropriate, which restrains or compels the specific performance of the wastewater discharge permit, order, or other requirement imposed by this ordinance on activities of the user.

The City may also seek such other action as is appropriate for legal and/or equitable relief, including a requirement for the user to conduct environmental remediation. A petition for injunctive relief shall not be a bar against, or a prerequisite for, taking any other action against a user. Injunctive relief shall be nonexclusive to other remedies available to the City.

#### 10.2 Civil Penalties

- A. A user which has violated or continues to violate any provision of this ordinance, a wastewater discharge permit, or order issued hereunder, or any other pretreatment standard or requirement shall be liable to the City for a maximum civil penalty of \$10,000 per violation, per day. In the case of a monthly or other long-term average discharge limit, penalties shall accrue for each day during the period of the violation.
- B. The Director may recover reasonable attorneys' fees, court costs, and other expenses associated with enforcement activities, including sampling and monitoring expenses, and the cost of any actual damages incurred by the City. The City may also add the monetary value of any economic benefit the user gained through noncompliance to the penalty.
- C. In determining the amount of civil liability, the Court shall take into account all relevant circumstances, including, but not limited to, the extent of harm caused by the violation, the magnitude and duration, any economic benefit gained through the user's violation, corrective actions by the user, the compliance history of the user, and any other factor as justice requires.
- D. Filing a suit for civil penalties shall not be a bar against, or a prerequisite for taking any other action against a user.

#### 10.3 Criminal Prosecution

- A. A user who violates any provision of this ordinance, a wastewater discharge permit, or order issued hereunder, or any other pretreatment standard or requirement shall, upon conviction, be guilty of a gross misdemeanor, punishable by a fine of not more than \$10,000 and/or one year in jail. Each day a violation occurs shall constitute a separate offense.
- B. A user who introduces any substance into the POTW which causes personal injury or property damage shall, upon conviction, be guilty of a gross misdemeanor and be subject to a penalty of not more than \$10,000 and/or one year in jail. Each day a violation occurs shall constitute a separate offense. This penalty shall be in addition to any other cause of action for personal injury or property damage available under State law.
- C. A user who knowingly makes any false statements, representations, or certifications in any application, record, report, plan, or other documentation filed, or required to be maintained, pursuant to this ordinance, wastewater discharge permit, or order issued hereunder, or who falsifies, tampers with, or knowingly renders inaccurate any monitoring device or method required under this ordinance shall, upon conviction, be guilty of a gross misdemeanor, and punished by a fine of not more than \$10,000 and/or one year in jail. Each day a violation occurs shall constitute a separate offense.

In addition, the user shall be subject to:

- (1) the provisions of 18 USC Section 1001 relating to fraud and false statements;
- (2) the provisions of Sections 309 (c) (4) of the Clean Water Act, as amended governing false statements, representation, or certification; and
- (3) the provision of Section 309 (c) (6) of the Clean Water Act, regarding responsible corporate officers.

#### 10.4 Remedies Non-Exclusive

The provisions in Sections 8 through 11 of this ordinance are not exclusive remedies. The City reserves the right to take any, all, or any combination of these actions against a non-compliant user. Enforcement of pretreatment violations will generally be in accordance with the City's enforcement response plan. However, the City reserves the right to take other action against any user when the circumstances warrant. Further, the

City is empowered to take more than one enforcement action against any non-compliant user. These actions may be taken concurrently.

#### **SECTION 11 - SUPPLEMENTAL ENFORCEMENT ACTION**

# 11.1 Performance Bonds

The Director may decline to issue or reissue a wastewater discharge permit to any user which has failed to comply with any provision of this ordinance, a previous wastewater discharge permit or order issued hereunder, or any other pretreatment standard or requirement unless such user first files a satisfactory bond, payable to the City, in a sum not to exceed a value determined by the Director to be necessary to achieve consistent compliance.

#### 11.2 Financial Assurances

The Director may decline to issue or reissue a wastewater discharge permit to any user which has failed to comply with any provision of this ordinance, a previous wastewater discharge permit or order issued hereunder, or any other pretreatment standard or requirement, unless the user first submits proof that it has obtained financial assurances sufficient to meet pretreatment requirements, and/or restore or repair damage to the POTW caused by its discharge.

#### 11.3 Service Severance

Whenever a user has violated or continues to violate any provision of this ordinance, a wastewater discharge permit or order issued hereunder, or any other pretreatment standard or requirement, water and/or sewer service to the user may be severed. Service will only recommence, at the user's expense, after it has satisfactorily demonstrated its ability to comply.

#### 11.4 Public Nuisances

A violation of any provision of this ordinance, wastewater discharge permit, or order issued hereunder, or any other pretreatment standard or requirement, is hereby declared a public nuisance and shall be corrected or abated as directed by the Director.

## 11.5 Contractor Listing

Users which have not achieved compliance with applicable pretreatment standards and

requirements are not eligible to receive a contractual award for the sale of goods or services to the City. Existing contracts for the sale of goods or services to the City held by a user found to be in significant noncompliance with pretreatment standards or requirements may be terminated at the discretion of the City.

# 11.6 Publication of Violations and/or Enforcement Actions.

The Director may publish violations and/or enforcement actions at any time, where monetary fines may be inappropriate in gaining compliance, or in addition to monetary fines. Violations and/or enforcement actions may also be published when the Director feels that public notice should be made, or at other appropriate times. The cost of such publications will be recovered from the user.

# 11.7 Penalties for Late Reports

A penalty of may be assessed to any user for each day that a report required by this ordinance, a permit or order issued hereunder is late, beginning one day after the date the report is due and shall escalate on 30 day intervals. Actions taken by the Director to collect late reporting penalties shall not limit the Director's authority to initiate other enforcement actions that may include penalties for late reporting violations.

# 11.8 Payment of Outstanding Fees and Penalties

The Director may decline to issue or reissue an individual wastewater discharge permit to any user who has failed to pay any outstanding fees, fines or penalties incurred as a result of any provision of this ordinance, a previous individual wastewater discharge permit, or any order issued hereunder.

#### **SECTION 12 - AFFIRMATIVE DEFENSES TO DISCHARGE VIOLATIONS**

## **12.1** Upset

- A. For the purposes of this section, upset means an exceptional incident in which there is unintentional and temporary noncompliance with applicable pretreatment standards because of factors beyond the reasonable control of the user. An upset does not include noncompliance to the extent caused by operational error, improperly designed treatment facilities, inadequate treatment facilities, lack of preventive maintenance, or careless or improper operation.
- B. An upset shall constitute an affirmative defense to an action brought for noncompliance with applicable pretreatment standards if the requirements of paragraph (C) are met.
- C. A user who wishes to establish the affirmative defense of upset shall demonstrate, through properly signed or initialed contemporaneous operating logs, or other relevant evidence that:
  - (1) An upset occurred and the user can identify the cause(s) of the upset;
  - (2) The facility was at the time being operated in a prudent and workmanlike manner and in compliance with applicable operation and maintenance procedures; and
  - (3) The user has submitted the following information to the Director within twenty four (24) hours of becoming aware of the upset [if this information is provided orally, a written submission must be provided within five (5) days]:
    - (a) A description of the indirect discharge and cause of noncompliance;
    - (b) The period of noncompliance, including exact dates and times or, if not corrected, the anticipated time the noncompliance is expected to continue; and
    - (c) Steps being taken and/or planned to reduce, eliminate, and prevent recurrence of the noncompliance.
- D. In any enforcement proceeding, the user seeking to establish the occurrence of

an upset shall have the burden of proof.

- E. A User may appeal any enforcement action due to an upset as provided under Section 9.9 of this ordinance.
- F. Users shall control production of all discharges to the extent necessary to maintain compliance with applicable pretreatment standards upon reduction, loss, or failure of its treatment facility until the facility is restored or an alternative method of treatment is provided. This requirement applies in the situation where, among other things, the primary source of power of the treatment facility is reduced, lost, or fails.

### 12.2 Prohibited Discharge Standards

A user shall have an affirmative defense to an enforcement action brought against it for noncompliance with the prohibitions in Section 2.1 A and B (3) through (7) if it can prove that it did not know, or have reason to know, that its discharge, alone or in conjunction with discharges from other sources, would cause pass through or interference and that either: (a) a local limit exists for each pollutant discharged and the user was in compliance with each limit directly prior to, and during, the pass through or interference; or (b) no local limit exists, but the discharge did not change substantially in nature or constituents from the user's prior discharge when the City was regularly in compliance with its NPDES permit, and in the case of interference, was in compliance with applicable sludge use or disposal requirements.

### 12.3 Bypass

- A. For the purposes of this section,
  - (1) Bypass means the intentional diversion of wastestreams from any portion of a user's treatment facility.
  - (2) Severe property damage means substantial physical damage to property, damage to the treatment facilities which causes them to become inoperable, or substantial and permanent loss of natural resources which can reasonably be expected to occur in the absence of a bypass. Severe property damage does not mean economic loss caused by delays in production.
- B. A user may allow any bypass to occur which does not cause applicable pretreatment standards or requirements to be violated, but only if it also is for

essential maintenance to assure efficient operation. These bypasses are not subject to the provision of paragraphs (C) and (D) of this section.

- C. (1) If a user knows in advance of the need for a bypass, it shall submit prior notice to the Director, at least ten (10) days before the date of the bypass, if possible.
  - (2) A user shall submit oral notice to the City of an unanticipated bypass that exceeds applicable pretreatment standards within twenty-four (24) hours from the time it becomes aware of the bypass. A written submission shall also be provided within five (5) days of the time the user becomes aware of the bypass. The written submission shall contain a description of the bypass and its cause; the duration of the bypass, including exact dates and times, and, if the bypass has not been corrected, the anticipated time it is expected to continue; and steps taken or planned to reduce, eliminate, and prevent reoccurrence of the bypass. The Director may waive the written report on a case-by-case basis if the oral report has been received within twenty-four (24) hours.
- D. (1) Bypass is prohibited, and the Director may take an enforcement action against a user for a bypass, unless
  - (a) Bypass was unavoidable to prevent loss of life, personal injury, or severe property damage;
  - (b) There were no feasible alternatives to the bypass, such as the use of auxiliary treatment facilities, retention of untreated wastes, or maintenance during normal periods of equipment downtime. This condition is not satisfied if adequate back-up equipment should have been installed in the exercise of reasonable engineering judgment to prevent a bypass which occurred during normal periods of equipment downtime or preventive maintenance; and
  - (c) The user submitted notices as required under paragraph (C) of this section.
  - (2) The Director may approve an anticipated bypass, after considering its adverse effects, if the Director determines that it will meet the three conditions listed in paragraph (D)(1) of this section.

#### **SECTION 13 - MISCELLANEOUS PROVISIONS**

#### 13.1 Pretreatment Charges and Fees

The City of Everett may adopt reasonable fees for reimbursement of costs of setting up and operating the City's Pretreatment Program. These fees will be set by ordinance.

These fees relate solely to the matters covered by this ordinance and are separate from all other rates or charges for sewer service, provided that the City shall collect said charges in the same manner as other sewer utility rates are collected, including but not limited to the sewer lien procedures provided under 35.67 RCW.

Fees may include, but are not limited to:

- A. Fees for wastewater discharge permits, including the cost of processing the permit applications, public noticing, issuing and administering the permit, and reviewing monitoring reports submitted by users;
- B. Fees for modifying or transferring permits;
- C. Fees for monitoring, inspection, surveillance and enforcement procedures including the cost of collection and analyzing a user's discharge;
- D. Fees for reviewing and responding to accidental discharge procedures and construction;
- E. Fees for preparing and executing enforcement action;
- F. Fees for filing appeals;
- G. Fees for High Strength Waste and Industrial Process flow; and
- H. Other fees as the City may deem necessary to carry out the requirements contained herein.
- I. Any user establishing a pattern of noncompliance, or having a history of noncompliance, or suspected of being in noncompliance, may require additional monitoring visits as deemed appropriate by the Director. Any additional inspections, sampling, surveillance monitoring activities, and analysis performed which detect noncompliance will be billed directly to the user.
- J. Enforcement Actions: All expenses in preparing enforcement actions may be

billed directly to the User.

- K. High Strength Waste Fees: Users having effluent concentrations of BOD and/or TSS in excess of 300 mg/L, and/or FOG concentrations in excess of 50 mg/L may be billed a High Strength Waste Surcharge. Surcharge rates will be established by the Director and based on cost of conveyance and treatment in the POTW.
- L. Industrial Flow Surcharge: An Industrial flow surcharge will be billed to all applicable Users based upon the amount of industrial waste flow.
- M. Cost Recovery: The City may recover any incurred expenses incurred due to noncompliance by a user, including, but not limited to; costs of opening a plugged sewer, costs of repairing a damaged sewer, costs of determining the cause of a SSO or CSO, and costs of cleaning up and mitigating a SSO or CSO.

All fees or charges will be collected by direct billing. Unless the Director has been made aware of extenuating circumstances that would prevent prompt payment, all fees are payable within 30 days of the billing. Fees past due will be considered a violation of this ordinance. Users not paying fees within 60 days of the billing period will be subject to permit suspension or termination of service.

### 13.2 Non-Liability

It is expressly the purpose of this ordinance to comply with the September 13, 1985 order of the Washington State Department of Ecology requiring the City of Everett to establish an industrial pre-treatment program and to provide for and promote the health, safety and welfare of the general public. It is not the intent of this ordinance to create or otherwise establish or designate any particular class or group of persons who will or should be especially protected or benefited by the terms or requirements of this ordinance.

It is the specific intent of this ordinance to place the obligation of complying with these regulations upon the applicant or Discharger and no provision nor any term used in this ordinance is intended to impose any duty whatsoever upon the City or any of its officers, employees or agents, except as provided under the Act or other related statutes of the United States or the State of Washington.

Nothing contained in this ordinance is intended to be nor shall be construed to create or form the basis for any tort liability on the part of the City or its officers, employees or agents, for any injury or damage resulting from the failure of an applicant or Discharger to comply with the provisions of this ordinance, or by reason or in consequence of any inspection, notice, order, certificate, permission or approval authorized or issued or done in connection with the implementation or enforcement of this ordinance, or

inaction on the part of the City related in any manner to the implementation or the enforcement of this ordinance by its officers, employees or agents.



### **EVERETT** City Council Agenda Item Cover Sheet

An ORDINANCE relating to Amendments to the City's Procurement Policy, Amending **Project title:** Ordinance 3781-20

Council Bill #	
CB 2404-01	
Agenda dates requ	uested:
Briefing	05/01/2024
Proposed action	05/08/2024
Consent	05 /45 /2024
Action Ordinance	05/15/2024 x
Public hearing	``
_	( No
Budget amendme	nt:
Yes	No
PowerPoint preser	ntation:
X Yes	No
Attachments:	
Ordinance	
Department(s) inv Finance	olved:
rillatice	
Contact person:	
Susy Haugen	
Phone number: (425) 257-8612	
Email:	
shaugen@everetty	va.gov
Initialed by:	
sh	
Department head	
Administration	
Council President	
Sourier resident	

Project:	Procurement Policy Amendments
Partner/Supplier:	N/A
Location:	N/A
Preceding action:	Ordinance <u>3781-20</u>
Fund:	N/A

Fiscal summary statement: None

Project summary statement: In 2020, the City Council adopted the City's Procurement Policy via Ordinance 3781-20. That Policy allows for revisions, additions and deletions to the Policy by the Mayor. Since the 2020 adoption, with recommendation by the Finance Department, the Mayor has approved certain administrative amendments, mostly updates required by law, clarifications, conforming processes to existing practices, and other technical matters. The Procurement Policy updated with those amendments is publicly available on the Procurement Division's website at <a href="https://www.everettwa.gov/319/Procurement">https://www.everettwa.gov/319/Procurement</a>.

This proposed ordinance would adopt two substantive, policy-related amendments to the Procurement Policy concerning relatively small procurements, for three key reasons:

- These procurements represent the bulk of procurements in number, and so processes associated with them can cumulatively cause significant staff burden.
- In addition, dollar thresholds related to small procurements can become outdated quickly in today's inflationary environment.
- Small procurements represent opportunities for small business, which may be helped by streamlining processes.

The two proposed amendments in the ordinance are:

Ordinance Section 1 (Small Works Roster). The City, like most Washington public agencies, uses a small works roster for projects less than \$350,000. Effective July 1, 2024, the Legislature revised the small works statute to now include an optional process to allow streamlined direct negotiation with small businesses and others. The proposed amendment would change the Procurement Policy to allow the City to use this new process.

Ordinance Section 2 (Departmental Solicitation). The Procurement Policy states that for nonconstruction procurements, the Procurement Division may informally solicit quotes. This amendment changes the Procurement Policy to allow this solicitation to be done also by City departments for procurements between \$10,000 and \$100,000, instead of this administrative work being solely shouldered by the Procurement Division.

#### Recommendation (exact action requested of Council):

Adopt Ordinance relating to Amendments to City's Procurement Policy, Amending Ordinance 3781-20.



ORDINANCE NO.	1		

An ORDINANCE relating to Amendments to the City's Procurement Policy, Amending Ordinance 3781- 20

#### WHEREAS,

- A. In 2020, the City Council adopted Ordinance 3781-20, which adopted the City's Procurement Policy. The Procurement Policy states in its Section 15 that the Mayor may make amendments to Procurement Policy, and the Mayor has made certain administrative amendments since 2020. The Procurement Policy, as updated by those amendments, is available for reference on the Procurement Division's public website.
- B. Other amendments to the Procurement Policy regarding small procurements are in the public interest and adopted by City Council with the ordinance, all as set forth below.

#### NOW, THEREFORE, THE CITY OF EVERETT DOES ORDAIN:

**Section 1.** A new section is added to Section 8 of the Procurement Policy as follows:

Effective July 1, 2024, the City may use the small works roster processes contained in RCW 39.04.152, including the processes in RCW 39.04.152(b) for small public works projects with an estimated cost less than \$150,000 for the purpose of increasing the utilization of small businesses.

On recommendation from the Finance Department, the Mayor will add subsections to this new section in the Procurement Policy as necessary for implementation. These subsections may include processes, procedures, and references to new standard small works forms. RCW 39.04.152 will not be implemented before the City establishes a "small, minority, women, and veteran-owned business utilization plan" as required by RCW 39.04.152(b).

#### **Section 2.** A new section is added to Section 7 of the Procurement Policy:

When a City department requires a procurement with a cost estimate of \$10,000 to \$100,000 (including shipping but not including sales tax), the department may use this section for solicitation of quotations from suppliers instead of the Procurement Division administering the solicitation. The department must obtain at least three competitive quotations. The departments must demonstrate due diligence to ensure eligible small business suppliers are invited to provide quotations. To the extent practical, departments using this section must equitably distribute opportunities among qualified suppliers.

On recommendation from the Finance Department, the Mayor will add subsections to this new section

in the Procurement Policy as necessary for implementation. These subsections may include processes, procedures, and references to new standard quotation forms.

<u>Section 3</u>. The City Clerk and the codifiers of this Ordinance are authorized to make necessary corrections to this Ordinance, including, but not limited to, the correction of scrivener's/clerical errors, references, ordinance numbering, section/subsection numbers, and any internal references.

<u>Section 4</u>. The City Council hereby declares that should any section, paragraph, sentence, clause or phrase of this ordinance be declared invalid for any reason, it is the intent of the City Council that it would have passed all portions of this ordinance independent of the elimination of any such portion as may be declared invalid.

<u>Section 5</u>. The enactment of this Ordinance shall not affect any case, proceeding, appeal, or other matter currently pending in any court or in any way modify any right or liability, civil or criminal, which may be in existence on the effective date of this Ordinance.

<u>Section 6</u>. It is expressly the purpose of this Ordinance to provide for and promote the health, safety, and welfare of the general public and not to create or otherwise establish or designate any particular class or group of persons who will or should be especially protected or benefited by the terms of this Ordinance. It is the specific intent of this Ordinance that no provision or any term used in this Ordinance is intended to impose any duty whatsoever upon the City or any of its officers or employees. Nothing contained in this Ordinance is intended nor shall be construed to create or form the basis of any liability on the part of the City, or its officers, employees, or agents, for any injury or damage resulting from any action or inaction on the part of the City related in any manner to the enforcement of this Ordinance by its officers, employees or agents.

Cassie Franklin, Mayor
ATTEST:
Marista Jorve, City Clerk
PASSED:
VALID:
PUBLISHED:
EFFECTIVE DATE:

## **EVERETT** City Council Agenda Item Cover Sheet

**Project title:** 

An Ordinance relating to Ambulance and Emergency First Aid Services, repealing, amending, and recodifying certain sections of Chapter 5.72 of the Everett Municipal Code.

Council Bill # interoffice use	Project: Amending Ordinance
CB 2404-02	Partner/Supplier: NA
Agenda dates requested:	Location: N/A
1 <sup>st</sup> Reading 5/08/24	Preceding action:
Consent	Fund:
Action 5/15/24 2 <sup>nd</sup> / 3 <sup>rd</sup> & Final Readings	
Ordinance X	Fiscal summary statement:
Public hearing Yes X No	There is no fiscal impact.
Budget amendment: Yes X No	Project summary statement:
PowerPoint presentation: Yes X No	Chapter 5.72 EMC contains ambulance licensing provisions that have never been used. This ordinance will repeal those sections.
Attachments: Ordinance  Department(s) involved:	Chapter 5.72 EMC also contains sections about Emergency Medical Services (EMS). Those sections do not belong in Title 5, which concerns business licenses. This ordinance will move those sections without amendment to a new EMC chapter 3.82,
Finance/Legal	which will be entitled "Emergency Medical Services."
Contact person: Susy Haugen	This ordinance also updates transport fees in EMC 5.72.210 (to be recodified as EMC 3.82.090).
Phone number:	Recommendation (exact action requested of Council):
X8612	Adopt Ordinance relating to Ambulance and Emergency First Aid Services, repealing, amending, and recodifying certain sections of Chapter 5.72 of the Everett Municipal
Email:	Code.
shaugen@everettwa.gov	
Initialed by:	
sh	
Department head	
Administration	
Council President	



ORDINANCE	NO.		
-----------	-----	--	--

AN ORDINANCE relating to Ambulance and Emergency First Aid Services, repealing, amending, and recodifying certain sections of Chapter 5.72 of the Everett Municipal Code.

#### WHEREAS,

- **A.** Chapter 5.72 EMC contains outdated licensing requirements for ambulances and first aid services, which have not been used by many years. One purpose of this ordinance is to repeal those.
- **B.** Chapter 5.72 EMC also contains other provisions related to the fire department and EMS services that do not belong in Title 5. The second purpose of this ordinance is to move those sections to an appropriate title of the Everett Municipal Code.

#### NOW, THEREFORE, THE CITY OF EVERETT DOES ORDAIN:

**Section 1.** The following sections of chapter 5.72 EMC are repealed:

```
5.72.020 (Title-Gender);
```

5.72.030 (Licenses required for ambulance service and certain emergency aid services);

5.72.040 (License not transferable);

5.72.050 (Required ambulance equipment—Response time);

5.72.060 (Standards for ambulance license—Liability insurance);

5.72.080 (Standards for ambulance company, ambulance attendant/emergency medical technician license);

5.72.090 (Expiration and renewal of license).

5.72.100 (Denial, revocation or suspension of license).

5.72.110 (Reports)

5.72.130 (Emergency medical system);

5.72.170 (Penalties);

5.72.180 (General Duty).

**Section 2.** EMC 5.72.210 (which also shall be recodified as EMC 3.82.090) is amended as follows:

#### 5.72.210 Operation and funding.

The following procedures are established for users of the city's fire department ALS and BLS

patient transport service:

- A. Charges to Be Made. All persons who are transported by fire department emergency paramedic/aid unit shall be charged for all services rendered by the fire department. Each person transported will be billed for all services provided. EMS members and nonmembers, as defined in subsection B of this section, shall be billed at the same rate.
- B. *EMS Membership Benefits*. By reason of property tax levies under RCW <u>84.52.069</u> for EMS, which are levied against property within the corporate limits of the city or are levied against property within the corporate limits of a jurisdiction for which the city has assumed contractual EMS responsibility, each resident of the city and those of contractual jurisdictions, by signing (by recipient or authorized representative) a city-approved EMS membership form containing an assignment of benefits to the city, together with an appropriate release of medical information, shall become an EMS member and be entitled to membership benefits as herein provided. An EMS member receiving transport by the fire department unit shall be deemed to have paid (by reason of the special levy imposed under RCW <u>84.52.069</u>) that portion of the charges incurred which is not payable by third parties and insurers, including but not limited to any insurance or medical benefits of any nature available to such member. This EMS membership benefit of coverage of charges in excess of available insurance or medical benefits shall cease when or if:
- 1. A member ceases to be a resident;
- 2. A member refuses to provide requested information pertaining to third party coverage or to provide appropriate releases of information and assignment of benefit to the city on forms provided by the city; or
- 3. Such EMS membership benefit is limited or extinguished by amendment or repeal of this chapter.
- C. Nonmembers. Persons receiving transport by fire department units who are not entitled to an EMS membership shall be required to pay all charges incurred. Where practical, the city, in accordance with procedures to be approved by the finance department, will, with the authorization of a nonmember receiving fire department transport services, first seek payment of charges incurred from such nonmember's insurance or other medical benefit provided. Such nonmember shall remain fully responsible for any amount due which is not paid by such third parties.
- D. *Medicare and Medicaid*. Eligible recipients of Medicare and Medicaid benefits shall be charged as the result of transport services by the fire department at only the maximum rate allowed under the Medicare and Medicaid federal programs, and the city shall accept as payment under the Medicare and Medicaid programs only such maximum amount as the city may collect pursuant to the applicable requirements and guidelines of the Medicare and Medicaid programs.
- E. Additions, Purchase, Compliance with Capital Facilities Plan. Additions and purchases to and

for the fire department EMS and patient transport services shall be made in accordance with the city's budget as the same now exists or as it may hereafter be amended.

- F. Additions, Purchases, Financing. The cost of making additions, or purchases to and for the fire department EMS and patient transport services, shall be paid from such sources and by such means as the city from time to time may direct, in accordance with state law and applicable regulations of the State Auditor.
- G. Emergency Medical Services Fund. There shall be created and established in the city's budget a special fund, to be known and designated as the emergency medical services fund. There shall be deposited in the emergency medical services fund: (1) all revenues collected pursuant to the fire department transport service fees established and set forth in subsection N of this section; (2) all revenues collected pursuant to any contract and/or agreement to provide all or a portion of the fire department transport service; (3) appropriations from the city, the county, state and federal government; and (4) such other funds as may be received for the use for fire department transport service. The emergency medical services fund shall be administered in accordance with the state laws and applicable regulations of the State Auditor.
- H. Assistance from Other Funds. Assistance for fire department EMS and transport services from other funds shall be accounted for in accordance with state law, applicable regulations of the State Auditor, and in accordance with RCW <u>36.32.470</u>.
- I. Fire Department Transport Service Fees and Rates. Fees for users of the city fire department transport service shall be fixed from time to time at the rates as set by ordinance of the city as amended from time to time; provided, that the fire department may adjust the charges fixed by ordinance yearly for any cost-of-living adjustment (COLA) increase as measured by the Consumer Price Index Pacific Cities and U.S. Cities average for the Seattle-Tacoma-Everett areas; and provided, that the fire chief is authorized to administratively set rates for medical supplies. Rates for services and medical supplies shall be set to provide for recovery of actual costs based upon an average charge, which will be reviewed annually.
- J. *Use of Emergency Medical Services Fees*. All proceeds derived from the fire department transport service fees authorized under this section shall be used solely for the operation, maintenance, and capital needs of fire department transport service.
- K. *Billing of Emergency Medical Services Fees.* Each patient who utilizes the fire department transport service shall be billed by the city in accordance with the fee schedule adopted in subsection N of this section. The department of finance shall establish a procedure to bill and collect fire department transport service fees for the services rendered. Under the procedure, the city may elect to bill the patient directly or bill the patient's insurance company. If the patient's insurance does not cover all of the cost of the ambulance service, the city will bill the remaining balance to the patient directly. The city may contract with a public or private entity to bill and collect the fire department transport service fees.

- L. *Civil Enforcement*. The fire department transport service fees imposed by this section may be collected by appropriate civil action instituted by the city attorney for that purpose.
- M. Fee Determination Criteria. Fire department transport service fees shall be set at a level to cover the actual costs incurred by the city in providing the fire department transport service. It is the intent of the city that fire department transport service fees shall be charged in exchange for targeted fire department transport service that alleviates the burden placed on such service by its users. In classifying customers served, or ALS, or BLS fire department transport service, the city may, in its discretion, consider any or all of the following factors: the difference in cost of service to various customers; the location of the various customers within the city; the difference of cost of maintenance, operation, repair and replacement of the various equipment used for fire department transport service; the different character of service furnished various customers; and any other criteria or matters which constitute a reasonable ground for distinction.
- N. Fire Department Transport Fee Imposed. Fire department transport fees shall be as follows:
- 1. Patients for whom the fire department provides ALS 1 transport shall be billed eight hundred ninety dollars one thousand and thirty nine dollars and thirty two cents (\$1039.32) per transport, the cost of the medical supplies used in the transport, and eighteen dollars and fifty cents twenty-one dollars and sixty cents (\$21.60) per mile. For the purposes of this subsection, "ALS 1" means the following:

Transportation by ground ambulance vehicle and the provision of medically necessary supplies and services, including the provision of an ALS assessment or at least one ALS intervention, which means a procedure that is, pursuant to state and local laws, beyond the scope of practice of an emergency medical technician—basic (EMT—basic).

2. Patients for whom the fire department provides ALS 2 transport shall be billed nine hundred eighty dollars one thousand one hundred forty-four dollars and forty-two cents (\$1144.42) per transport, the cost of the medical supplies used in the transport, and eighteen dollars and fifty cents twenty-one dollars and sixty cents (\$21.60) per mile. For the purposes of this subsection, "ALS 2" means as follows:

Transportation by ground ambulance vehicle and the provision of medically necessary supplies and services including either:

- a. At least three separate administrations of one or more medications by intravenous push/bolus or by continuous infusion in accordance with approved protocols, including, but not limited to, the American Heart Association Advanced Cardiac Life Support (ACLS) protocol; or
- b. At least one of the following ALS 2 procedures:
- Manual defibrillation/cardioversion;
- (2) Endotracheal intubation;



- (3) Central venous line;
- (4) Cardiac pacing;
- (5) Chest decompression;
- (6) Surgical airway; or
- (7) Intraosseous line.
- 3. Patients for whom BLS transport is provided by the fire department shall be billed <u>seven</u> <u>hundred ninety four dollars and eight cents</u> (\$794.08) <u>six hundred eighty dollars</u> per transport, the cost of the medical supplies used in the transport, and <u>eighteen dollars and fifty cents</u> <u>twenty-one dollars and sixty cents</u> (\$21.60) per mile. For the purposes of this subsection, "BLS" means as follows:

Transportation by ground ambulance vehicle and the provision of medically necessary supplies and services involving noninvasive emergency medical services requiring basic medical treatment skills. The ambulance must be staffed by an individual who is qualified, pursuant to state and local laws, to be an emergency medical technician—basic (EMT—basic).

- 4. The fire chief is authorized to administratively set the cost of medical supplies. A current list of such costs is available at the fire department.
- 5. The rates and fees in this subsection N are effective until June 30, 2024. The fire chief, in accordance with subsection I above, will increase these rates and fees effective each July 1 thereafter.
- O. Compliance with Medicare and Medicaid Requirements. This chapter and the fire department transport service fees hereunder shall be construed and implemented in a manner consistent with applicable Medicare and Medicaid requirements. If any part of this chapter is found to conflict with Medicare and Medicaid requirements, the conflicting part of this chapter shall be inoperative to the extent of the conflict and such finding or determination shall not affect the operation of the remainder of this chapter

**Section 3**. The following sections of chapter 5.72 EMC shall be recodified as a new chapter 3.82 EMC entitled "Emergency Medical Services" as follows:

Current Code Section	New Code Section
5.72.010 (Definitions)	3.82.010 (Definitions)
5.72.070 (Duties of fire chief)	3.82.020 (Duties of fire chief)
5.72.120 (City emergency medical services director)	3.82.030 (City emergency medical services director)

5.72.140 (Emergency medical response)	3.82.040 (Emergency medical response)
5.72.150 (Ambulance contract)	3.82.050 (Ambulance contract)
5.72.160 (Training)	3.82.060 (Training)
5.72.190 (Findings and declaration)	3.82.070 (Findings and declaration)
5.72.200 (Fire department EMS	3.82.080 (Fire department EMS
transportation)	transportation)
5.72.210 (Operation and funding)	3.82.090 (Operation and funding)
5.72.220 (Additional funding)	3.82.100 (Additional funding)

**Section 3.** The following is provided for reference and may not be complete:

EMC Repealed/Recodified by this Ordinance	Ordinance History of EMC Repealed/Recodified by this Ordinance
EMC 5.72.010	(Ord. 2792-04 § 2, 2004; Ord. 2308-98 § 1, 1998.)
EMC 5.72.020	(Ord. 2308-98 § 2, 1998.)
EMC 5.72.030	(Ord. 3206-10 § 1, 2010; Ord. 2308-98 § 3, 1998.)
EMC 5.72.040	(Ord. 2308-98 § 4, 1998.)
EMC 5.72.050	(Ord. 2308-98 § 5, 1998.)
EMC 5.72.060	(Ord. 2308-98 § 6, 1998.)
EMC 5.72.070	(Ord. 2308-98 § 7, 1998.)
EMC 5.72.080	(Ord. 2308-98 § 8, 1998.)
EMC 5.72.090	(Ord. 2308-98 § 9, 1998.)
EMC 5.72.100	(Ord. 2308-98 § 10, 1998.)
EMC 5.72.110	(Ord. 2308-98 § 11, 1998.)
EMC 5.72.120	(Ord. 2308-98 § 12, 1998.)
EMC 5.72.130	(Ord. 2308-98 § 13, 1998.)
EMC 5.72.140	(Ord. 2308-98 § 14, 1998.)
EMC 5.72.150	(Ord. 3206-10 § 2, 2010; Ord. 2308-98 § 15, 1998.)
EMC 5.72.160	(Ord. 2308-98 § 16, 1998.)
EMC 5.72.170	(Ord. 2308-98 § 17, 1998.)
EMC 5.72.180	(Ord. 2308-98 § 19, 1998.)
EMC 5.72.190	(Ord. 2792-04 § 1, 2004; Ord. 2308-98 § 23, 1998.)
EMC 5.72.190	(Ord. 2792-04 § 1, 2004; Ord. 2308-98 § 24, 1998.)
EMC 5.72.200	(Ord. 2792-04 § 1, 2004; Ord. 2308-98 § 24, 1998.)
EMC 5.72.210	(Ord. 3722-20 § 1, 2020; Ord. 3205-10 § 1, 2010; Ord. 3049-07 § 1,

	2007; Ord. 2792-04 § 1, 2004; Ord. 2308-98 § 25, 1998.)	
EMC 5.72.220	(Ord. 2792-04 § 1, 2004; Ord. 2308-98 § 26, 1998.)	

<u>Section 4.</u> The City Clerk and the codifiers of this Ordinance are authorized to make necessary corrections to this Ordinance including, but not limited to, the correction of scrivener's/clerical errors, references, ordinance numbering, section/subsection numbers, and any internal references.

<u>Section 5</u>. The City Council hereby declares that should any section, paragraph, sentence, clause or phrase of this ordinance be declared invalid for any reason, it is the intent of the City Council that it would have passed all portions of this ordinance independent of the elimination of any such portion as may be declared invalid.

<u>Section 6</u>. The enactment of this Ordinance shall not affect any case, proceeding, appeal or other matter currently pending in any court or in any way modify any right or liability, civil or criminal, which may be in existence on the effective date of this Ordinance.

<u>Section 7</u>. It is expressly the purpose of this Ordinance to provide for and promote the health, safety and welfare of the general public and not to create or otherwise establish or designate any particular class or group of persons who will or should be especially protected or benefited by the terms of this Ordinance. It is the specific intent of this Ordinance that no provision or any term used in this Ordinance is intended to impose any duty whatsoever upon the City or any of its officers or employees. Nothing contained in this Ordinance is intended nor shall be construed to create or form the basis of any liability on the part of the City, or its officers, employees or agents, for any injury or damage resulting from any action or inaction on the part of the City related in any manner to the enforcement of this Ordinance by its officers, employees or agents.

Cassie Franklin, Mayor	
ATTEST:	
Marista Jorve, City Clerk	
PASSED:	
VALID:	
PUBLISHED:	
EFFECTIVE DATE:	
ORDINANCE	Ī



## **EVERETT** City Council Agenda Item Cover Sheet

**Project title:** 

Administration

Council President

AN ORDINANCE relating to special business licenses and regulations and repealing Chapters 5.20, 5.92, 5.96, and 5.102 of the Everett Municipal Code

Council Bill # interoffice use	Project:	Amending Ordinance to Title 5 of the Everett Municipal Code	
CB 2404-03	Partner/Supplier: NA		
Agenda dates requested:	Location:	NA	
	Preceding action:		
Briefing	Fund:		
1st Reading 5/8/24 Consent Action 5/15/24	Fiscal summary stat	tement:	
2 <sup>nd</sup> / 3 <sup>rd</sup> & Final Readings Ordinance X Public hearing	No fiscal impact.		
Yes X No	Project summary st	atement:	
Budget amendment: Yes X No	This amending ordir obsolete sections.	nance modernizes Title 5 of the Everett Municipal Code by repealing	
PowerPoint presentation:			
Yes X No	5.20	Dancehalls	
Attachments:	5.92	Garbage Removal	
Ordinance	5.96	Massage Parlors, Body Painting Studios, and Public Bathhouses	
Department(s) involved:	5.102	Circuses, Carnivals, Amusement Centers, and Go-Kart Centers	
Finance/Legal	ll		
Contact person: Susy Haugen	Recommendation (	exact action requested of Council):	
Phone number: X8612	Adopt ordinance relating to special business licenses and regulations and repealing Chapters 5.20, 5.92, 5.96, and 5.102 of the Everett Municipal Code		
Email: shaugen@everettwa.gov			
Initialed by: Sh Department head			



ORDINANCE	NO.		
-----------	-----	--	--

AN ORDINANCE relating to special business licenses and regulations and repealing Chapters 5.20, 5.92, 5.96 and 5.102 of the Everett Municipal Code.

#### WHEREAS,

- A. Title 5 of the EMC contains the City's special business licenses and regulations.
- **B.** Most were adopted more than 30 years ago and are obsolete. Many have been unused for many years, sometimes decades, and some may conflict with state law.
- **C.** In particular, the following chapters of Title 5 are no longer needed:

5.20	Dancehalls
5.92	Garbage Removal
5.96	Massage Parlors, Body Painting Studios, and Public Bathhouses
5.102	Circuses, Carnivals, Amusement Centers, and Go-Kart Centers

#### NOW, THEREFORE, THE CITY OF EVERETT DOES ORDAIN:

**Section 1.** Chapters 5.20, 5.92, 5.96, and 5.102 of the Everett Municipal Code are hereby repealed.

<u>Section 2.</u> The City Clerk and the codifiers of this Ordinance are authorized to make necessary corrections to this Ordinance including, but not limited to, the correction of scrivener's/clerical errors, references, ordinance numbering, section/subsection numbers, and any internal references.

<u>Section 3</u>. The City Council hereby declares that should any section, paragraph, sentence, clause or phrase of this ordinance be declared invalid for any reason, it is the intent of the City Council that it would have passed all portions of this ordinance independent of the elimination of any such portion as may be declared invalid.

<u>Section 4.</u> The enactment of this Ordinance shall not affect any case, proceeding, appeal or other matter currently pending in any court or in any way modify any right or liability, civil or criminal, which may be in existence on the effective date of this Ordinance.

Section 5. It is expressly the purpose of this Ordinance to provide for and promote the health, safety and welfare of the general public and not to create or otherwise establish or designate any particular class or group of persons who will or should be especially protected or benefited by the terms of this Ordinance. It is the specific intent of this Ordinance that no provision or any term used in this Ordinance is intended to impose any duty whatsoever upon the City or any of its officers or employees. Nothing contained in this Ordinance is intended nor shall be construed to create or form the basis of any liability on the part of the City, or its officers, employees or agents, for any injury or damage resulting from any action or inaction on the part of the City related in any manner to the enforcement of this Ordinance by its officers, employees or agents.

Cassie Franklin, Mayor	
ATTEST:	
City Clerk	
PASSED:	
VALID:	
PUBLISHED:	
EFFECTIVE DATE:	



**Project title:** 

An Ordinance Relating to Admissions Tax, Repealing Chapter 3.20 of the Everett Municipal Code and Adopting a New Chapter 3.20 of the Everett Municipal Code.

Council Bill # interoffice use	Project: Admissions Tax Ordinance
CB 2404-04	
	Partner/Supplier: NA
Agenda dates requested:	Location: N/A
Briefing	Preceding action: N/A
1 <sup>st</sup> Reading: 5/8/24 Consent	Fund: N/A
Action 5/15/24  2nd / 3rd & Final Readings  Ordinance X  Public hearing  Yes X No	Fiscal summary statement:  No fiscal impact
Budget amendment:	Project summary statement:
Yes X No  PowerPoint presentation:	The Admissions Tax Ordinance was adopted over 50 years ago and requires significant updating and clarification.
Yes X No  Attachments: Ordinance	Key updates include clarification of Admissions Tax exemptions, removal of conflicting provisions relating to golf and making the tax rate for all taxpayers 5%, and significant simplification of how the tax is applied and calculated.
Department(s) involved: Finance	The proposed code as revised will improve administration and be much easier for the public to understand and follow.
Contact person: Director Susy Haugen Phone number:	Recommendation (exact action requested of Council):  Adopt the Ordinance relating to Admissions Tax, repealing Chapter 3.20 of the Everett Municipa Code and Adopting a New Chapter 3.20 of the Everett Municipal Code.
Email:	
Initialed by:	
sh	
Department head	
Administration	
Council President	



ORDINANCE NO.	
---------------	--

An Ordinance Relating to Admissions Tax, Repealing Chapter 3.20 of the Everett Municipal Code and Adopting a New Chapter 3.20 of the Everett Municipal Code.

#### WHEREAS,

- A. The admissions tax code, which is chapter 3.20 EMC, was adopted over 50 years ago and requires updating and clarification.
- B. The purpose of this ordinance is to replace the current chapter 3.20 EMC with a new chapter 3.20 EMC with updating and clarification.

#### NOW THEREFORE, THE CITY OF EVERETT DOES ORDAIN:

**Section 1.** The City Council hereby finds the foregoing recitals are adopted in support of this Ordinance.

<u>Section 2.</u> Chapter 3.20 EMC is hereby repealed and replaced with the Chapter 3.20 EMC as set forth in Exhibit A attached and incorporated herein.

<u>Section 3.</u> The City Clerk and codifiers of the ordinance are authorized to make necessary corrections to this Ordinance including, but not limited to, the correction of scrivener/clerical errors, references, ordinance numbering, section/subsection numbers and any references thereto.

**Section 4.** Should any section, paragraph, sentence, clause or phrase of this Ordinance, or its application to any person or situation, be declared unconstitutional or otherwise invalid for any reason, or should any portion of this Ordinance be preempted by state or federal law or regulation, such decision or preemption shall not affect the validity of the remaining portions of this Ordinance or its application to other persons or circumstances. The City Council of the City of Everett hereby declares that it would have adopted this Ordinance and each section, subsection, sentence, clause, phrase, or portion thereof irrespective of the fact that any one or more sections, subsections, clauses, phrases, or portions be declared invalid or unconstitutional.

<u>Section 5.</u> The enactment of this Ordinance shall not affect any case, proceeding, appeal or other matter currently pending in any court or in any way modify any right or liability, civil or criminal, which may be in existence on the effective date of this Ordinance.

**Section 6.** It is expressly the purpose of this Ordinance to provide for and promote the health, safety and welfare of the general public and not to create or otherwise establish or designate any particular class or group of persons who will or should be especially protected or benefited by the terms of this Ordinance. It is the specific intent of this Ordinance that no provision or any term used in this Ordinance is intended to impose any duty whatsoever upon the City or any of its officers or employees. Nothing contained in this Ordinance is intended nor shall be construed to create or form the basis of any liability on the part of the City, or its officers, employees or agents, for any injury or damage resulting from any action or inaction on the part of the City related in any manner to the enforcement of this Ordinance by its officers, employees or agents.

Cassie Franklin, Mayor	
ATTEST	
City Clerk	
PASSED:	
VALID:	
PUBLISHED:	
FEFECTIVE DATE:	

#### **EXHIBIT A**

### EMC Chapter 3.20 ADMISSIONS TAX

3.20.010	Definitions.
3.20.020	Tax levied—Amount.
3.20.030	Tax not levied—Exemptions.
3.20.040	${\it Collection\ of\ tax-Duties\ of\ collector-Returns\ required}.$
3.20.050	Registration with city clerk.
3.20.060	Penalty for violations.

#### 3.20.010 Definitions.

For purposes of this chapter, these words and phrases shall have the following meanings:

- A. "Admission charge," in addition to its usual and ordinary meaning, includes a charge made for season tickets or subscriptions; a cover charge or a charge made for use of seats and tables, reserved or otherwise, and similar accommodations; a required charge made for food and refreshments in lieu of a cover charge in any place where any entertainment, recreation, or amusement is provided; and a charge made for rental or use of equipment or facilities for purposes of recreation or amusement (provided that where the rental of the equipment or facilities is necessary to the enjoyment of the privilege for which a general admission is charged, the combined charge shall be considered as the admission charge).
- B. The "clerk" means the City clerk.
- C. A "place" can be either indoors or outdoors and includes, but is not limited to; theaters, auditoriums, music venues, drinking establishments, stadiums, athletic pavilions and fields, baseball and athletic parks, bowling alleys, golf courses, circuses, carnivals, swimming pools, and amusement parks.
- D. "Person" means any individual, receiver, assignee, firm, co-partnership, joint venture, corporation, company, joint stock association, society, entity, or any group of individuals acting as a unit, whether mutual, cooperative or fraternal. Regardless of the foregoing, when an individual or entity pays an admission charge to an organization recognized by the United States of America as exempt from federal income taxation pursuant to Section 501(c)(1) or (3) of the Internal Revenue Code, such paying individual or entity is not, as to that admission charge, a "person" for the purposes of this chapter.
- E. "Admissions tax" means the tax imposed under EMC 3.20.020 and this Chapter.

### 3.20.020 Tax levied—Amount.

Unless subject to exemption under EMC 3.20.030 or RCW 35.21.280 or successor law, there is levied and imposed, upon every person who pays an admission charge to any place, a tax of five percent paid for the admission charge.

#### 3.20.030 Tax not levied—Exemptions.

- A. An admission charge \$5.00 or less is exempt from admissions tax.
- B. An admission charge for admission to the Regional Center owned, operated and maintained by the Everett Public Facilities District under Chapter 35.57 RCW, commonly referred to as the Everett Events Center, is exempt from admissions tax. This exemption includes exemption from admissions tax for any admission charge to paid to Everett Public Facilities District or paid to any person with whom the Everett Public Facilities District has either directly or indirectly contracted for the putting on of activities such as games, plays, performances, displays, or other similar activities in the Everett Events Center.
- C. Amounts paid to any person for initiation fees, or paid as dues on a membership basis in lieu of admissions charges for members, are exempt from admissions tax. However, admission charges paid by non-members are subject to admissions tax.
- D. Amounts paid to any person for admission by season ticket or subscription are exempt from admissions tax if the amount which would be charged to the season ticket holder or subscriber for a single admission is \$5.00 or less. Otherwise, amounts paid for admission by season ticket or subscription are subject to admissions tax.

#### 3.20.040 Collection of tax—Duties of collector—Returns required—Administration.

- A. Admissions tax, unless exempted under this Chapter, shall be collected at the time an admission charge is paid by any person for admission to any place. The admissions tax shall be collected by the person receiving the admission charge and shall be held in trust until paid to the City.
- B. Any person required to collect admissions tax, unless exempted under this Chapter, who fails to collect such admissions tax shall be personally liable to the City for the amount of such tax and, unless remittance is made to the City as required by this Chapter, shall be guilty of a violation of this chapter in accordance with Chapter 3.19 EMC.
- C. Admissions tax collected by any person shall be reported and remitted by that person to the clerk in quarterly installments on a form(s) provided by the clerk.
- D. Any person, reasonably believed to be required to collect admissions tax, may be subject to an examination or audit by the clerk in accordance with EMC 3.19.060.

#### 3.20.050 Registration with city clerk.

Any person conducting or operating any place which requires an admission charge for entry shall obtain a general business license from the clerk and shall indicate in its application for such license that admissions tax is likely to be collected by that person.

#### 3.20.060 Administration.

This Chapter shall be administered in accordance with Chapter 3.19 of the Everett Municipal Code.



## **PROCLAMATION**

**Whereas,** Jewish American Heritage Month is celebrated each May as an opportunity to honor generations of Jewish Americans and their contributions to American history, culture and society;

And, Everett proudly embraces the great diversity among those who live, work and visit our city;

**And,** Jewish people are an important part of the city's fabric, with members of Temple Beth Or in Everett contributing to our community in many important ways;

**And,** we recognize the many different ways Jewish Americans connect to their Jewish identity, including culturally, religiously, ethnically and by following Jewish values;

**And,** as we celebrate the deep history of the Jewish American community, we also must acknowledge the increasing presence of antisemitism in Jewish communities throughout our state and nation;

**And,** we must be proactive in educating people about antisemitism as well as in identifying and addressing harmful anti-Jewish rhetoric;

**And**, the City of Everett stands firmly against antisemitism and condemns all forms of hate and bigotry;

**And,** we reaffirm our commitment to combating antisemitism wherever it exists and preserving the public safety, security, well-being and equal treatment of our Jewish community members;

**And,** Everett is made stronger by the rich diversity in our community and we are proud to celebrate our Jewish American community members this month.

**NOW, THEREFORE, I,** Cassie Franklin, Mayor of the City of Everett, and on behalf of its City Council, do hereby proclaim May 2024 as

#### "JEWISH AMERICAN HERITAGE MONTH"

And urge the residents of our community to join us in celebrating Jewish American Heritage and standing up against antisemitism and all other forms of hate impacting the Jewish community.

Signed this 1st day of May 2024.

Mayor Cassie Franklin



# EVERETT CITY COUNCIL Public Comment Form

Thank you for being here today. Please fill out this form to speak at the council meeting.

State your name and city of residence when you begin speaking. Each person is asked to limit comments to three minutes. This allows everyone a fair opportunity to speak. Return this form to the council administrator before the meeting begins.

All comments must be relevant and delivered to the Council as a whole in a respectful manner. The following comments are not allowed:

- Comments on any kind of campaigning, whether for or against ballot measures or candidates running for office
- Comments advertising any product
- Comments focused on personal matters that are unrelated to City business

You can also submit a comment and attend meetings online at **everettwa.gov/city council**. Click on "Council meeting public comment sign up form." This must be done at least 30 minutes prior to the meeting. Additional instructions are available on the web page.

City staff may wish to contact you for follow up, therefore, your contact information is appreciated.

DATE:
NAME (required):
CITY (required):
EMAIL (optional):PHONE (optional):
DISTRICT (circle one): 1 2 3 4 5 Not sure Don't live in city
When would you like to deliver your comments: Is your topic on today's agenda?
During the comment period that will follow the agenda item  AGENDA ITEM #:
During the general public comment. Please state the topic you would like to speak on: First jf Posi & De